U'H NOOK NOH KHUNUK (IN OUR OWN WORDS): LAKE BABINE NATION PARENTS' UNDERSTANDING OF SCHOOL ASSESSMENT

By

Travis Holyk
Master of Arts, University of Northern BC 2001
Bachelor of Arts, Okanagan University College, 1999

THESIS SUBMITTED IN PARTIAL FULFILMENT OF THE REQUIREMENTS FOR THE DEGREE OF

DOCTOR OF EDUCATION

In the Faculty of Education

© Travis Holyk 2011

SIMON FRASER UNIVERSITY

Spring 2011

All rights reserved. However, in accordance with the *Copyright Act of Canada*, this work may be reproduced, without authorization, under the conditions for *Fair Dealing*. Therefore, limited reproduction of this work for the purposes of private study, research, criticism, review and news reporting is likely to be in accordance with the law, particularly if cited appropriately.

APPROVAL

| Name: | Travis Holyk |
|-------------------------|--|
| Degree: | Doctor of Education |
| Title of Thesis: | U'h Nook Noh Khunuk (In Our Own Words): Lake Babine Nation Parents' Understanding of School Assessment |
| Examining Committee: | |
| Chair: | |
| | Dr. Geoff Madoc-Jones Chair, Senior Lecturer |
| | Dr. Daniel Laitsch Senior Supervisor, Assistant Professor, SFU |
| | Dr. Byron Robbie Co-Supervisor, Adjunct Professor, SFU |
| | Mr. Warner Adam Member Executive Director, Carrier Sekani Family Services |
| | Dr. Michelle Pidgeon Internal Examiner, Assistant Professor, SFU |
| | Dr. Tina Fraser External Examiner Assistant Professor, School of Education, UNBC |
| Date Defended/Approved: | 21 April 2011 |



Declaration of Partial Copyright Licence

The author, whose copyright is declared on the title page of this work, has granted to Simon Fraser University the right to lend this thesis, project or extended essay to users of the Simon Fraser University Library, and to make partial or single copies only for such users or in response to a request from the library of any other university, or other educational institution, on its own behalf or for one of its users.

The author has further granted permission to Simon Fraser University to keep or make a digital copy for use in its circulating collection (currently available to the public at the "Institutional Repository" link of the SFU Library website <www.lib.sfu.ca> at: http://ir.lib.sfu.ca/handle/1892/112) and, without changing the content, to translate the thesis/project or extended essays, if technically possible, to any medium or format for the purpose of preservation of the digital work.

The author has further agreed that permission for multiple copying of this work for scholarly purposes may be granted by either the author or the Dean of Graduate Studies.

It is understood that copying or publication of this work for financial gain shall not be allowed without the author's written permission.

Permission for public performance, or limited permission for private scholarly use, of any multimedia materials forming part of this work, may have been granted by the author. This information may be found on the separately catalogued multimedia material and in the signed Partial Copyright Licence.

While licensing SFU to permit the above uses, the author retains copyright in the thesis, project or extended essays, including the right to change the work for subsequent purposes, including editing and publishing the work in whole or in part, and licensing other parties, as the author may desire.

The original Partial Copyright Licence attesting to these terms, and signed by this author, may be found in the original bound copy of this work, retained in the Simon Fraser University Archive.

Simon Fraser University Library Burnaby, BC, Canada



STATEMENT OF ETHICS APPROVAL

The author, whose name appears on the title page of this work, has obtained, for the research described in this work, either:

(a) Human research ethics approval from the Simon Fraser University Office of Research Ethics.

or

(b) Advance approval of the animal care protocol from the University Animal Care Committee of Simon Fraser University;

or has conducted the research

(c) as a co-investigator, collaborator or research assistant in a research project approved in advance,

or

(d) as a member of a course approved in advance for minimal risk human research, by the Office of Research Ethics.

A copy of the approval letter has been filed at the Theses Office of the University Library at the time of submission of this thesis or project.

The original application for approval and letter of approval are filed with the relevant offices. Inquiries may be directed to those authorities.

> Simon Fraser University Library Simon Fraser University Burnaby, BC, Canada

> > Last update: Spring 2010

ABSTRACT

This research analyzes the perception of assessment among parents of Grade 4 students within the Lake Babine Nation. The Lake Babine Nation, located in central British Columbia, represents 2228 members and has three currently inhabited communities. Eighteen semi-structured interviews were completed and parents provided perspective regarding their understanding of assessment and how schools can better involve parents in the education process. Major findings include the identification of communication as a key component of student success in school. Communication was noted as important in providing parents with information necessary to assist their children, improve levels of trust and contend with historical factors influencing parent's perceptions of school. Parents emphasized education as necessary for establishing a better life and placed importance on their children graduating with a Dogwood Certificate as a measure of success. Parents supported measures such as provincial testing to ensure the level of education a child receives is similar anywhere in the province.

Keywords: Assessment; First Nations; Lake Babine Nation; Education; Parents;

Elementary; Culture; Communication

DEDICATION

To my grandfather, George Hanschen, who instilled in us the importance of Education.

ACKNOWLEDGEMENTS

This dissertation was created in the spirit of collaboration between researcher and community. I was fortunate to have the support of a number of individuals in the completion of this work and for their guidance I am eternally grateful. The process taught me as much as the document itself.

This research would not have taken place without the support of Warner Adam and Carrier Sekani Family Services.

The Late Perry Shawana, without whose influence I never would have attempted to complete a doctorate. He saw the best in everyone and encouraged them to accomplish their goals. His knowledge and ability to exchange ideas is missed by all whose lives he touched.

The Lake Babine Nation and Chief Wilf Adam, and former Chief Betty Patrick, thank you for approving of the research as a worthwhile endeavor and allowing me to work with the community.

In the spirit of community based research, Rosa Johnny was integral to the research liaising with community participants and assisting with transcription. Ruby Adam and Georgina Patrick assisted in providing contact information for parents involved in the study. Darcy Dennis, Evelyn George *tla* Benna Rathburn, *mussi cho nowh gunik soh dee leh*.

Mussi Cho to Members of the Lake Babine Nation Steering Committee: Jim Williams, Nancy Williams, Youlanda Spenst, Verna Powers and Murphy Patrick Jr. who acted as advisors and guides to this research, providing information and their time in order to ensure the benefit of the information to the community.

NEARBC provided both Fellowship and research funding to this project, Chris Ross assisted with formatting and Evelyn George provided a traditional lunch to celebrate the dissertation defense.

Thank you to Dan Laitsch for all your insight into this work and for agreeing to be my supervisor despite all of your time commitments. And, thanks Warner and Byron Robbie for their help in preparing the Dissertation. The diverse knowledge of the Dissertation committee added to the development of the dissertation and created an improved document.

Payton, Morgan and Carmen Holyk, thanks for reminding me there are things more important than research.

Mom and Dad, you never take the credit you deserve for the support you have provided me throughout my life, thanks again.

TABLE OF CONTENTS

| Approval | ii |
|---|------|
| Abstract | iii |
| Dedication | iv |
| Acknowledgements | v |
| Table of Contents | vi |
| List of Tables | viii |
| 1: Chapter One: Introduction | 1 |
| 1.1 Colonialism and the Canadian Education System | 5 |
| 1.2 The Current Situation of Aboriginal Education | |
| 1.3 Assessment | 12 |
| 1.4 BC Context | 15 |
| 1.5 Involving First Nations Parents in their Children's Education | 18 |
| 1.6 Parent Involvement and High Standards | |
| 1.7 Final Remarks | 21 |
| 2: Chapter Two: Assessment in Historical Context | 32 |
| 2.1 Assessment and Indigenous Students | 39 |
| 2.2 Eras of Assessment in British Columbia | |
| 2.3 Intelligence Testing | |
| 2.4 Foundation Skills Assessment | |
| 2.5 Final Remarks | 56 |
| 3: Chapter Three: The Lake Babine Nation: Worldview and Education | 57 |
| 3.1 The Lake Babine Nation | 60 |
| 3.2 Social Imaginary | |
| 3.3 Spirituality and Worldview | |
| 3.4 Governance | |
| 3.5 Economics | |
| 3.6 Reclaiming Social Imaginary | |
| | |
| 4: Chapter Four: Methods | |
| 4.1 History of Research in Indigenous Communities | |
| 4.2 Research Framework and Priorities | |
| 4.3 Sample Overview- Building Lasting Relationships | |
| 4.4 Participatory Action Research | |
| T.O COMMUNITY LINGUISTIC TITLE TO THE TOTAL COMMUNITY LINGUISTIC TO THE TOTAL COMMUNITY LINGUIST LINGUISTIC TO THE TOTAL COMMUNITY LINGUISTIC | 91 |

| 4.6 Questionnaire Design | 101 |
|--|-----|
| 4.7 Research Sample | |
| 4.8 Data Collection | |
| 4.9 Ownership of the Collected Data | |
| 4.10 Data Analysis | 113 |
| 5: Chapter Five: Results | 125 |
| 5.1 Demographic Information | 127 |
| 5.2 Education | 129 |
| 5.3 Achievement in School | 135 |
| 5.4 Role of the School | 140 |
| 5.5 Information Received | |
| 5.6 Behaviour of parents after receiving information | |
| 5.7 Agree with Results | 148 |
| 5.8 Communication | |
| 5.9 Teacher's Methods of Assessment | |
| 5.10 Strategies used to check | |
| 5.11 Challenging Carrier Students | |
| 5.12 Extra Reading Classes | |
| 5.13 Passing to Next Grade | |
| 5.14 Provincial Testing | 166 |
| 5.15 Additional Forms of Assessment | |
| 5.16 Final Remarks | 173 |
| 6: Chapter 6 Discussion, Recommendations and Implications | 175 |
| 6.1 Caveat | 175 |
| 6.2 Discussion | 176 |
| 6.3 Key Findings | |
| 6.4 Recommendations | |
| 6.5 Implications | 192 |
| 7: Reference List | 194 |
| 8: Appendix A: LBN Research Protocol | 208 |
| 9: Appendix B: LBN Letter of Approval | 213 |
| 10: Appendix C: Research Questions and Interview Protocol | 215 |
| 11: Appendix D: Informed Consent By Participants In a Research Study | |

LIST OF TABLES

| Table 1. Carrier First Nations Values, Practices and Methods Utilized | 90 |
|---|-----|
| Table 2 Community Consultation Steps | 97 |
| Table 3. LBN Steering Committee Assistance with Questionnaire | 102 |
| Table 4. A Priori Themes | 115 |
| Table 5. Sample of Transcript Excerpt Edited for a priori Theme | 116 |
| Table 6. Words Commonly Occurring | 117 |
| Table 7. Sample Excerpt of Transcript into a Single Document | 120 |
| Table 8. Sample of Theme document – Code Identification | 121 |
| Table 9. Example of Education Theme Coding | 123 |
| Table 10. Residency | 128 |
| Table 11. Education Characteristics | 128 |
| Table 12. Income and Employment | 129 |
| Table 13. Carrier Educated Theme Coding | 133 |
| Table 14. Achievement Theme Coding | 136 |
| Table 15. Theme Subjects | 137 |
| Table 16. Theme Information Received | 142 |
| Table 17. Theme Behaviour after Receiving Information | 146 |
| Table 18. Strategies to check Student Progress | 157 |

1: CHAPTER ONE: INTRODUCTION

This dissertation is in large part a result of my continuing relationship with the Carrier Sekani peoples through my involvement in Carrier Sekani Family Services as the Director of Research and Policy Development. Carrier Sekani Family Services is an agency responsible for health, social and legal services to eleven member First Nations, including the Lake Babine Nation, in north central British Columbia. I am not First Nations but have worked with the community for 10 years and have learned important lessons about conducting research respectfully that serve as the core values in this dissertation. Carrier and Sekani people maintain that they have an inherent Aboriginal right to be self-determining which includes the ability to use Carrier and Sekani institutions in maintaining social order in their communities, and that this right is now protected under Section 35 of the *Constitution Act*, 1982 (Holyk, Shawana & Adam, 2005).

Assuming responsibility for the care and protection of children, as well as the maintenance of family structures and governance systems are expressions of

self-determination. Indigenous rights are further affirmed in the United Nations

Declaration on the Rights of Indigenous Peoples (2007) which notes the right to
self-determination in pursuing economic, social and cultural development. Article
14 specifically addressing education asserting,

Indigenous peoples have the right to establish and control their education systems and institutions providing education in their own languages, in a manner appropriate to their cultural methods of teaching and learning (UN Declaration on the Rights of Indigenous Peoples, 2007).

Premised on this rationale, Carrier Sekani Family Services has worked hard to reassert control in areas that have been sources of attack on community values and teachings such as maintenance of Carrier governance systems, working toward the cultural relevance of classroom materials (curriculum development and administration included) as well as preservation of family and guardianship of children.

Working at Carrier Sekani Family Services enabled me to dialogue with leaders who are at the centre of the political and social struggle and determine sites of resistance, one of which, education, became the start of my general dissertation topic. The Lake Babine Nation is a member of Carrier Sekani Family Services receiving social, research and child welfare services. Leadership at Carrier Sekani Family Services have continually recognized education as integral to improving the lives of Carrier Citizens. Education is also understood within the context of the negative impacts of colonization on the lives of First Nations peoples in Canada, of which the Lake Babine Nation represented in this dissertation are included.

Despite the negative impacts of the education system and its colonial roots, the reality for First Nations students today is that in order to improve their life chances they must negotiate the education system. Research indicates BC youth who drop out before completing high school are less healthy and have poorer family functioning than do those who graduate (Joint Special Report 2007). Clyde Hertzman (1998) argues, "if childhood interventions were to improve health and wellbeing throughout the lifecycle, and fundamentally address the socioeconomic gradient in health status, they would be more likely to be social or educational interventions than health interventions per se" (p.19). In Canada, not graduating is associated with a number of indicators of well being including time spent on income assistance and poverty. Richards and Vining (2004), examining the 1996 and 2001 Canadian Census data indicate, "as the education level of Aboriginals rises, so do their median incomes" (p.2). Income and education are positively correlated, increasing by stages between those who graduate high school, those who complete a trade certificate and those who complete a university degree (Richards & Vining, 2004).

It was in the summer of 2006, while at Simon Fraser University, that I decided to focus my research at the intersection of Aboriginal parents¹ and educational assessment. During the summer semester, the Education Program invited a number of guest speakers to the university to speak on topics related to education. One speaker, Peter Cowley of the Fraser Institute, was invited to speak on his co-authored report, *Report Card on Aboriginal Education in British*

¹ In this research, I use the term parent to generally describe caregivers of children. It includes parents, relatives or other community members who may be looking after children. Foster parents who are not from the Lake Babine Nation were not included in the study.

Columbia: 2006 Edition. In his discussion, he noted the poorer achievement of Aboriginal students and the need for regular assessment of Aboriginal students and the public reporting of results. I questioned Peter Cowley regarding the Fraser Institute's determination of indicators of success, and, whether the Fraser Institute had consulted with First Nations communities to obtain their views regarding appropriate indicators. I also noted the potential damage results could cause in terms of promoting racism by publicly presenting schools with predominantly First Nation's students in a negative light without explanation of the social or cultural context. His response was, if First Nations are not happy with our indicators they are welcome to produce their own (Cowley, July 19, 2006). His statement encouraged me to think about parent's views of assessment and whether First Nations parents had been consulted regarding what they expect their children to receive from public education. Upon returning to Carrier Sekani Family Services and discussing with leadership² the concept of assessment and parents' involvement in education the following questions became the research topics of this dissertation; What do Lake Babine Nation parents of Grade 4 children deem as important indicators of their children's achievement in school and; what is Lake Babine Nation parent's understanding of their children's academic assessment?

The goal of this research became to examine the perceptions of parents of Grade 4 students from the Lake Babine Nation regarding the education system and in particular the processes used in the system to assess student learning.

-

² The consultation process with Carrier leaders is discussed in greater detail in the methods chapter.

The analysis included definitions of student achievement and the concept of education from the Lake Babine Nation parent's perspective in order to address what parents believe their children should know and thus identify those methods most likely to be meaningful to parents.

The Lake Babine Nation is the largest community within Carrier Sekani Family Services representing 2284 registered members³ (INAC, 2008). Of this total, roughly 1500 people live on reserve and the remaining individuals reside in surrounding communities as well as distant cities (INAC, 2008). The Lake Babine Nation currently has three communities/reserves inhabited year round, only two of which are within their traditional territory and all of which participated in this research.⁴ The Lakes District, School District No. 91, is the district servicing on reserve Lake Babine children specifically as the Lake Babine community of Fort Babine has a band run school to Grade 7. Post Grade 7 children travel to Smithers to attend classes in School District No. 54. In total, during the 2008/2009 school year in which I conducted this research, there were 36 children from 35 families registered with the Lake Babine Nation in Grade 4.

1.1 Colonialism and the Canadian Education System

The Carrier and Sekani people's right to self-determination has been eroded in part by provisions of the *Indian Act* and related policies that attempted

The LBN Steering Committee contends membership is higher but for the purposes of the research I used those registered with Indian and Northern Affairs Canada to determine registered population.

⁴ In the 1950s members of the Lake Babine Nation were moved from their traditional territory to Burns Lake to ease service provision. The community of Woyenne was created for this population. (Fiske, 2000).

to govern the lives of indigenous peoples in Canada. In 1876, all laws affecting Indian people were combined under one piece of legislation known as the *Indian Act*, which stated as its mandate the protection, civilization and assimilation of Indians. The consolidated *Indian Act* is a comprehensive piece of legislation providing legislated direction toward almost all activities and aspects of First Nation communities. Under the *Act*, most vestiges of powers reside with the Minister of Indian Affairs and Indian Agents. The underlying intent of this legislation was the elimination of Indigenous worldviews and any jurisdiction over their territories⁵ (Joseph, 1991). The three areas in particular that were addressed under the *Act* included band membership, local government construction and land ownership⁶ (Moore, 1978).

The government, aided by the Catholic and Anglican churches, enhanced its programs of political and cultural destruction by attacking the family structure, which is the mechanism for the transmission of traditional knowledge. This was largely completed through the removal of children from their homes, and placing them in residential schools in order to prevent them from practicing their culture or passing on knowledge through separation from parents and other teachers (Kelm, 1998). As noted by Ball (2004),

One of the main avenues for subjugating Indigenous peoples to colonial culture and governance has been through the imposition of education, most powerfully through the "Indian Residential Schools" program, that denies the legitimacy of thought, lifestyles, religions and languages of First Nations people (p.457).

The legislation also solidified the lack of recognition, present from the arrival of Europeans, of the matrilineal organization of Carrier communities and the organization of family and property.

⁶ Land ownership included placing Indians on reserves to make way for further European settlement.

Aboriginal education during the residential school era, from roughly 1867 to 1945, concentrated on eradicating all traces of First Nations languages, practices and beliefs. The system, controlled by the Federal government and largely administered by churches taught practical skills with "substandard curricula and teaching practices" with no attention given to developing curriculum geared toward language difficulties or sociological needs of students (Fiske, 1981, p.1). Construction of Indigenous education made public access unattainable, excluding First Nations from the greater society while damaging cultural practice. According to Graveline (1998)

Colonial forms of education, particularly residential schools, have contributed greatly to the efforts to eradicate Traditional forms of Aboriginal consciousness. The Ancestral practice of education as integral to daily life and to family and communal relationships shifted radically through colonial pedagogical measures (p.27).

The United Nations report on the situation of human rights and fundamental freedoms of Indigenous people concluded, "Aboriginal peoples in Canada are still trying to overcome the heritage of a colonial education system, which severely disrupted families, their cultures and identities" (quoted in Helin, 2006, p.210). The history of education within First Nations communities since contact has reflected a lack of rights to choose appropriate education options and the lack of power to make decisions on specific issues including facilities, locations, teachers, curriculum and programs⁷. As noted in transcripts from testimony, including the McKenna-McBride Royal Commission Report (1916), First Nations' parents and leaders continually voiced their concerns about

⁷ Steps have been made at a National level, Residential School Truth and Reconciliation Committee and a District level, Aboriginal Education Enhancement Agreements to improve communication with First Nations and reconcile the legacy of residential schooling.

education⁸ and made recommendations for change (Haig- Brown, 1995).

However, the government largely ignored these requests, focusing instead on its agenda of assimilation.

From the first introduction of European models of formal education, First Nations' people saw education as an important tool in their interaction with non-Aboriginal society and attempted to participate in structuring effective education for their children (Haig- Brown, 1995, p.50).

Beginning in 1945, the provincial government introduced an open door policy allowing students to travel off reserve to receive an education in the provincial system. This was a departure from earlier policies of isolation and reflects a government shift in discourse to integration. The federal government, responsible for First Nations, relies on provincial curricula and standards. The provincial education system, designed for white, urban and middle class culture and values was alien to life on reserve. Battiste (2000) suggests that for children whose languages and cultures are different from the mainstream expectations the education system has been a site of "cognitive imperialism" denying First Nations people the right to exercise their own culture (p.193).

Education, or the transmission of knowledge, whether formal or informal in structure, plays a significant role in the perseverance of a community or nation.

Education is the manner in which the worldview, practices, values and knowledge system are transferred to subsequent generations. Education is the

Oncerns included selection of teachers, public boarding or day school selection and curriculum (including the importance of literacy and learning viable skills)

⁹ The Second World War forced re-examination of policies and previous treatment of First Nations as separate and inferior was questioned. Recommendations included First Nations integration into universal social programs. Social programs are provincial responsibility so there was negotiation as to how financial arrangements could be developed to integrate (assimilate) First Nations into provincial education.

principle means of teaching citizens the lessons of membership within a society and thus preparing members for life within a society. Education provides an important component of human development, teaching citizens important knowledge regarding their environment, culture and intellectual heritage (Gutmann 1995, p.60). As noted in the *Royal Commission on Aboriginal Peoples*, education "shapes the language and pathways of thinking, the contours of character and values, the social skills and creative potential of the individual" (1996, Volume 3). The effectiveness of education is measured by how well citizens are prepared for their roles within society and course curricula is culturally constructed, grounded in the worldviews, beliefs and norms of those who conceptualize and teach the curricula (Ball, 2004). Battiste (2000) argues that through the education system, since there in no single culture in Canada, those in power use education to affirm the social and political status quo.

Western society typically uses schooling as a core institution for preparing citizens. ¹⁰ In British Columbia, at the age of five it is compulsory for children to enter the formal education system. While family and other social forces also influence children, from age five until their late teens it is expected that children spend a large portion of their day in classrooms learning curriculum approved by the Ministry of Education. As such, the formal education system is a compelling presence in the lives of all citizens of British Columbia, including First Nations, and is a significant contributor to the manner in which each citizen envisions their connection to others in society, or social imaginary as termed by Taylor (2004). ¹¹

¹⁰ First Nations children in BC are now also required to be enrolled in the BC school system.

¹¹ The concept of Social Imaginary is discussed in greater detail in Chapter 3.

Battiste (2000) indicates, "public schooling has not been benign. It has been used as a means to perpetuate damaging myths about Aboriginal cultures, languages, beliefs and ways of life" (p.194).

1.2 The Current Situation of Aboriginal Education

Despite the historically damaging impact of the school system, Carrier parents reflect the same aspirations as all parents in terms of having their children graduate and go on to university (Holyk & George, 2003). The Lake Babine Community Needs Assessment (2003), designed to determine the social service needs of Lake Babine peoples, sheds light on the fundamental importance of education to the Lake Babine members. High school graduation was very important to 93% of the respondents, 3% said it was important, 1% said somewhat important and 3% did not answer¹² (Holyk & George 2003, p.27). The vision of the members who took part in the survey is for the younger generation to lead the Nation to become independent and self-sufficient. They believe educated youth will be better equipped to help their Nation, including less fortunate members. Participants also saw the importance of post-secondary education as very significant. Eighty-nine percent of respondents said that postsecondary education was very important, 7% said it was important, 1% said it was somewhat important and 2% had no answer (p.28). Respondents felt that continuing to gain education beyond high school would improve the quality of life

¹² This research asked questions of 100 Lake Babine community members, a Carrier First Nation, through structured questionnaires.

for children and all people in the future. They also reflected the need for more professionals in the community and viewed education as the manner to achieve this goal.

While there have been broad changes in First Nations education over the past 30 years, the achievement gap in education between First Nation and other Canadian students still exists. As noted by Ungerleider,

Canadian public schools have failed the majority of Aboriginal students. Their success rates are much lower than those of students from other backgrounds. Significantly fewer Aboriginal students graduate than do non-Aboriginal students and those who graduate have too often been the recipients of an inferior education. Aboriginal students attend post-secondary education at about half the rate of non-Aboriginal students (2003, p.143).

Current education policies fail to realize the parity of First Nation achievement in education compared to other Canadians. The majority of Aboriginal youth do not complete high school and they leave the school system without the requisite skills for employment. As a result, rather than nurturing the individual, the schooling experience typically erodes identity and self-worth (Watt-Cloutier, 2000).

School District No. 91, one of the two districts in which Lake Babine

Nation is situated, is a reflection of the lower success rates experienced by

Aboriginal students. As of the 2005/06 school year, 30% of the total student

population was Aboriginal and 868 students had Indian status and belonged to a

¹³ The Royal Commission on Aboriginal Peoples also notes that students leave without First Nations' language or cultural knowledge of their people.

specific Carrier First Nation (School District No. 91, 2006). As noted with the general trend of Aboriginal education, in 2005 the school completion rate for Aboriginal students, while continuing to improve since 1996, was still low at 43% —compared to non-Aboriginal completion rates of 72% during the same period (p.7). Students of Aboriginal ancestry in the district achieve at lower rates than non-Aboriginal students in Grade 4 and 7 numeracy as well as reading and writing, the areas noted as indicators of achievement for the Ministry of Education. With only a snap shot provided through quantitative statistics there is no analysis as to why this is the case. Additionally, without knowledge of how the parents view assessment and assessment outcomes, it is difficult to determine how meaningful such statistics are to the community and how important the community feels it is to change such indicators.

1.3 Assessment

A reduction in the gap between the general student population and First Nations in areas including literacy, numeracy, and school completion requires action on a number of fronts. Assessment is an important area for further study because information provided in such assessments guides direction of teachers and administrators in their performance plans and can have a direct impact on the community's engagement in educational efforts. There are two main

¹⁴ In British Columbia an Aboriginal student is anyone who self identifies as being of Aboriginal ancestry.

¹⁵ In 1996 the completion rate for Aboriginal students was 21.1%.

¹⁶ 2005 scores for those meeting or exceeding standards: Numeracy Grade 4 (58%), Grade 7 (51%), Writing Grade 4 (74%), Grade 7 (63%), Reading Grade 4 (49%), Grade 7 (37%).

purposes of assessment, assessment on a student level and assessment on a system level (Laitsch, 2005). Teachers and administrators use assessment at a student level to evaluate and improve instruction and student performance. Administrators use system assessment to measure system performance and improve the education system, including educator performance at a school, district and provincial level (Laitsch, 2005). Within this research, I kept the term assessment open to interpretation by parents asking questions of a systemic nature and classroom student level to gain parent perceptions of various types and levels of assessment. Questions included examination of parents' views of the role of school, determining success in school, Foundation Skills Assessment, provincial comparison of students as well as specific questions regarding information received, knowledge of classroom assessment, and strategies used to check student progress.

Student assessment provides feedback regarding student performance in order to inform professionals and parents about the progress of their children and the effectiveness of educational programs and as such teachers use a variety of assessment methods for gathering the information they need to assess, evaluate 17 and report on regarding student learning (Ministry of Education, 2000; Ungerleider, 2003). Achievement is typically gauged through performance indicators that describe and analyze key aspects of schooling. Indicators have three basic functions. First, they assist in evaluating and monitoring student performance by providing an at-a-glance indication (measurement) of student

¹⁷ Assessment is generally used to refer to all measurable activities teachers use to determine student knowledge and skills whereas evaluation is typically used to determine the value of programming.

learning (Hulpia and Valcke, 2004). The second basic function is communication - indicators help to inform stakeholders about the state of student learning and the education system in general. Lastly, indicators provide standardization by illustrating to what extent the situation deviates from established benchmarks, standards or goals (Hulpia and Valcke, 2004). This research provides insight into the importance of assessment functions, particularly communication between parents and teachers.

On a day-to-day basis, teachers use classroom assessment as an information gathering device. Classroom assessment methods include such activities as questioning, homework, projects, lab work, journals, tests and quizzes (Laitsch, 2003). The results of classroom assessment are not generalizeable beyond the context of the classroom but do provide immediate and detailed feedback to teachers and students regarding how effectively students have learned subject matter (Laitsch, 2003). Teachers use each method depending on the purpose and intended impact of the learning opportunity. The range of assessment tools has expanded improving opportunities for communication with stakeholder and parents within the public school system (Ungerleider, 2003). As noted by Laitsch (2005), "educators argue for the use of multiple measures in evaluating students—so that a more complete picture of the student can be generated" (para.8).

A key component of assessment in the modern education system focuses on testing. According to Ungerleider (2003), "teachers use tests to assess student learning and the success of their teaching. Tests are also used to

motivate students to learn prescribed material and to make sure that all of the students in a school or district at a particular grade level have achieved common outcomes considered important" (p.249). Norm-referenced assessments are aimed at determining how well students perform in relation to one another. Criterion- referenced assessments are primarily used to test an individual student's knowledge of information deemed appropriate for the students taking the test (Ungerleider, 2003). Criterion reference tests typically include tests and quizzes performed by teachers in order to determine whether a student has learned desired curriculum.

Volante (2006) affirms the mixing of large and classroom-based assessments as a method of supporting or challenging the limitations of each method. Another advantage of this utilization approach is that it leads to improved student retention rates. ¹⁸ Systems that rely on multiple measures such as portfolios, performance assessment, grades in courses tied to provincial standards, and student exhibitions of learning tend to maintain higher and steadier rates of graduation as opposed to test-only graduation systems (Volante, 2006).

1.4 BC Context

The Provincial government requires schools to report regularly about student progress to parents. School Districts¹⁹ administer public schools under the *School Act (1996)*. The Minister, through the *School Act* directs learning and

15

¹⁸ Volante argues over-emphasis on high test scores as pre-requisites to graduation have the unintended result of pushing students from the school system.

¹⁹ School District No. 91 in particular.

assessment in provincial schools through Orders outlining the responsibilities of schools. For example, the *School Act* defines assessment and provides guidelines for Board responsibilities as well as duties in the administration and completion of assessment. The Ministry of Education also provides a set of standards (success indicators) for reading, writing, numeracy and social responsibility designed to help teachers improve their in class assessment of student work.²⁰ Most jurisdictions, including British Columbia, provide curriculum guidelines for teachers outlining recommended materials, instructional strategies, and methods of assessment (Ungerleider, 2003).

In British Columbia provincial examinations count for 40% of a student's overall mark in examinable Grade 12 subjects (Ungerleider, 2003). For students in British Columbia, graduation requirements begin in Grade 10 with students completing specified courses and writing at least five provincial exams.

Mandatory exam²¹ subjects include English 10, Principles of Math 10,

Applications of Math 10 or Essentials Math 10 and Science 10 (which count for 20% of final mark), Socials 11 or Civic Studies 11 or First Nations 12 (which count for 20% of final mark) and Language Arts 12, including English 12 or Communications 12 (which count for 40% of final mark) (Ministry of Education, 2010b).

²⁰ Performance Standards are broken down by Kindergarten to Grade 3, Grades 4 and 5, Grades 6 to 8 and Grades 8 to 10.

Student progress reports are legislated Under the *School Act* through the Student Progress Report Order. Outlined is the content required for Kindergarten to Grade 3, Grade 4 and 5, Grade 6 and 7, and Grade 8 to 12.

Students have the option of taking examinations related to specific Grade 12 level courses. Full credit may be earned for these courses whether or not the related provincial examination is taken. If students choose to write these exams, results will count for 40% of the final course mark. Many Post-Secondary institutions continue to require students to take exams for all provincially-examinable courses as an entrance requirement.

Important to this research, and described in greater detail in Chapter 2, is the Foundation Skills Assessment, first administered to children in Grade 4. The Foundation Skills Assessment is an annual province-wide assessment of students' academic skills. The skills tested are linked to provincial curriculum and performance standards, designed and delivered by teachers, and administered each year to Grade 4 and 7 students in public and provincially funded independent schools. The main purpose of the assessment is to evaluate how well students are achieving basic skills (criterion referenced) in reading comprehension, writing and numeracy (Ministry of Education, 2008b). According to the Ministry of Education, Foundation Skills Assessment reading scores are a reliable predictor of a student's likelihood of completing school on time (Ministry of Education, 2008b). The Ministry of Education reports results of the annual Foundation Skills Assessment at the provincial, district, school and individual student levels enabling school districts to provide support for students through early intervention and inform teaching strategies for subsequent years and provide feedback to districts for regional and local improvement. Foundation Skills Assessment scores do not count toward student grades.

With lower success rates of First Nations students and recognition at a provincial level to improve the education outcomes of First Nations students, Districts track Aboriginal student progress, including information linking students to particular First Nations' communities. The tracking includes both information from standardized tests such as Foundation Skills Assessment as well as local

measures defined in Aboriginal Education Enhancement Agreements and Local Education Agreements.

Aboriginal Enhancement Agreements, which evolved from a Memorandum of Understanding signed in 1999²², are agreements between a school district, local First Nations, and the Ministry of Education designed to enhance shared decision making and goal setting in order to improve education outcomes of First Nations students (Ministry of Education, 2010a). Agreements also recognize the importance of First Nations culture and language and require the incorporation of the culture and language of the local First Nations where the district is located. The Nechako Lakes School District No. 91 Agreement includes such goals as: improved relations between the district and local Aboriginal communities and parents; improved literacy and numeracy and graduation rates; improved attainment rates of Dogwood diplomas for Aboriginal students and increased student and public awareness of Aboriginal languages and cultures. The performance indicators associated with parent satisfaction with school-home communication, a major component of this research, are addressed in detail in Chapter 5.

1.5 Involving First Nations Parents in their Children's Education

The provincial government has recognized the need for improved First Nations outcomes in the school system throughout the province (Ministry of Education, 2008c). However, the involvement of parents in the assessment

²² The Memorandum of Understanding was signed by the Chiefs Action Committee, the Provincial Education Minister, the Federal Minister of Indian and Northern Affairs and the President of the BC Teachers Federation.

process continues to be an absent component in the Ministry's strategy to improve outcomes. This research analyzed not only parent involvement in the school but also the parent's involvement in their children's schoolwork. Research indicates that family participation in the education of children improves the quality and impact of schooling (Grolnick, Benjet, Kurowski & Apostoleris, 1997; Martinez, Martinez & Perez 2004; UNESCO, 1995). Weiss et al. (2009) indicate after-school learning, summer learning, and family involvement in learning are major predictors of children's development, educational achievement, and school success. Epstein (2001) and Martinez, Martinez and Perez (2004) indicate that families from low socio-economic status benefit most from school-parent partnerships because parents may lack academic knowledge is assisting their children. Brady (1996) suggests that difficulties in school faced by many First Nations' students can be explained by low socio-economic status than any other indicator.

While this is the case, literature suggests First Nation parents' involvement in the school system remains low due to some of the historical issues surrounding parent relationships with schools and districts in the province (School District No. 91, 2006). In addition, the number of First Nations individuals involved in the school system as teachers also remains very low.²³ This research attempted to understand parent involvement in homework and communication levels with their child's teacher. If parents do not understand the tools used or do not agree with the measures, community participation can be difficult to achieve.

²³ At the time of the writing of this document there were two Aboriginal teachers in the district, one of Carrier ancestry.

Research indicates cooperation, shared vision and goals, frequent monitoring, evaluation and positive reinforcement, communication between stakeholders regarding pupil rights and expectations and parental involvement are keys to school effectiveness (Hulpia & Valcke, 2004). The National Indian Brotherhood document *Indian control of Indian Education* (1972) policy statement marked a watershed in Aboriginal education sending a clear, unequivocal call for local control of education by First Nations communities and parents. It highlighted the failure of federal and provincial governments to implement appropriate policies to address First Nations goals for education. One of the key recommendations was local control over education, including First Nations parental involvement in culturally based curriculum. Control, as defined in the National Indian Brotherhood policy is defined as "the right to direct the education of our children. It includes the freedom to choose among many options and alternatives and to make decisions on specific issues" (Haig-Brown, 1995, p.25). Freire (2009) suggests, "pedagogy must be forged with, not for, the oppressed" (p.48).

The ability to choose from options and make decisions is contingent on having information and parents must be involved in determining the education outcomes for their children and communication with teachers and school administrators in order to be informed. Direct involvement also includes parents being a partner in defining education as well as education achievement for their children.²⁴ The UNESCO report (1995) supports the need for parent involvement

²⁴ The principle of this research, the need to define culturally appropriate indicators, is shaped by this position.

stating, "it is unproductive to focus exclusively on school-based programmes when research indicates that family processes and participation in the education of children greatly enhances the impact of schooling and improves its quality" (p.12).

The goals of education that are embodied in assessment strategies are defined by non-Aboriginal authorities, and while some Aboriginal parents and communities may share these goals, it should not be assumed that they will place them above their own goals for the education of their children (or that all First Nations communities support the same goals). According to Pushor and Murphy (2004), the assumption that educators must overcome is the notion that school culture and environment are superior to home culture and environment, particularly in marginalized communities, and that parents' ways of knowing are of lesser value than professional experts.

Carrier and Western systems have different ways of perceiving and creating meaning about the world. Translation and exchange of knowledge requires mutual acknowledgement and acceptance of what the other has to offer (Edwards, 2007). To foster resilience at an individual and community level multiple types of knowledge and knowledge holders are essential. Janet Smylie (2007) argues that each type of evidence requires different approaches and skills for assessment. Developing innovative approaches to assessment will help rebalance the historical and current marginalization of Indigenous knowledge.

Nelson-Barber and Trumbull (2007) posit,

If in fact, the use of local wisdom, recognition of culture, and active involvement of community are mainstays in the established standards of

education practice in Indigenous communities (and contribute to student success), why is this understanding not applied to the realm of assessment? Are we confident that test developers have broad enough understandings of Indigenous students' ways of knowing to enable them to construct appropriate assessments? (p.134)

The inclusion of the voice of parents and determining their knowledge and perspective of the system also relates to another core strategy of both this research and education in the province, creating conditions for members of First Nations communities to become involved in the public schools their children attend. As stated by Ungerleider (2003), "many Aboriginal parents feel uncomfortable approaching teachers and principals about issues affecting their children" (p.146). Past experience, whether due to residential school abuse or experience in the public school where parents had negative experiences of being sent to the office results in avoidance of the school, particularly if they must meet in those same offices because their children are now the ones in trouble (LBN Steering Committee, November 6, 2009). Often parents are unaware of their rights involving the education of their children, such as the right to consult with the principal or to see their children's work and progress. Parents may also be unaware of how tools such as assessment are used by the school and the impacts this has on children.²⁵

The level of education of many First Nations' parents compared to teachers and the culture of colonization is still a significant barrier to communication. Negative experiences of parents within the school system, including residential school experiences and systemic racism experienced in the

-

²⁵ Both in terms of student sense of belonging but also impacts on a student's future. The level of parent knowledge will be examined in this research.

post residential school era, create situations where parents are less likely to become involved or feel empowered to advocate for their children in a system that has historically been demeaning (LBN Steering Committee, November 6, 2009). In addition, parents who have experienced negative school experiences are more likely to accept what teachers and other professionals claim regarding their children without adequate justification, if the teachers consult parents at all.²⁶

As noted by Freire (2009), cultural invasion has resulted in many individuals who consider themselves inferior, which is a necessary tool of oppression. This is the case for many parents who have been influenced negatively for generations within the residential school system or been influenced by systemic racism in a system that has not respected their culture or world-view. A Lake Babine Nation Steering Committee member noted that her cohort was the first group of First Nations students allowed to attend the non–Aboriginal school in Granisle and were victimized daily by students and adults upon entrance (LBN Steering Committee, November 6, 2009). The discrepancy undoubtedly results in difficulty for parents in advocating on behalf of their children in the education system. The inequity of input manifests in a number of ways including lack of parent involvement in meetings with teachers and principals and lack of understanding of their rights as parents (LBN Steering Committee, November 6, 2009).

²⁶ This is an area that will be researched in greater detail in the study.

1.6 Parent Involvement and High Standards

Parent involvement in the development of indicators of success is very important for First Nations communities because how districts determine success will influence the e'ducation students receive including the curriculum topics deemed important. The definition of achievement focuses the attention of schools on specific outcomes in order to improve targets in defined areas. If the indicator of success is completion rates²⁷, the result may be simply reducing the challenge and pushing children through the system in order to meet targets (Watt-Cloutier, 2000). As noted by the provincial government "the key indicator of success will be reducing the discrepancy in high school graduation rates by 2015 between Aboriginal and non-Aboriginal learners (Ministry of Aboriginal Relations and Reconciliation, 2008, p.8). This is an important indicator noted by parents in this research only if it results in graduation with a Dogwood and not a leaving certificate.

International research reflects similar apprehension with parents from American Indian communities alarmed that public schools often assessed and placed their children in special needs programs (Robinson-Zanartu & Majel Dixon, 1996). Kavanagh suggests that too often First Nations' students are labelled as having special needs due to inappropriate testing techniques. Thirty-four percent of respondents, representing 55 Bands in the United States indicated their children had been evaluated for placement in special education (Robinson-Zanartu & Majel Dixon, 1996). While the schools believe alternative

²⁷ High School completion is typically the success indicator used to define how students, and thus school districts are doing.

classes best meet the academic needs of these students by reducing the class size and increasing teaching time, clearly, there is at times a discrepancy between school and parent viewpoints and additional analysis is required. This research adds to this discussion reflecting the need to inform parents of the assessment techniques used and continue communication so that parents are aware of how their children are progressing in alternative or extra classes.

The knowledge of parents in providing direction to their children also relates to provincial exams, as many students are unaware of the impacts of not participating in provincial exams that are not mandatory. As a result, students are not prepared for entrance to university programs because they have avoided subjects required at a university level (P. Michell, Personal Communication February 11, 2010). Similarly, parents who have not graduated high school typically do not understand the implications of their child entering alternative programs and what that will mean for the child in the long term. The use of alternative schools is a contentious issue with districts arguing that intensive remediation and support for students not succeeding in the regular school system will improve outcomes and many Lake Babine Nation parents seeing it as a method of removing children from classes based on ethnicity (LBN Steering Committee, November 6, 2009).

Parents who have not completed their own education are ill equipped to assist children in areas such as homework. This places children at a

Children who have been placed in alternative programs will not have the skills necessary to pass provincial exams necessary for university entrance. This is important from the beginning of the education system because if a child falls behind even in grade one, with the first assessment at grade four it will be very difficult to bring children back to standards required, setting the child up for lower achievement throughout their academic life.

disadvantage systemically and practically. Bourdieu and Passeron (1977), term this concept "social capital" asserting forms of knowledge, skills, education and values provide an individual with advantage or disadvantage based on status within society. The mismatch between home and school language often directly influences First Nations students' achievement in literacy and numeracy placing children at a disadvantage when those skills and knowledge are the focus of testing (Klenowski, 2009). ²⁹

Helin (2006) argues for a focus on educational achievement that is demanding of children in order to reduce the sense of failure that occurs through watering down programs or segregating children. This argument supports the parents of the Lake Babine Nation who call for completion of Grade12 as well as successful completion of core subjects. According to Helin (2006) "given the right environment, it is evident that kids, whether Aboriginal or otherwise, respond to high academic expectations placed on them with results" (p.213). Watt-Cloutier (2000), argues, "the watering down of programs, the lowering of standards and expectations is a form of structural racism that we must make every attempt to stop" (p.115).

In some communities, academic standards and rigour have been lowered in the name of respect for the "different learning styles" of Aboriginal peoples or in order to meet targets set (p.115). As an example, Helin cites the Grandview/?Uuquinak'uuh Elementary School in East Vancouver, which is roughly 50% Aboriginal and demonstrated a considerable improvement on the

²⁹ A one example, in 2005 only 20% of Aboriginal students took the Grade 12 English Exam and 5% participated in the Grade 12 math final exam. This will have implications for university entrance.

Foundation Skills Assessment between 2001 and 2004 by focusing on shared vision for academic success. In 2001 only 22% of Grade 4 students passed the reading test, 63% passed writing and 42% passed numeracy (Helin, 2006, p.212). The school avoided culturally centred curriculum that segregated Aboriginal students focusing instead on creating an atmosphere of success. Some of the key elements denoted in creating a learning environment included high expectations, focus on academics, early literacy programs, regular assessment and strong parent and community support (p.214). By 2004, 88% of Grade 4 students met or exceeded expectations in reading, writing and numeracy (Helin, 2006). It is useful to set high goals for students and ensure students are encouraged to meet those goals. Research supports the East Vancouver example indicating there is a strong correlation between teacher expectations and student achievement (Rampal, Singh & Didyk (1984) cited in Kavanagh, 1998, p.80). In addition, failure to develop literacy and numeracy skills during primary school related to a disproportionate number of Indigenous adolescents not attending school (Mellor & Corrigan, 2004).

1.7 Final Remarks

There is a considerable amount of information available to direct educational policy at a macro-level but community specific research is limited (Hulpia & Valcke, 2004). The assessment tools used, even those of School District No. 91, while providing important information in specific areas, do not reflect the complete picture of community needs. Beyond the narrow scope of the

research, there are philosophical issues in how assessment is conducted when working with First Nations' communities. Research that focuses in-situ, or within a particular culture, is much more appropriate to First Nations conceptions of knowledge (Berkes, 1999). Methods of Indigenous knowledge generation and application are participatory, community based, experiential, and reflective of local geography (Smylie, Martin, Kaplaan-Myrth, Steele, Tait, & Hogg, 2003). First Nations communities have argued against pan-Indian models that do not reflect group differences in the knowledge utilization process. Lafrance (2000) argues that while careers require knowledge provided by the sciences and mathematics, comprehensive education must not displace the values and knowledge of First Nations peoples.

In most school effectiveness research there is an over-reliance on large-scale quantitative research and positivist epistemology at the peril of more detailed qualitative methodologies (Thrupp, 2001). As such, Ministry-derived standards often neglect the social situations of parents, diversity of school aims and community perceptions of achievement. Parent feedback at a community level regarding performance of students, classes and schools is an integral component of assessment that will guide the improvement of processes at a micro (community) level (Hara & Burke 1998). Large- scale testing results in a narrow focus of achievement and defines educational success only in terms that can be quantifiably measured (Morley & Rassool, 1999). The result is that we know First Nations children are achieving at a lower rate than non- Aboriginal children but this information does not explain why this is the case. As noted by

Ungerleider, 2003), "large- scale testing schemes do not by themselves improve the performance of students. Without making sure that a large number of other factors are present, it would be unrealistic and unsound to embark upon such assessment. The use of large- scale assessments to inform changes at the school and classroom level must address the aforementioned obstacles and challenges" (p.263).³⁰

As has been demonstrated throughout the history of interaction between the Canadian governments and First Nations, working in isolation within the perspective that external researchers and policy makers know best without involving those directly affected, is ineffective. Allowing parents control over the development and maintenance of their children's education is an important step for all parties. There remains a lack of understanding of parent's understanding of assessment and the perspective of First Nations parents remains limited within the literature creating an important area for dialogue.

This research, through qualitative methods outlined in Chapter 4, invited parents to address the complex interplay among community, school, class and individual factors affecting what children learn (Ungerleider 2003). In doing so, a secondary goal is to raise awareness of the importance of assessment and strengthen the role of communities and families in children's education. As noted by Martinez, Martinez and Perez (2004), traditionally families have been left out of the assessment process despite parents' awareness of the needs of their children. The research process will create a grass roots approach in which

³⁰ While all schools are required to use the FSA, and report the results at the school and district level, most schools also use a variety of school-wide literacy and numeracy assessments to learn more about how students are doing with regard to specific literacy and numeracy skills.

stakeholders are actively involved in implementing improvement strategies. The research adds to the field of assessment by shedding light on parent knowledge of assessment and involvement in their children's education as well as providing some suggestions for better practice in Aboriginal education. The research contributes to ongoing consideration of partnerships between parents and teachers when developing and administering and analyzing student assessment (Martinez, Martinez & Perez, 2004).

The structure of the dissertation in addressing parents perceptions of education and assessment are as follows. Chapter Two provides a brief historical overview of assessment and the political and social movements during varying eras that have influenced assessment. The discussion provides context for the perceptions of Lake Babine parents regarding assessment. It also assists in explaining how First Nations, as people on the margins, have been excluded from the assessment process and thus their views of what is important for their children to learn have not been included in methods or curriculum.

Chapter Three addresses the colonization process and its impacts on the social imaginary of the Lake Babine Nation. The forming and reforming of Lake Babine social imaginary due to the influence of other societies and the reconciliation of the colonization process provides an important lens for addressing contemporary Lake Babine society and parents' understanding of education and assessment. The multiple social imaginaries present in community are reflected in the research findings in Chapter 5.

Chapter Four provides an explanation of the methods used in the research. The methods exemplify how research should be conducted in First Nations communities and reflects the overall theme of the research that First Nations should be included in all decisions affecting their nations.

Chapter Five presents the findings of the research. Within the chapter, I share the voice of parents by providing their understanding of a number of concepts including definitions of education and assessment. Chapter 6 concludes the dissertation by discussing the key findings of Chapter 5 and providing recommendations and implications.

2: CHAPTER TWO: ASSESSMENT IN HISTORICAL CONTEXT

The transmission of knowledge and the mechanisms for assessing what students have learned (education) can be seen as a cyclical process in virtually all cultures. It begins with the premise, whether conscious or unconscious, of what members need to know and the method for conveying this information, both of which are embedded in the community's worldview. Knowledge transfer also includes an approach for determining if the learner has acquired the information. This process, often termed assessment, determines learning through a variety of strategies from informal observation to formal examination. The word assessment comes from the Latin "assidere," which translates "to sit by one's side" (Wiggins, 1993). The process is cyclical in that the knowledge provider and student continually engage in the process to determine what the student has learned and what information requires further explanation.

The assessment process can be conscious and unconscious, however, for the purposes of this research I focus on the process of measuring, documenting

and reporting student understanding, knowledge and skills attained in British Columbia Ministry of Education delineated curriculum. While a historical overview of assessment could be a dissertation topic on its own, it is important to this discussion to provide a brief historical overview of assessment, particularly the influence of large-scale assessment and the imposition of Western concepts of assessment as it shapes the perceptions of Lake Babine parents regarding assessment and testing.

Historical, social and political systems influence parents understanding and views of the role of education and assessment within the school system. Assessment is inherently a social and political event, as reflected in the historical and contemporary use of assessment tools both internationally and specifically in British Columbia. Members of the Lake Babine Nation are influenced by the larger social debate regarding what children should know and how what they know should be assessed. It is clear from discussion with the Lake Babine Nation Steering Committee guiding this research that the current influence of large-scale assessment has entered their consciousness when talking about assessment. The steering committee wanted the term assessment replaced with test or testing when speaking with parents, as they believed this would be the term parents knew. There was considerable dialogue regarding this point as testing is just one facet of assessment and I did not want to narrow the scope of the question solely to tests, as there are many forms of assessment available. 31 However, testing pervades both Western and Lake Babine cultures alike through mediums

³¹ As seen in chapter 4, parents also use the term test more often than assessment in the interviews supporting the suggestion by the LBN Steering Committee that this is the term parents recognize.

including schools, media and trivia-based leisure activities. There is a general perception that tests indicate achievement and are a legitimate means of determining who succeeds and who fails, and, while some parents challenge the validity of testing, they are still proud when their child improves on a spelling test.

The means of assessing student achievement - including instruments, rationale and formats - have changed over time reflecting broader social and political epochs. Assessment informs societal goals for education in relation to what the government, guided by public and economic pressure, deems important knowledge for students. In terms of assessment, public confidence in the education system has specific consequences for policy. Reporting standard performance measures reassures the public that the government monitors the education system and maintains efficient use of public funds (Volante & Ben Jaafar, 2008). In addition, the government, when challenged by the public regarding education, favours poll sensitive reforms such as standardized testing and shifts in curriculum standards in an attempt to satisfy public confidence (Guppy & Davies, 1999). The easiest goal to implement is directly measurable standards, typically through manipulation of standardized tests (Hillocks, 2002). Linking politics and education further, Hillcocks (2002) indicates, "when the demand for testing is made law, someone has to decide what the tests will examine, what the content of the subject matter is, and what part of that subject is legitimate for testing" (p.16). Just as curriculum development is a political process, determining what will be included in assessment is also political

(Shepard, 1989). Canada attempts to balance the standardization of assessment with the concept of diversity within the nation (Volante & Ben Jaafar, 2008).

In Canada, while each province is responsible for education and thus assessment, approaches to large-scale assessment reflect the global trend that urges educational reforms through accountability measures (Ben Jaafar & Anderson, 2007). In Canada, the historical use of large- scale assessment continues to increase. Every province and territory in Canada, with the exception of Nunavut, administers some form of mandated large-scale assessment (Volante & Ben Jaafar, 2008). A provincially mandated school system means that there are different curricula in each of the educational jurisdictions and subsequently a different criterion-referenced assessment. However, Canada still has a role and attempts to verify its global position within education by participating in various international large- scale assessments. According to Volante and Ben Jafar (2008), the assessment process in Canada embraces a common standard for assessment despite 13 independently governed jurisdictions. The assessment process attempts to satisfy Canadian citizens that action is being taken at a local, provincial, national and international level, preparing students to compete worldwide.

The democratization of education has also changed the role of assessment with public demanding access to education and standardized outcomes for all students. Allowing access to education for all is important within discussions concerning First Nations because it also parallels human rights movements designed to be inclusive of all peoples in accessing services such as

education. For First Nations, this also becomes complex because inclusion has been associated with policies of assimilation imposed by provincial and federal authorities. It was with amendments to the *Indian Act*, in 1951, which coincides with examining human rights on a global scale post World War Two, that the Federal government altered policy of segregated education for First Nations children allowing attendance at public schools. Groups questioned previous treatment of First Nations as separate and inferior and there was a move towards increased assimilation, termed integration.

The present assessment culture in British Columbia is a legacy of early Western societies and practitioners have adopted assessment procedures of the past into the mass education system of the present. Testing developments in British Columbia have been similar to initiatives elsewhere in the western world. Assessment, even in the early stages of Western Education, was a reflection of political and knowledge systems of the day and played a key role in shaping curriculum and determining readiness or selection for occupations. Before the beginning of the twentieth century, many of the motivations and procedures that characterize assessment today were already present (Wilbrink, 1997). Largescale assessments since their inception have typically been used as accountability mechanisms serving broad societal and political goals and largescale assessment is well entrenched in government practice within the province of British Columbia. It represents a key feature of organizational structures and practices of public school over time, well beyond recent recognition and debate (Raptis & Fleming, 2006). Politically the government of British Columbia has

generally been interested in assessment of group learning, largely dependent on large-scale student assessment, for promoting accountability of public schools, rather than the assessment of individual learning, which officials view as the role of the classroom teacher (Raptis & Fleming, 2006). As such, provincial large-scale assessment programs originate in legislation, which sets out the academic areas to be tested and the grade levels for the tests to occur (Hillocks, 2002).

In terms of accountability, large- scale assessment in British Columbia tends to have high stakes consequences for students rather than teachers or administrators (Volante, 2006). Teachers and administrators may feel pressure to improve test scores, particularly when scores are published or schools are ranked, and teachers may be required to develop an improvement plan submitted to administrators but there is rarely official sanction (Volante, 2006). As well, no bonus or merit pay incentives are attached to improved test scores, as is the case in other countries. The high stakes for students are reserved for upper divisions, where assessment results may count as a percentage of the final marks for graduation. In the lower grades, including Grade 4, large-scale assessments do not have direct consequences for students and are intended to be used for improvement purposes rather than passing a grade or graduation.

An area of consistency has been the divide between government-initiated large-scale assessment and teacher-implemented classroom based assessment. Large- scale assessment creates a system where instruction is the responsibility of the teacher and assessment is the responsibility of experts who use elaborate

measures for test development³² (Dochy & McDowell, 1997). The work of experts regarding assessment started largely due to demand for objectivity and fairness in assessment through standardization. It fits with the scientific paradigm of external observers viewing an object from outside as a scientist does using a microscope. However, educators have also used assessment as a political tool, accepting the scientific methods associated with assessment to increase professional designation, particularly beginning in the 1920s, and for challenging methods that threaten autonomy throughout assessment periods.

On a national and international level, examinations are used to assess what students have learned in common areas of curriculum in order to gauge levels of performance nationally and internationally for comparable age groups. Beginning in 1993 the Council of Ministers of Education Canada established a national testing program known as the Student Achievement Indicators Program (SAIP). SAIP consists of tests administered to Canadian students from 13 to 16 years of age in mathematics, reading, writing and science (Ungerleider 2003).

Canada also participates in international assessments of student achievement including the Programme for International Student Assessment (PISA). The PISA assesses the performance of 15 year-old students in over 6000 schools across 32 countries in reading, mathematics and science literacies using common tests (Marks, 2006).

_

³² In recent years the Ministry of Education has involved teachers, as experts, in the development of tests in an attempt to connect teaching and large-scale assessment.

³³ While the majority of national and international examinations are at the post secondary level, it is important to note their use as well as the fact that it is important to prepare students from a young age so that they do not fall behind prior to partaking in provincial, national or international exams.

2.1 Assessment and Indigenous Students

The policies and practices of educational assessment have not met the needs of First Nations students (Nelson-Barber & Trumbull, 2007). Historically, curriculum and assessment has imposed a Western worldview on First Nations people with the implication that the imposed worldview is superior (Battiste, 2000). Klenowski (2009), in discussing the lower performance rates of Australian Indigenous students, argues that improvement is more than an equity issue and must be addressed pedagogically and in terms of assessment practice.

Research from Canada, the United States, New Zealand and Australia affirm the connection between cultural continuity and education outcomes, and call for new assessment practices that link to culturally responsive pedagogy (Gould, 2008; Harris, 2009; Hickling-Hudson & Ahlquist, 2003; Klenowski, 2009; Pewewardy, 2002; Robinson- Zanartu, 1996; Sheilds, 1997; Svihla et al., 2009).

Assessment is a cultural process associated with assumptions about how it should be conducted and how students should participate (Nelson-Barber & Trumbull, 2007). In some cases, assessment policy interferes with attempts to make assessment appropriate for First Nations learners by driving curricular change, rather than measuring learning in an equitable manner (Klenowski, 2009; Nelson-Barber & Trumbull, 2007). Researchers have indicated the negative impacts of curriculum that does not reflect First Nations cultures and the role of assessment in perpetuating such biases (Battiste, 2000; Bordeaux 1995; Klenowski, 2009; Nelson-Barber & Trumbull, 2007; Robinson-Zanartu & Majel-Dixon, 1996; Kavanagh, 1998). Control of history legitimizes the social structure

and the decisions of those who are writing the accounts. From first contact, European writers distributed negative characterizations of First Nations throughout the European continent in order to justify expansionism (Klug & Whitfield, 2003). Writers portrayed First Nations as lacking culture and Christian morals, rationalizing the Western duty to save indigenous peoples. The negative characterization continued through generations entering the curriculum of the school system. Materials used in school reveal that while the majority of events depict the European viewpoint, a large number of historical events were not even recorded, diminishing the role of First Nations in Canadian history. The message sent to children through such materials is that they no longer exist, having no future in current societies (Klug & Whitfield, 2003, p.280). Taylor argues through omission those excluded receive a demeaning picture of themselves as though creativity and excellence is solely a Western male endeavour (1994, p.65). He continues, "enlarging and changing the curriculum is therefore essential not so much in the name of broader culture for everyone as in order to give due recognition to the hitherto excluded" (Taylor, 1994, p.66).

Assessment plays a large role in the lack of integration of culturally appropriate curriculum. Assessment is directly associated with education standards and the determination of school effectiveness, based on narrowly defined core subjects, and plays a large role in promoting teacher focus on largely Western forms of knowledge. Even if there is recognition of the need for inclusion of First Nations culture in curriculum, not including culture as a core of assessment sends a subconscious message to teachers who typically prefer

formula-based methods or rely on principles and formats of large-scale testing when they construct informal assessments (Nelson-Barber & Trumbull, 2007). The culture created under longstanding testing policies and practices communicates to teachers that programs created outside of the standard rubric are less valuable than rigorous coursework of the mainstream educational program, thus limiting their inclusion in the classroom (Klenowski, 2009). Shepard (1989) indicates, "researchers have found ample evidence that testing shapes instruction" (p.5). A statement made by Principal Krause, from BC, reflects the perception of culturally centred curriculum when she suggests curriculum based on Indigenous principles is "long on cultural sensitivity, selfesteem and hugs, but very short on literacy" (quoted in Helin, 2006, p.213). With the pressure to meet targets, schools often view programs related to culture or language as add-ons or supplementary.

Despite improvements in materials and curriculum, educators often do not write texts and design learning environments specifically for First Nations' students resulting in a cultural gap between the home and school. Curriculum guides for public schools now address inclusion of First Nations' culture in materials. However, educators have typically not fully incorporated Indigenous culture in the curriculum nor have they integrated parental involvement into the broader educational decisions made in schools (Robinson-Zanartu & Majel-Dixon, 1996). Shifts in pedagogy must be holistically driven and focused on local wisdom and recognition of culture and worldviews as main components of standards of educational practice in First Nations communities (Nelson-Barber &

Trumbull, 2007; Reyhner, 1992). True integration does not rely on one-time activities, or "add-on" units, it employs materials from the child's culture and history to illustrate principles and concepts (Hilliard, 1991/92).

Determining what will be included in assessment is a political event associated with culture-based assumptions (Nelson-Barber & Trumbull, 2007; Shepard, 1989). Learning and assessment are socially negotiated and embedded within cultural communities that have different views regarding the nature of learning and valued knowledge (Klenowski, 2008). Nelson-Barber and Trumbull (2007) argue "those who are not members of [privileged] groups, and have different experiences and values, will be affected by those assessments developed using the perspectives of those with power" (p.85). Without changes to assessment and curriculum in unison, standardized tests, in effect measure how well First Nations' students learn curriculum deemed by the larger society to be appropriate but may not necessarily reflect parent's views of what their children should know.

Standardized tests go through a consensus building process where teams of curriculum experts and teachers agree to the content of the tests (Shepard, 1989). First Nations are typically under-represented in the rank of expert and teacher and as a result, the learning needs of First Nations people are under-represented in standard tests. In addition, content analysis is completed to ensure test objectives are well matched to widely used textbooks. While educators have made efforts to increase First Nations content in text books, and remove content that is offensive to indigenous peoples, the content is still not

significant enough to represent First Nations, particularly the geographic, linguistic and cultural diversity that exist (Nelson-Barber & Trumbull, 2007).

Procedures to develop standardized tests can also have a homogenizing effect, limiting depth of content in order to reach the breadth of material covered. As a result, cultural content is reduced to a few questions on an overall exam. Similarly, test construction is constrained by emphasis on basic skills needed to move forward, often excluding the full range of important instructional objectives (Shepard, 1989). Stiggins (1997) found that similar issues are reflected in classroom assessment where teachers often use formats of standardized testing when constructing their own informal assessment.

As noted by Nelson- Barber and Trumbull (2007), "given the longstanding detrimental effects of testing policies and practices on the educational outcomes of countless Indigenous students, finding ways to minimize testing biases and reveal students' strengths is a high research priority" (p.132).

2.2 Eras of Assessment in British Columbia

Raptis and Fleming (2006) note that since large- scale assessment of student learning began in 1876 in the province of British Columbia there have been four specific testing eras. The First era of assessment (roughly 1876-1924), began in order to determine who had gained knowledge and skills needed to move beyond primary education. The Victoria High School, which opened in 1876, was the only institution of higher learning in the province. The provincial government required a system of determining who was appropriate for admission

and followed the lead of other Canadian provinces, particularly Ontario, and implemented its first large-scale examinations (Raptis & Fleming, 2006). The high school entrance exam consisted of problem solving and open- ended questions requiring students to use subject matter to make connections and support solutions within the subjects of arithmetic, spelling, grammar and geography. For admission, the evaluation guidelines required at least 30% knowledge attainment in each subject, with exception to grammar, which was determined to be foundational to all further study and thus required 50% (Raptis & Fleming, 2006). The assessment was elitist, using assessment instruments to select limited numbers of individuals to continue their studies and enter professions of teaching, law and medicine. No such measures were created for northern British Columbia or rural communities at this time.

The concerns of educators raised during this period are similar to those noted today including the pressure felt to teach only information contained on exams. The pressure from government and others placed on the assessment results created a system where unprepared students were excluded until teachers were reasonably sure they could pass. Still, the average pass rate over the era was roughly 55% (Raptis & Fleming, 2006). The exams were successful as sorting mechanisms, illustrating what students did not know and used to exclude individuals from high school, and as a result admission to professions, rather than as tool for improved education (Raptis & Fleming, 2006).

As with the social, economic and political pressure influencing assessment in earlier periods around the world, social reformers, community leaders,

politicians and educators during this period of history in British Columbia challenged the structure of assessment and the impact that student inability to pass high school entrance exams would have on the province. Leaders challenged the focus on grammar as an unnecessary barrier to those students who would not require linguistics for future endeavours. By 1911, pressured by employment demands of an expanding economy, the Education Office prepared different exams for rural and urban schools to increase pass rates and in 1918, the Education Office regulations were eased to allow promotion to high school based on recommendations of teachers and principals (Raptis & Fleming, 2006).

The exclusive nature of schooling and assessment shifted dramatically in the mid-1920s when the province's social context and views of schooling changed. The urbanization and industrialization of British Columbia resulted in a second phase of assessment created through demands on schools to prepare increasing numbers of students for trade and commerce. Mandatory school attendance laws³⁴ resulted in increasing numbers of students who needed to be assessed to determine their progress. Instruments used to assess student performance changed with a shift in views of schooling and curriculum, regarded by administrators as the tools for preparing as many students as possible with the knowledge and skills necessary for life within an industrial economy. Mass education was, in part, a response to the requirements for a skilled and literate workforce (Macionis & Gerber, 2005).

³⁴ In 1920 the Canadian government adopted mandatory education to age sixteen (Macionis & Gerber, 2005, p.521).

The second phase of assessment (1925-1972) - influenced by the modernizing of the provincial economy, promotion of greater access to education and higher levels of learning for all citizens - focused primarily on standardized intelligence testing as an objective measure for assigning student placement³⁵ (Raptis & Fleming, 2006). In addition, the department of education introduced standardized, multiple-choice testing used throughout this era in various subjects. The multiple-choice format used during this era differed from earlier methods, which focused on open-ended essay style questions.

Reflecting the social and political nature of testing, particular subjects of study have dominated the testing agenda of specific eras, determined to be indicators of the key learning needed by students at specific times. For example, in British Columbia from 1925 to 1972 the majority of government-administered tests were multiple-choice with mathematics featured prominently. Since 1976, literacy and numeracy have dominated large- scale testing as the foundation to learning in other subject areas (Raptis & Fleming, 2006).

2.3 Intelligence Testing

Intelligence testing developed during the 1925-1972³⁶ period was highly political and created great mistrust of testing in minority populations. The

_

³⁵ It should be mentioned again that First Nations were exempted from public schools until 1951 and at this time were only allowed access when numbers resulted in the need for increased enrollment.

³⁶ In 1905 Alfred Binet and Theodore Simon announced the development of an objective measure capable of diagnosing different degrees of mental retardation. In 1916 Stanford psychologist Lewis Terman released revision of Binet's work that became the Stanford-Binet test (Becker, 2003). The tests have been used since with the fifth revision occurring in 2003 (Becker, 2003).

American Psychological Association, in response to the debate regarding intelligence testing during this period suggested that "research findings were often assessed not so much on their merits or their scientific standing as on their supposed political implications" (Neisser et. al, 1996, p.77). The evolution of intelligence testing, while turbulent, must be considered because much of the theory of standardized assessments evolved from work done on standardized intelligence testing (Laitsch, 2005). Historically, institutions used intelligence tests to establish the inferiority of non-white peoples, largely based on cultural biases contained within the testing measures (Gould, 2008). Intelligence testing in the education system often resulted in placing immigrant and marginalized ethnic groups into low-level courses and vocational schools arguing they had low mental abilities (Laitsch, 2005).

Intelligence testing has been used in other controversial ways including the eugenics social movement of the mid twentieth century (Depew, 2010; Laitsch, 2005; Leyva, 2009; McLaren, 1990; Roman et. al, 2009). Proponents of the eugenics movement attached themselves to intelligence testing as a scientific foundation used to identify individuals, including First Nations, who could be defined as mentally incompetent on the basis of genetics (Roman et al., 2009). Proponents of eugenics then used the information gathered from intelligence testing as justification for the institutionalization of individuals or placing them on basic skill tracks. During this period many countries, including Canada, enacted eugenics policies and programs including genetic screening, promotion of racial and genetic segregation and sterilization based on the eugenics movement. Law

in two Canadian provinces, BC and Alberta, enforced eugenics. Alberta enacted a Sexual Sterilization Act in 1928, creating a Eugenics Board with the power to authorize the sexual sterilization of individuals. BC enacted similar legislation in 1933 and both provinces had the laws in effect until 1972 (McLaren, 1990). In Alberta, at least 2822 individuals were officially sterilized while in BC the number is unclear as records of the program were destroyed (McLaren, 1990).

Opponents of the early use of intelligence testing argue that its use in assessment failed to account for factors including cultural bias, language and socio-economic status when attempting to determine cognitive competence (Gould, 2008; Klenowski, 2009; Turkheimer et al, 2003). Scholars opposed to the intelligence testing have argued intelligence testing to be the basis for scientific racism against marginalized groups (Croizet & Dutrevis, 2004; Cronshaw et al., 2006). Gould (1996) states intelligence-testing supports,

the abstraction of intelligence as a single entity, its location within the brain, its quantification as one number for each individual, and the use of these numbers to rank people in a single series of worthiness, invariably to find that oppressed and disadvantaged groups—races, classes, or sexes—are innately inferior and deserve their status (p. 24).

The importance placed on scientific measurement in British Columbia shares similar support with the rise of modern states in nineteenth century Europe. State influence, particularly the bureaucratization of states, was critical in assessment shifts all over Europe. Where earlier family background and wealth were decisive in obtaining attractive government positions, merit was becoming a key criterion. Assessment, as noted by Wilbrink (1997), played a large role in occupational acquisition as, "examinations were the main way of

becoming a civil servant" (p.42). With bureaucratic states basing systems on meritocracy, the marking system shifted to reflect changes and objective assessment based on competitive examinations gained favour. In addition, the curricula itself was changed as objectivity demanded curriculum content be narrowed and adjusted to make objective assessment feasible (Wilbrink, 1997).

In British Columbia, various groups supported standardized assessment instruments for social and professional reasons. Educators supported the scientific approach of standardized testing as a means of raising professional status by linking education to scientific enquiry. Educators and researchers viewed the adoption of a program of standardized testing as a more accurate and objective assessment of students' academic abilities, ranking standardized tests as the pinnacle in the hierarchy of tests. Administrators also supported standardized intelligence testing as an objective means for ensuring teachers followed curriculum at all grade levels. In addition, the use of standardized tests allowed for comparison between schools, increasing public awareness of testing results and challenging performance based on tests rather than other social factors. Governments provided data regarding provincial and district averages so that educators and the public could compare their school's results with other districts, including rural and urban schools as well as schools in other countries using standardized exams (Raptis & Fleming, 2006).

The idea of objective measurement is reflected in teaching and learning models of the era. Science attempts to understand the entire universe by reducing all of the mysteries of the universe into a finite set of laws (Knudtson

and Suzuki 1992). It is the responsibility of educators to teach students those laws in an objective manner. In mainstream education, proponents view culture as something separate from learning and from relationships. This attitude is the result of the Western reductionist worldview, which tends to compartmentalize and categorize life and institutions including learning (Robinson-Zanartu & Majel-Dixon, 1996). The principle of philosophical realism premised on the argument there are universal truths outside of individuals and independent of their thoughts that individuals can study empirically, pervades the school system. The teacher is an objective individual responsible for presenting facts.

The dominance of normative tests reflects the form of subject matter and teacher's beliefs regarding the nature of evidence and principles of fairness (Shepard, 2000). Standardized assessment methods are consistent with theoretical frameworks closely aligned to scientific measurement; assessment is an official event separate from instruction. The methods of assessment include recall, completion, matching and multiple choice questions designed to measure specified ideas within curriculum through formula-based methods designed to be objective (Shepard, 2000). Traditional views of testing, curriculum and teaching reflect the view that learning occurs by accumulating bits of knowledge and precise standards of measurement are required to ensure mastery of each skill at the desired level. Teaching and learning operate under the same model as the factory assembly line. Learning is sequenced and hierarchical and tests are frequent to ensure mastery prior to moving on to the next objective (Shepard, 2000). In this method, schools require students to learn pieces of overall

concepts, indoctrinating students with facts and memorization so they are prepared for the future (Klug & Whitfield, 2003).

By 1972, the provincial government phased out standardized intelligence testing, again largely as a result of changing social and political views including heightened calls for social equality (Raptis & Fleming, 2006). The debate regarding large- scale assessment has gained momentum since the 1970s resulting in unprecedented calls for reform and accountability within public schools (Berliner & Biddle, 1995). In addition, social movements of the 1960s and 1970s challenged the validity of large-scale measures of student achievement and the assessment of individual students as focused on Whitemiddle class values. Provincial governments were cautious regarding individual rights and reported only aggregated data designed to revise education practice. The provincial government shifted attention from individual students to the government itself through a system of program evaluations designed to measure effectiveness of government programs and services, including education. The focus of assessment became informing the public about student achievement and assisting teachers and curriculum developers in improving courses through curriculum design, professional development and resource allocation. As one example, the 1976 language arts assessment resulted in one-hundred recommendations related to curriculum development and teaching strategies (Raptis & Fleming, 2006, p.1202).

The major assessment model of this era, initiated by the new Education Minister in 1975³⁷, was the Provincial Learning Assessment Program (Raptis & Fleming, 2006). The purposes of the program were to "evaluate student progress" over time, account to the public for strengths and weaknesses of the K-12 system, provide individual districts and schools with performance data, support curriculum change and educational research, and provide management information for resource allocation" (Raptis & Fleming, 2006, p.1202). The program was in all provincial and public funded independent schools and with limited exceptions, all students in Grades 4, 7 and 10 were expected to participate (Ministry of Education, 2008a). Based on the provincial curriculum, panels made up of 60 teachers, 20 from each grade category, determined what knowledge and skill level students from various grades were expected to obtain (Ministry of Education, 2008a). The children were assessed relative to provincial standards and expectations set by the panel. With the exception of reading and writing, the Provincial Learning Assessment Program focused primarily on multiple-choice as the most frequently used format to determine subject knowledge.

2.4 Foundation Skills Assessment

In 1999 parental pressure and requests from numerous district administrators resulted in the government changing the provincial testing program to its current form, the Foundation Skills Assessment (FSA). As with the

_

³⁷ The Social Credit Party was elected in 1975.

Provincial Learning Assessment Program (which was not officially supplanted by FSA but has not been used since FSA's inception), the Foundation Skills Assessment is an annual province-wide assessment of students' academic skills. The skills tested are linked to provincial curriculum and performance standards. The Foundation Skills Assessment narrows the scope from previous assessment programs, focusing instead on reading comprehension, writing and numeracy as areas requisite to all other learning (Ministry of Education 2008b). The assessment, designed and delivered by teachers, is administered each year to Grade 4 and 7 students in public and provincially funded independent schools. The Ministry of Education reports results of the annual FSA at the provincial, district, school and individual student levels. Parents receive their child's individual FSA results indicating whether their child does not yet meet expectations, meets expectations or exceeds expectations in the three core areas. The Foundation Skills Assessment also breaks out data on First Nations' students at a school, district and provincial level.

The Foundation Skills Assessment remains highly politicized and contested by various groups. Educators continue to resist the tests as measures of educational quality (Volante, 2006). The BC Teachers Federation does not support the assessment method arguing that it is used solely to rank schools when money could be better spent elsewhere (BC Teachers Federation, 2008). The Federation encourages teachers not to assist in distributing or marking the student assessments. Critics of large-scale assessments such as the Foundation Skills Assessment argue the tests hold teachers responsible for such results

when results are dependent on other external factors, provide information that does not translate well to improving classroom practice and damages student self-concept leading to disengagement (Darling-Hammond, 2004; Meaghan & Casas, 1995; Stiggins, 2002). Critics suggest that the tests widen the gap between groups such as First Nations because they do not account for factors known to affect student performance such as socio-economic status, language of origin, and physical and emotional health (Valencia & Villarreal, 2003). Volante (2006) argues that more attention should be given to communication strategies that report the results including reporting limits of the test and avoiding ranking systems that promote misinformation or inappropriate competition (Volante, 2006). Other unintended consequences of improper reporting of results can include the possibility of narrowing curriculum by guiding what knowledge is measured, and by extension taught. This often means teachers may not emphasize subjects related to First Nation's culture that are supplementary to other subjects in the curriculum and thus not included in testing at the same levels.

One of the reasons for rejection of the Foundation Skills Assessment by the BC Teachers Federation is the manner in which organizations such as the Fraser Institute present the information. The Fraser Institute uses a number of indicators, based on a school's average scaled FSA performance to rank the performance of individual schools (Cowley & Easton, 2007). Reflecting the interconnection of politics and assessment, the Fraser Institute, a right wing political organization, uses assessment measures in an agenda noted as an

attempt to discredit public education in order to privatize the education system (Cowley, July 17, 2007). As noted by Guppy and Davies (1999), "teacher's unions have been reluctant to admit that confidence [in education systems] has declined. Instead, they have pointed to "special interest groups," often the New Right and religious groups, as the sole doubters of the merits of public education" (p.278). In the *Report Card On Aboriginal Education in BC, 2006 Edition*, Cowley and Easton (2006) argue that Aboriginal educational performance compared to other Canadians can only improve if "Aboriginal Students' academic achievement is regularly assessed and school-by-school results are publicly reported" (p.3).

Public debate has resulted in calls for the increased use of other assessment methods to take precedent over large-scale assessment.

Assessment practices in British Columbia are changing and teachers recognize the need for alternatives, or enhancement, to standardized testing methods. In response, the Ministry of Education has created an assessment handbook series devoted to student self-assessment, student centred conferences, performance assessment and portfolio assessment (Ministry of Education, 1994). The range of assessments, including notification of risk of failure, conferences, performance based and standardized tests and written comments have expanded assessment tools used in the public school system (Ungerleider, 2003). However, assessment tools that carry the most weight are the standardized measures that are typically publicized and that schools use to track performance.

2.5 Final Remarks

Parents perceptions of assessment are influenced by the images portrayed in the media and through political debate. As noted in this chapter, assessment is much larger than what takes place in the classroom, and is influenced by historical, social and political systems. As members of those systems, parents are active participants in assessment both acting to create and reacting to the information provided regarding assessment. The Lake Babine Nation is not exempt from the influence of philosophies of objectivity that permeate the climate of assessment and the methods used. The focus on large- scale assessment in this chapter is not to suggest that it is the only process of assessment, it is discussed as a means to demonstrate how social and political systems influence how we determine the best methods of assessing students. Parents' perceptions of assessment, including large- scale assessment, will be explored in greater detail in Chapter 5.

3: CHAPTER THREE: THE LAKE BABINE NATION: WORLDVIEW AND EDUCATION

The role education plays in the lives of members of the Lake Babine

Nation has changed significantly with the introduction of Western economic,
cultural and social systems. The involvement of the Lake Babine peoples with
settlers and the Canadian state in the past two hundred years, and relationships
fostered continue to affect the worldview of the Lake Babine peoples. The
concept of social imaginary, as termed by Taylor (2004), is very useful in
examining the changing worldview of the Lake Babine peoples and is important
to this discussion in terms of what Freire (2009) terms the "duality of the
oppressed" (p.55). Freire argues that often the oppressed "are contradictory,
divided beings shaped by and existing in a concrete situation of oppression"
(p.55). The more those invaded are alienated from the spirit of their own culture
and from themselves the more they take on the characteristics of the oppressor
and reflect the duality of their own and invading cultures. The social imaginary is

³⁸

³⁸ Earliest European residency in the Lake Babine Region is noted as circa 1822 (Fiske, 2000, p.36).

their connection to others within the society (Taylor, 2004, p.23). The exploration of this notion is important to later discussion regarding parents' conceptions of assessment because social imaginary guides understanding and the present Lake Babine Nation worldview cannot be understood without explanation of the colonial history that has affected the nation. The colonial culture, and its oppressive nature, is embedded in the education system. To understand how education in Canada continues to suppress Lake Babine consciousness I must address the shaping of modern social imaginary.

This chapter will contribute to the dissertation by addressing the Lake Babine Nation social imaginary through a discussion of the spiritual, economic and political spheres present in contemporary Lake Babine moral order as a tool for examining parents' understanding of education and assessment. Freire (2009) argues "societies which are dual...invaded, and dependent...cannot develop because they are alienated; their political, economic and cultural decision-making power is located outside of themselves, in the invader society" (p.161). In keeping with the premise of this research, that quality research with First Nations must empower individuals and provide tools for resistance by increasing understanding of the impacts of colonization, the discussion provides an opportunity to address the historical influences on the Lake Babine Nation and their role in shaping the Nation's modern social imaginary. Battiste (2000) states, "Aboriginal consciousness cannot be maintained without first challenging the assumptions of modern society" (p.193). It is a useful tool in understanding the

establishment of a philosophical shift from community to individual focus and the issues created when the shift in social imaginary remains in flux all of which shapes how parents view the concept of education and the information it is important for their children to learn.

Indigenous social imaginaries, including the Lake Babine Nation, are unique and require analysis because they diverge from the Western social imaginary largely because of the different historical circumstances of colonization that continue to shape their worldview (Dussel, 1996, 2002). The understanding of colonization and its impacts on the oppressed builds on the premise of this research to be a tool of resistance, empowering the Lake Babine Nation. As Freire (2009) indicates "the resulting comprehension tends to be increasingly critical and thus constantly less alienated" (p.81). Divergent worldviews play a role in the manner in which parents view education and the education system in general. Western society has forcefully placed notions of a just society over Indigenous spiritual, political, economic and cultural systems already in existence at the time of contact and while Indigenous nations have incorporated some aspects, others are continually challenged because they remain too divergent from Indigenous philosophy. The *Indian Act* (1876), the main piece of legislation governing Canadian First Nations' people for the past 130 years, specifically targets two areas of Indigenous society noted by Taylor (2004) as crucial to modernity, economics (land ownership) and governance. This chapter examines the historical impacts of colonization on Lake Babine systems, particularly church, state and industrial alienation of the Nation from their economic, political,

spiritual and cultural systems in the twentieth century, and the attempts by the Lake Babine Nation to reclaim their own social imaginary within a pluralistic society.

3.1 The Lake Babine Nation

The people of the Lake Babine Nation, *Nedut'en*, have always lived along Lake Babine. The lake extends approximately 177 kilometres in north-central British Columbia (Fiske & Patrick, 2000). Prior to 1939, many members of the Nation lived in the village of Old Fort, across the lake from the present village of Granisle and the Lake Babine Nation community of Tachet. In the 1940s, the first roads were built within Lake Babine territory and a number of families moved to Topley Landing and Decker Lake where men worked on the railway. In 1956, the Department of Indian Affairs moved the majority of members out of Fort Babine and Pendelton Bay into Burns Lake in order to reduce the travel of officials, improve housing and other services, and enable children to attend mission schools. Prior to this period, many children missed school for part of the year assisting their fathers with hunting and trapping (Fiske & Patrick, 2000).

The Lake Babine Nation currently has three communities inhabited year round, only two of which are within their traditional territory (INAC, 2008). The Lake Babine Nation comprises 2284 members and of this total, roughly 1500 people live on reserve while the remaining individuals reside in surrounding communities as well as in distant cities (INAC 2008). The largest inhabited community and administrative centre of the Lake Babine Nation, known as

Woyenne, is located 220 km west of Prince George within the Village of Burns Lake. Children in Burns Lake typically attend public school at Muriel Mould Primary School (Kindergarten to Grade 3), Decker Lake Elementary or William Konkin Elementary School (Grade 4 to 7). The community of Tachet is isolated, situated 12 kilometres from the community of Granisle, 100 kilometres east of Burns Lake. Children from the community of 130 permanent residents attend Babine Elementary- Secondary School located in Granisle³⁹. Fort Babine is located at the northern tip of Babine Lake, approximately 100 kilometres north of Smithers. It is a small community with 60 year-round residents. Children attend a band run school in Fort Babine until Grade 7 at which time they travel to Smithers.

The Lake Babine Peoples, who are part of the larger Carrier Nation, speak a distinctive dialect of the Athapaskan language group (Fiske & Patrick, 2000). Lake Babine society is matrilineal, although men and women share important roles in the community. For example, both men and women inherit and are groomed to be healers as well as hereditary leaders (Holyk, Shawana & Adam, 2005). Lands and resources were the responsibility of hereditary leaders who effectively managed the resources within the clan system. The Lake Babine peoples relied, and continue to rely, specifically on the salmon harvest each summer and throughout the year supplemented their diet with large and small game and fresh water fish taken from the territory. 40

_

³⁹ At times there is a band run school which children attend but it was not open when the research was being completed.

⁴⁰ As noted by Fiske (2000), "changes in living patterns and resource harvesting have been continuous since the earliest colonial period" (p.34).

Lake Babine people maintain a governance system commonly referred to as the *bah'lats* or potlatch. Four primary clans make up Lake Babine society *Likh ji bu, Gilhanten, Jihl tse yu* and *Likh sta Mis yu* (Bear, Caribou, Frog, and Beaver respectively), each with several sub-clans. Generally, there is one head clansman for each primary clan and a hereditary chief who holds rank in the *bah'lats* as a wing chief represents each sub clan. These positions are passed down through family or clan lineages as well as the result of selections guided by clan Elders. The role of the head clansmen is varied and includes being main spokesperson for the clan they represent, looking after people's welfare, and providing direction to clan members.

The bah'lats, organized around the clan system, is the core economic, political, social, legal and spiritual institution of the Lake Babine peoples. While protocol is flexible and adaptive to the differing practices of each community, the guiding principles of the bah'lats system are shared among the communities. The bah'lats is inclusive of all members of the nation and all formal business in the bah'lats is conducted in an open and transparent environment where clan members (whether hereditary chiefs or not) are witnesses to transactions that occur. As witnesses, individuals are expected to commit to memory the details of transactions, and in the case of hereditary chiefs to recount through oral histories the transactions at future feasts when those transactions are relevant. The bah'lats brings together members of the community for major events such as birth, coming of age, taking of hereditary names, and death. It introduces children to the beliefs and workings of Lake Babine society and connects people to the

land. With its central role in all aspects of life, the bah'lats is a key component in fostering social obligation and civic responsibility.

3.2 Social Imaginary

In order to discuss the development of the present Lake Babine social imaginary, an explanation of Taylor's characterization of the modern social imaginary is needed. For Taylor (2004), the social imaginary is the manner in which people imagine their social existence and conceptualize their connection to others within the society. The social imaginary is the common understanding that is shared by the majority of citizens, guiding common practices and unconsciously reinforcing those practices as legitimate. The social imaginary provides individuals of a particular society with a "map of social space" that makes interaction between people possible, predictable, and understandable (Taylor, 2004, p.25). Taylor (2004) claims that the social imaginary "incorporates a sense of the normal expectations we have of each other, the kind of common understanding that enables us to carry out the collective practices that make up our social life" (p.24). The understanding of the group is both factual, in terms of the explicit rules of how things should function, and normative, in terms of unconscious ideal of what action ought to be. Within the moral order created, the group has a sense of not only what is right but also the context of what makes these norms realizable and in turn worth striving for (Taylor, 2004). The moral order conveys how individuals ought to live together in society and the social imaginary assists in providing individuals with the internal framework that make

society seem just. The social imaginary is what enables the practices of society through the common understanding of the moral order.

The social imaginary, which is socially imbedded, meaning those within a society cannot imagine themselves outside of a certain matrix, can be shifted as new theories and knowledge systems are introduced. Such a phenomenon is what Taylor describes as the modern social order, which is based on individualism as opposed to community-centred organization. According to Taylor (2004), our first self-understanding was deeply imbedded in a social construct; our essential identity was as a father, a son, and a member of a clan, community and nation. It is a recent phenomenon that we consider ourselves as individuals within a community and this process involved a profound change in our moral world. In Western Society the self-understanding of identity and one's role within society, which today offers unprecedented primacy to the individual is what Taylor (2004) terms the "Great disembedding" (p.50). According to Taylor, the collective nature of community and the internal and unconscious connection between each person, the community, and the cosmos has been displaced by an understanding of society as made up of individuals who are part of a community. Each individual has relationships to the community as a whole and to personal devotion in which religion is separated from the developing secular world of the economy, public space, and the self-governing individual.

Lake Babine social imaginary continues to adapt, experiencing an everincreasing adoption of certain concepts such individualism that reflect changes in economic and governance structure in community. However, the process is not entirely realized.⁴¹ As with society as a whole, the more the Lake Babine are alienated from the spirit of their own culture, market economics continue to seep into the Lake Babine system and disrupt values including the concept of community and the guiding principle of reciprocity. The introduction of private interest is beginning to take precedent over common good and has the potential to depose the cultural, spiritual and political traditions the Lake Babine society is built upon⁴².

In traditional Lake Babine social systems an individual acts not necessarily to maximize her own potential but as a public minded member of a community (Denis, 1997). As Taylor (2004) notes, if individuals imagine their social existence as actions on behalf of the whole society, which is the case for Lake Babine philosophy as well as economic and political systems, the social imaginary informs the individual's sense of self in relation to others. The philosophical basis of Lake Babine social order is that people are connected to each other and their environment. This connection results in obligations to maintain balance within society beyond individual interests such that if a person is balanced he/she is in a position to fulfil his/her responsibilities to the whole.

_

⁴¹ This is not to suggest that concepts of individualism did not exist prior to contact as noted by Shawana in his discussion of private and public knowledge among the Carrier Peoples (Shawana 2007). It is too simplistic to suggest communitarian notions are the sole concept and may suggest why certain aspects of Western systems have been adopted while others are still challenged.

⁴² The education system reinforces Western values and practices that at times are opposed to Carrier philosophy. For example, the competitive hierarchical structure of schools, and assessment that emphasizes individualism, is foreign to Carrier values and a direct threat to the importance of community (Nelson-Barber & Trumbull, 2009). The use of competition as opposed to cooperative learning models, place students at a disadvantage by forcing them to learn in an environment that challenges their worldview. The value of community is discussed by parents in greater detail in chapter 5.

Maintaining group balance is in the best interest of the individual, so community and individual interests are one in the same. In Lake Babine culture, everyone has a role in the community and makes decisions to provide maximum benefit to the group. For example, when a community member is experiencing problems, setting the problem right is a concern of the whole community and community supports, such as Medicine people and Hereditary Chiefs, are responsible for restoring balance on behalf of the entire community (Denis, 1997). Particularly in times of crisis the community works together in order to support those requiring assistance. For example, mechanisms such as the *bah'lats* ensure the redistribution of goods when people are in need (such as when a family member passes away).

While the philosophical base of Lake Babine culture is still intact, it is interesting to trace shifts in Western social imaginary and the manner in which those in the West began to see European philosophical traditions as the pinnacle achievement of society. The developmental shifts in the Western social imaginary foreshadow the manner in which colonizers viewed and treated First Nations upon contact, as well as how Lake Babine society continues to address living within dual worldviews. The transformation of Western social imaginary, including the Renaissance notion of civility in Europe, drives the assumption that "those who lack excellences, refinements and the importance of achievements valued in western society" are savage and in need of refinement (Taylor 2004,

⁴³ While I speak of two worldviews this is not meant to be entirely dichotomous as this research suggests no worldview is static or universal leaving room for internal disagreements and shifts as new conceptions arise.

p.35)⁴⁴. Roughly six centuries ago a number of reform movements in Western Europe led to some significant changes in society and in the social imaginary of European societies. The notion of civility, reflecting the transition that European societies were undergoing beginning in 1400, was seen as the result of taming of the originally wild, raw nature. According to Taylor (2004), this process continued for roughly four centuries until by 1800 a normal civilized society was determined to be one that ensured continued domestic peace, where commerce had replaced war as the paramount activity of the state.

While an extensive discussion of European transformation is beyond the scope of this chapter, it is important to mention a number of systems of control including religious reform, indoctrination in ethics of industriousness, production and moral refinement because they are particularly analogous to views that were later imposed on Indigenous peoples of North America (Taylor, 2004, p.44). In Europe, the push for civility in all spheres meant imposing features of the ideal of civility on an increasing segment of the population in order to provide a self-understanding of sociability, ideals and the training required to fulfil their roles. Examples of moral refinement noted by Taylor (2004) are particularly enlightening regarding the treatment of First Nations centuries later. For example, Erasmus condemned the Carnival he saw in Siena in 1509 as "unchristian" on the grounds it contained traces of ancient paganism and the over indulgence of individuals (the same argument made against the *Bah'lats*) (Taylor, 2004, p.43). Governments and church authorities came down hard, attempting to

⁴⁴ The assumption of cultural superiority of European worldviews are built into the current education policy. Curriculum projects European knowledge as universal, normative and supreme (Battiste, 2000, p.193).

eradicate such public displays. In Lake Babine territory, Oblates emphasized the repression of undesired behaviour through close supervision and harsh public punishments including whipping, confinement and public confession. Father Morice, a Catholic Priest who served in Carrier territory from 1885 until 1905, insisted on public confession and denied confession to anyone who attended a *bah'lats*. Morice also refused to visit the community of Fort Babine on various occasions because members were known to have participated in the *bah'lats* (Fiske & Patrick, 2000).

In much the same manner as the "rod of Circe" was used in Europe to "tameth both man and beast that are touched therewith, whereby each one is brought in awe and due obedience where before they were all fierce and unruly," upon settlement in North America First Nations people were believed to be in need of discipline in order to make them human according to Western conceptions of civilization (Justus Lipsius as cited in Taylor 2004, p.38). Colonizers did not see First Nations, such as the Lake Babine Nation, as being distinct culturally and politically but rather as connected to nature and lacking civility. Western doctrine was premised on the notion that in order to become complete citizens individuals must be trained, disciplined and formed. It was believed this training was based on severe discipline in order to tame raw nature. As Taylor (2004) notes, "the child embodies the "natural" condition of lawlessness and has to be made over" (p.39). These new forms of discipline include moral, economic, religious and political intervention.

In order to intervene, Taylor (2004) indicates, "reforming states saw religion as a very good source of discipline and churches as handy instruments" (p.39). In North America, the churches also saw this as a beneficial partnership because it opened access to society and individuals, particularly children, deemed in need of intervention and conversion. The First Priest to visit Carrier territory was Father Demers who travelled to Stoney Creek and Fort St. James in 1842 (Munro, 1945). Father John Nobili entered Lake Babine territory three years later bringing the Gospel message. As one example of the attempted conversion to Western systems, Father Nobili endeavoured to abolish the Carrier custom of cremating their dead (Munro, 1945).

Spiritual recovery and rescue of civil order were intrinsically linked and church officials argued First Nations were "handicapped by the lack of effective discipline in economic, moral and socially helpful habits" (Kelm, 1998, p.63). One important aspect of the strategy for discipline was targeted at First Nations' children. Education, and the residential school was seen as a means of producing cultural conformity designed to produce civilized individuals well suited for their particular social roles (Kelm, 1998). Residential school officials were convinced that strict obedience to a rigid routine would produce disciplined workers and assimilated citizens. Lejac Residential School opened in Carrier territory in 1922 and remained an instrument of attempted assimilation for roughly 50 years (Fiske, 1981).

The Federal government joined forces with the church to Christianize and civilize First Nations children by supplying funding as well as creating legislation

designed to make attendance at residential schools compulsory (Kelm, 1998). A report by Nicolas Davin, which was commissioned by the Federal government after the enactment of the *Indian Act* and led to the establishment of the residential schools, provides insight into the position of the government. Davin strongly recommended residential schooling suggesting that First Nations' children could be best educated apart from their parents and bush life placing them in institutions where values, language and culture of the Western nations would predominate (Kelm, 1998). Focusing on the child and his embodiment of the natural Davin indicated, "if anything is to be done with the Indian, we must catch him very young" (as cited in Haig-Brown, 1995, p.51). The Canadian government removed children from their homes and communities and subjected them to pedagogy designed to demean their culture and provide discourse regarding the superiority of Western worldviews (Kelm, 1998).

Administrators, attempting to shift the worldview of First Nations children, did not allow for open dialogue within Lake Babine society. Administrators only allowed parents to visit with their children for two and one half hours per week while their children attended residential school at Lejac and all visits were supervised (Fiske, 1981). Visits were also restricted solely to parents or guardians further challenging the role of extended family and elders in the translation of knowledge. The denial of extended families and communities resulted in a rupture in the connection between youth and knowledge holders.

The heart of any society, children, were forcibly removed from the community. Without support from extended families and community, the physical,

emotional, sexual and cultural abuses that occurred required children to turn inward, and resulted in dependence on self. The abuse resulted in a survivor mentality where individuals retreated from the public sphere and avoided sharing culture and language. The lack of cultural understanding among younger community members is considered a critical area of concern and has significant implications for the development of Lake Babine social imaginary. Elders believe the younger generations do not understand or practice their culture while the younger generation expresses a willingness to learn and is disappointed at not being taught such important aspects as the language and worldview.

3.3 Spirituality and Worldview

The fundamental feature of Lake Babine philosophy was, and continues to be, that all of life is a manifestation of spirituality. As noted in the Royal Commission on Aboriginal Peoples (1996) speaking more generically regarding First Nations philosophy but equally reflecting of Lake Babine worldview, "we come from spirit; we live and move surrounded by spirit; and when we leave this life we return to the spirit world. All perceptions are conditioned by spiritual forces, and all actions have repercussions in spiritual reality" (as cited in Denis, 1997, p.14). In Lake Babine worldview every aspect of the environment is alive and filled with spirit and worthy of great respect. It is how people behave and how they integrate which creates a vision of society where every member has a role and a sense of value. The Lake Babine people share a belief in the existence of unseen powers and a number of customs are performed in this

regard (Aasen, 1992). In Lake Babine tradition, spiritual life is intrinsically linked with the social life and is reflected in procedures that connect spiritual and physical practice. Lake Babine behaviour is conducted in relation to spiritual energy. The Lake Babine people believe whatever energy is expressed, good or bad, will be visited on the individual in the future. This belief guides the respect demonstrated towards all other beings including practices performed prior to activities such as hunting designed to demonstrate respect (Holyk, Adam & Shawana, 2005).

Contrary to the primacy placed on the individual in Western society, Lake Babine spirituality and worldview create a sense of place that connects all individuals guided by the ancestors through a shared history, language and culture. This deep-rooted shared existence provides a moral code for ways of being through common action (Denis, 1997). In Lake Babine spirituality, individuals relate to the creator and other spirits as a society (Taylor, 2004). According to Taylor (2004), growing up in a world where the social imaginary accepts the shared existence with spiritual forces sets limits on one's sense of self, embedding individuals in society as well as the cosmos.

The attempted removal of Lake Babine worldview and spirituality can thus been seen as an attack on the Lake Babine concept of community. Spirituality and in particular the removal of spiritual forces in favour of a rational order was one of the key objectives of the shift from natural to civil order (Taylor, 2004). The Lake Babine inseparably link spiritual life with social life and the church worked hard to dispose of Carrier concepts of spirituality. As noted by Fiske & Patrick

(2000), "the Babine did not necessarily deem transgressions to be an affront to spiritual obligations, as did the Roman Catholics, but saw them as an expression of disregard for the code of respect that define appropriate interactions" (p.152). Christianity played a large role in the shift in conception from family, clan and community obligation challenging individuals to leave such spheres and accept their individual place in the kingdom of Heaven. According to Taylor (2004), we see this reflected in the manner churches operated "where one was not simply a member by virtue of birth but had to join by answering a personal call" (p.62). Similarly, in accepting the Christian religion, First Nations first had to denounce all traditional forms of spirituality in order to be accepted into the church (Kelm, 1998). Thus, what was required of members was a denouncement of community in favour of personal salvation. In practice, priests did not provide last rites to those who did not denounce Lake Babine practice and often prevented individuals from being buried in the church graveyard.

The Lake Babine nation struggled to reconcile their spirituality with Christianity and often shifted between belief systems depending on the context. According to Jenness (1934), the supernatural world has not been abolished but rather the community added a second layer of religion and incorporated knowledge of the Christian religion within traditional system (as cited in Aasen, 1992). The Lake Babine never intended to abandon their own philosophies but rather integrate Christian concepts and practices into their own belief system (Fiske, 2000). In this manner the Babine were defining the formation of a new social imaginary. However the Lake Babine social imaginary continues to be

developed as the Lake Babine balance incongruencies between Western and traditional philosophical traditions including divergent views of economics, spirituality and governance (Taylor, 2004). Elders typically recognize that Lake Babine society was not secularized in the past and want self-governance to contain aspects of spirituality (Aasen, 1992).

3.4 Governance

The principle shift in spirituality within Western societies is guided by the view that Modernism, society's political self-making, can and will bring about fundamental human emancipation through systems of governance structured around the secular. According to Taylor (2004), the transformed social imaginary shifts focus from the spiritual realm, to a vision that elections are the only source of legitimate power. The process removes the Creator, or more generally the spiritual from the centre of human life and replaces it with society, which maximizes individual freedom by providing security and prosperity⁴⁵ (Taylor, 2004). The process of popular sovereignty, or self-rule renders individual's lives and property safe under the law. The system becomes so embedded in the social imaginary of the society that the state is able to maintain its claim to the legitimate legal authority, enforced by police, judges, social workers and courts as well as by consent of equal citizens (Denis, 1997). However, members of the Lake Babine view those seen as the legitimate legal authority in Western society as actors in the process of cultural invasion (Freire, 2009).

_

⁴⁵ This again can be seen as an attack on community by providing Individual rights.

The Canadian state that regarded the Lake Babine peoples as lacking political systems, based on the manner in which the Lake Babine system opposed the concept of liberalism and the state imposed Western political, cultural and economic systems over existing systems. By the 1890s Fiske (2000) notes that the government was taking an active judicial role in the affairs of the Lake Babine peoples. In 1880, the federal Department of Indian Affairs was created and empowered the Superintendent General to impose the elected band council system on any band he determined to be ready (Tobias, 1991, p.133). The new regulations incorporated into the *Indian Act* also state that once implemented only those elected according to the provisions of the *Indian Act* were enabled to speak on behalf of the band, attempting to eradicate hereditary leadership. In 1884, the *Indian Act* further extended the ability of the Superintendent to direct political affairs, including the ability to call and supervise elections, as well as call and supervise band meetings. In recognition that many nations were electing traditional leaders, the *Indian Act* allowed the superintendent to dismiss any leader he determined to be incompetent or immoral (Tobias, 1991). This piece of legislation was often used to remove hereditary leaders or anyone not following the direction of the Indian Agent. In a continued attack on the hereditary system, the Indian Act also banned the potlatch because "they promoted pagan beliefs and were anathema to the development of a concept of private property" (Tobias, 1991, p.135).

Contrary to Western views of Lake Babine governance, Lake Babine history recognizes the integration of the *Bah'lats* as a method of preserving

peace at a time when civil unrest between nations was at a peak. The Lake Babine determined the distribution of food and gifts at a *bah'lats* as a method of maintaining reciprocal obligations and providing economic balance within society as well as a method of witnessing business of the nation. The Lake Babine people regard the system as transparent because all business is conducted in an open forum. Transparency within Lake Babine society requires discourse, which is defined publicly and continually repeated and acknowledged through group process. The government was concerned about assembly and attempted to abolish the *bah'lats* because as Freire (2009) notes, "cultural revolution develops the practice of permanent dialogue between leaders and people" (p.160). The hereditary system reflects egalitarian values in that even the most powerful chief will not act without consulting wing chiefs and clan members prior to decisions being made. As such, the system provides the Lake Babine nation with a means to maintain order and reciprocal economic exchange.

3.5 Economics

While Lake Babine and Western systems place importance on organized society through governance as well as economic exchange, the philosophical underpinnings differ dramatically. In Western systems of economics, concepts including maximizing individual benefit began to play a central role in social stability with the shift to the liberal ethic. According to Taylor (2004), "because the whole theory emphasized a kind of profitable exchange, one could begin to see a political society itself through a quasi-economic metaphor" (p.71). Leadership in

Western societies determined that the path to peace and orderly existence was through economic activity. Taylor (2004) notes, "the more a society turns to commerce, the more polished and civilized it becomes, the more it excels in the arts of peace" (p.74). Both Lake Babine concepts of connection based mutual obligation and the commercial, Western system provided connection through interlocking activities of production, exchange and consumption that created reliance on individuals within society (Taylor, 2004).

Post Enlightenment, this translated into an increased emphasis on productive and material features of society that support individuals in maximizing their potential for the benefit of society as a whole. The basic normative principle is premised on the individual within society. Personal independence is a moral ideal maintained by the fact that "disciplined, honest imaginative entrepreneurial people were seen as the cornerstone of the new society, which combined order and progress" (Taylor, 2004, p.150). In theory, supporters of the Western economic paradigm suggest individuals serve each other's needs, help each other, and behave like rational and social creatures in order to maximize their own potential; in doing so, the society or collective benefits (Taylor, 2004).

Within the Lake Babine economic system, the clan system managed and protected resources. The clan, through hereditary leadership, provided a means for ownership and sharing of resources within and outside of the membership (Hudson, 1983, p.58). Each clan had several title-holders who controlled access to resources on behalf of the clan. The resources of other clans could only be obtained through sharing (allowing access to territory) or exchange (formal trade)

(Hudson, 1983). Title meant the ability to use a particular resource area and the capacity to utilize the land meant drawing on the labour of all clan members thus creating a system of interdependence. In addition to clan interdependence the cyclical nature of resources and production of higher salmon yields on various river systems depending on their cycle, required an escalating dependence on other clans in order to survive. The dependence created obligations and respectful relations with members of the nation including a considerable emphasis placed on reciprocity and support. Such obligations were reduced with changes to Lake Babine economic system.

The transformation of the Lake Babine economic system began during the fur trade but was accelerated as trapping companies gained access to trapping territories and wage labour became part of the Lake Babine economic base (Hudson, 1983). While the Lake Babine actively participated in trade during the early period of contact, this was done while still maintaining their property rights, laws and social structures. They were active in trade with other nations prior to contact so the fur trade extended these relations. However, with settlement Hudson indicates, "economically, the region became integrated into an expanding industrial capitalism, which replaced mercantile interests" (Hudson, 1983, p.106). The establishment of registered trap lines, placed over existing territories of the Hereditary Chiefs undermined the power of the Lake Babine system. Clan leaders lost control over resource use as trap line registration shifted land to individual or family trapping groups. Even when clan leaders attempted to rectify this by registering their hereditary lands, often the land

allotment, cut as square blocks on the territory did not match the natural geographic features of hereditary territories and issues of overlap between hereditary territories became an issue. The Lake Babine also had to compete with non-Indigenous peoples for access to land and resources placing further strain on Lake Babine economic and political systems by forcing individuals out of their own territories taken by non-indigenous peoples and forcing them to compete for other clan territories. Matrilineal property rights, tracing your lineage through the female line, and the hereditary system of governance was undermined because of trap line registration strictly to men.

Pressure by the church and government further destabilized the matrilineal inheritance rights to estates dramatically affecting not only the role of the passing on of hereditary names but also the rights of citizens (Hudson, 1983). Under the direction of the Catholic Church, families were to be male headed and men were reminded "in the family it is the father who is the master" (Morice as cited in Fiske 1981, p.92). The Oblates assumed in order to "become thorough Christians and imitate the whites they must sacrifice nearly all their social institutions and customs... [including] heredity through the female line" (Morice as cited in Fiske, 1981, p.92). As such, matrilineal systems were forcibly challenged and effort was made to institute paternal systems.

Economically, the power of the hereditary system was increasingly challenged through federal government restrictions on fishing methods. Under pressure to eliminate interior salmon fishing by First Nations, the federal government challenged the Lake Babine use of fish weirs and traps (Hudson,

1983). In 1905 through 1907 fisheries officers destroyed Lake Babine fishing weirs and in 1907 members of the nation were arrested (p.107). According to Fiske & Patrick (2000), "the struggle over fishing rights clearly illustrates the most dramatic loss of legal authority suffered by the hereditary chiefs since the inception of the anti-potlatch law of 1884" (p.167). The compensation received through the Barricade Treaty did not achieve social goals of Lake Babine including respect and reciprocity. Seizure of their fish and other natural resources discounted and disempowered the laws of the *bah'lats* and the role of Hereditary Chiefs responsible for the protection of their territories (Fiske & Patrick, 2000).

The act of destroying fish weirs had an important impact on the emphasis placed on reciprocity and support. The elimination of weirs at Lake Babine meant the virtual cessation of trade and the end to the use of Lake Babine for surplus fish production; thus forcing individuals into wage labour. Similarly, the government further emphasized individual labour through government provision of farming implements, nets and protection of fishing stations in exchange for the dismantling of weirs (Hudson, 1983). Through direct government intervention "the collective labour required to erect and maintain the weirs was replaced by family-owned means of fishing production" (Hudson, 1983, p.108). With weirs banned, the clan-based production was no longer as effective in resource use and local groups began to move seasonally across various watersheds to produce rather than trade resources needed.

Despite the influences of capital, the Lake Babine system was able to adapt to a large extent and redistributive mechanisms continue to operate.

Hunting, trapping and fishing remain integral to Lake Babine social systems and the hereditary system has adapted to meet challenges imposed on their authority. The wage economy has resulted in cash and commercial goods becoming a key component of the *bah'lats* system while the core values have remained intact.⁴⁶ As noted by Taylor, "new imaginary doesn't just displace the old one. It reinterprets the key values of the older tradition but retains the sense of its origin in its earlier tradition" (p.153).

3.6 Reclaiming Social Imaginary

Just as Western social imaginary has been shaped by war, revolution and economic change, the Lake Babine Social imaginary is largely a response to and transformation of practices resulting from both colonization and the attempts to decolonize. Colonization has played a large role in developing the Lake Babine vision of their place within time and space as well as characteristics that establish the Lake Babine peoples as a nation. The racism faced by First Nations in the general population made the ideal of individual equality easy to challenge by the Lake Babine who remained on the margins and thus still required a sense of community. Taylor (2004) contends, "one of the most powerful narrative modes centres around the nation" (p. 176). The Lake Babine peoples are believed to belong together "in virtue of common language, common culture, common religion, or history of common action" (Taylor, 2004, p.177). The past two hundred years of contact have fundamentally shaped Lake Babine knowledge of

_

⁴⁶ It is a constant struggle for leadership to ensure the values and systems remain intact.

their relation to other nations as well as their history. One of the paramount victories seen by the Lake Babine Nation is that despite all of the attempts at cultural destruction from the outside the nation is still intact and behind the political nation stands a pre-existing cultural nation.

One important aspect of the development of Lake Babine social imaginary is the reclamation of the public sphere. As Taylor (2004) notes,

the public sphere is a common space in which members of a society are deemed to meet through a variety of media: print, electronic, and also face to face encounters; to discuss matters of common interest; and thus be able to form a common mind about these (p.83).

In the public sphere, widely separated people sharing the same view are linked through a space of discussion, wherein they are able to exchange ideas with others and reach common understanding (Taylor, 2004). While forums of public discussion involve various mediums, the bah'lats has remained a central institution in the maintenance of common interests. The bah'lats is a common space in which people come together to celebrate and witness major events and engage in common purposes such as assumption of hereditary names, rites of passage, acknowledgement of death and resolution of disputes. The bah'lats system and its importance in the creation of a national identity is guarded as not only an example of the perseverance of the Lake Babine peoples but also as a core institution in the preservation of Lake Babine identity and thus social imaginary. As one elder noted "we have more in common than our differences. We all follow our mother's clan. We all belong to clans and we can have pride in that" (as cited in Aasen, 1992, p.64). It is a core tool in the perpetuation of the Lake Babine dialect and culture. The bah'lats provides a key link to the ancestors and fosters social obligation and civic responsibility through maintenance of reciprocal obligation between clan members.

The Lake Babine continue to cope with incompatibilities between Lake Babine values and those of the Western society as their own social imaginary is formed. With the *bah'lats* as a core feature of the Lake Babine social imaginary, there is concern that a shift to a solely western system of governance will result in a loss of Lake Babine culture and identity. As noted by Fiske & Patrick (2000), "contemporary customary law has emerged from a series of struggles over meaning, morality, and spirituality as an integrated phenomenon that takes meaning from a political and economic environment that is profoundly different from the one in which it originated" (p.176). Problems in modelling government arise when Lake Babine leaders attempt to translate Lake Babine institutions into systems found in state society (Aasen, 1992). The *bah'lats* has clearly been impacted by traders, missionaries and the appointment of democratically elected leaders in Lake Babine communities and continues to develop as the Lake Babine social imaginary is formed and reformed.

3.7 Final Remarks

The Lake Babine Nation has been impacted by spiritual, political, economic and cultural systems that have shaped the Western world and have been introduced, if not imposed on the Lake Babine peoples. As noted, in this analysis the Lake Babine continue to reconcile their worldview with Western ways of knowing. In doing so, the Lake Babine social imaginary continues to

evolve, as issues of historical and contemporary life are reconciled in theory and practice. The Western concept of a modern social imaginary has not fully become imbedded in Lake Babine society because the Lake Babine still live in a pluralistic society where traditional concepts and Western ideologies conflict. As such, the Lake Babine social order is complex and people rely on traditional Lake Babine systems when a situation deems the system appropriate and Western concepts when dealing with situations where Western ideology is appropriate. Freire (2009) argues "The solution is not to "integrate" [the oppressed] into the structure of oppression, but to transform that structure so that they can become "beings for themselves" (p.74). The Lake Babine people continue to develop their social imaginary while being influenced by the dominant Western system, which is pervasive in their lives. The Lake Babine moral order continues to struggle with the cognitive imperialism used to challenge or disclaim their knowledge base and values. However, the Lake Babine social imaginary is not static and continues to develop as new concepts are introduced from various sources including other Indigenous nations. Discussion in this chapter reflects the historical and cultural complexity of the Lake Babine Nation and the multiple social imaginaries that are now present within the Nation (Lemert, 1994). This analysis creates an understanding of how divergent histories and worldviews influence the Lake Babine Nation parents' understanding of education and assessment within the school system. As I note in later chapters, parents demonstrate the flux between Western and Lake Babine worldviews when discussing subjects required by their children and assessment methods.

The Lake Babine Nation social imaginary continues to be formed and reformed due to the influence of other societies and the reconciliation of the colonization process. The battle against colonial processes has resulted in a number of conflicts within LBN society including individual vs. community, hereditary governance vs. elected leadership, and Catholicism vs. Carrier Spirituality. The uniqueness of LBN society supports Taylor's statement,

If we define modernity in terms of the spread of the modern bureaucratic state, market economies, science, and technology, it is easy to go on nourishing the illusion that modernity is a single process destined to occur everywhere in the same forms, ultimately bringing convergence and uniformity to our world. Whereas my foundational hunch is that we have to speak of "multiple modernities," different ways of erecting and animating the institutional forms that are becoming inescapable (p.195).

4: CHAPTER FOUR: METHODS

4.1 History of Research in Indigenous Communities

The history of research in Indigenous communities is important to acknowledge because it shapes the community perception of research and willingness of community members to engage in the process. Culturally respectful research includes an understanding of the distinct nature of First Nations communities, as opposed to pan-Indian approaches, as well as the historical context of research in First Nations communities (Canadian Institutes of Health Research, 2007). According to Steinhauer (2002), Indigenous research methods must emphasize "the social, historical and political contexts which shape [First Nations'] experiences, lives, positions and futures" (p.72). The colonial history imposed on First Nations, including the Lake Babine Nation, reflects a legacy of cultural disruption and trauma that continues to influence communities (Caldwell et al., 2005). The historical circumstances of First Nations peoples and their contact with Western systems, including research and the role it has played in the colonization process, affect the manner in which researchers

must conduct ethical research. Research cannot be carried out in indigenous communities without "understanding the complex ways in which the pursuit of knowledge is deeply imbedded in the multiple layers of imperial and colonial practices" (Smith, 1999, p.2). A parallel relationship exists between the historical treatment of First Nations and the research methods used to study them.

Understanding of contemporary communities requires knowledge of the history of First Nations including the dispossession of lands and attempted eradication of First Nations culture, political systems, language and spirituality as well as the role research played in facilitating such processes (Caldwell et al., 2005). Research during the contact period, conducted by both explorers and researchers reflects a justification for European expansionism and the devaluing of Indigenous Knowledge. Research, informed by ideology, legitimated the policies and practices of Western governments and the imposition of decisions affecting every aspect of First Nations' lives. The research legitimated the attempted destruction of culture and expropriation of lands (Smith, 1999).

Formal research completed since the 19th century typically legitimated the status quo. Following premises reflected in the *Indian Act* legislation researchers and policy makers were external to First Nations communities.⁴⁷ In addition, while First Nations populations have been continually researched, their health, social and legal status continues to fall below other Canadians (Kendall, 2009). The dominant culture dictated questions derived from outside of the community with research being done to rather than with the communities. Researchers have at

_

⁴⁷ The *Indian Act* indicates the responsibility of Federal Government for Indians and Lands reserved for Indians.

times regarded communities as a source of data with research questions and results creating no local benefit to the community, and in some cases harming the peoples involved (St. Denis, 1992). The presentation of inaccurate or insensitive research has caused stigmatization or legitimization of the status quo regarding the perception of First Nations within the dominant culture. Researchers have also either unwittingly or unethically placed sensitive knowledge of flora and fauna in the public domain, providing information to corporations or other bodies who use Indigenous knowledge for financial gain without any obligation to return the benefits to the community (St. Denis, 1992). Many researchers have collected data without the full knowledge and consent of participants or respect for the culture and individuals being studied (Caldwell et al., 2005). On occasion, researchers have taken the cultural property and human remains of First Nations for exhibition or sale. Within Carrier Sekani Family Services specifically, of which Lake Babine Nation is a member, there have been a number of ethical breaches including publishing information without consent, researching without agreement by the nation, attempts to control ownership of community information and research conducted without informed consent of participants (Holyk, 2008). As a result, First Nations' communities regard research as a pillar for the regulation and realization of colonization and many individuals regard research with mistrust and apprehension.

Despite the problems associated with the past conduct of researchers, research when done with respect and community-driven, has become a tool of resistance, addressing social issues within the context of self-determination. First

Nations organizations, including Carrier Sekani Family Services have developed policies regarding the practice of research involving their nations, demanding control of research activities and knowledge produced (Smith, 1999). Community organizations recognize the importance of research in informing decision-making, improving programming and designing programs that meet the needs of the community members they serve. Carrier Sekani Family Services developed ethics guidelines for research in order to ensure research reflects the interests of the community, is culture specific and brings direct benefit to the community.

4.2 Research Framework and Priorities

One of the core goals of the research was to exemplify methods of respectful research in First Nations communities. As a result, the research involved the community in all stages of research conception and design. I determined the central questions of the research through consultation with leadership of the Lake Babine Nation. Out of that process, two questions were generated:

What do Lake Babine Nation parents of Grade 4 children deem as important indicators of their children's achievement in school? And,

What do Lake Babine Nation parents understand related to their children's academic assessment?

In order to address these questions I used a qualitative approach to research applying conventional methods of data collection, specifically

participatory action research (utilizing interviews as the primary data collection method), while maintaining Carrier First Nations' values, practices and methods (see Table 1).

Table 1. Carrier First Nations Values, Practices and Methods Utilized

| | Actions Taken |
|--------------------------|---|
| Respect for Participants | Anonymity of Participants was protected by protecting access and dissemination of personal information and not including information that identified smaller communities where there were few participants. |
| | Participants signed a consent form and agreed to interview being recorded |
| | Participants were given a copy of transcripts to ensure accuracy and remove anything they did not want included. |
| | Honorarium (50 dollar gift certificate to local food store) provided to show respect for participation and information provided. |
| | Lake Babine Nation (LBN) Steering Committee reviewed results for accuracy of information, quotes used, information that should be omitted or altered to protect participants or First Nation |
| Benefit of the Research | Engaged community leaders in discussion regarding an education research topic of interest to the community |
| | LBN Steering Committee Agreement to the benefits of the research |

| Direct involvement of Community | Participatory Research MethodsEngagement of Community Leadership |
|--|---|
| | Involvement of LBN Steering Committee (comprised of Carrier parents, elected and hereditary leadership, youth and education professionals) |
| | Involvement of an LBN Research Assistant. |
| Information Provided to Community | Meetings with Chief |
| | Transcripts provided to participants |
| | Presentations at various stages to LBN Steering Committee |
| | Conference presentation where numerous community members were in attendance |
| | Community Presentation |
| | Copy of final report provided to Chief and Council and member of LBN Steering Committee |
| Originality of the Research Proposed | Views of parents regarding assessment have not previously been completed in LBN |
| | LBN Steering Committee agreed to the usefulness of the project |
| | Chief and LBN Steering Committee reviewed and altered proposed questions |
| Increasing Community Capacity to Complete Research | Youth included on the LBN Steering Committee to learn research methods |
| | Community Based Research Assistant trained |
| Management of Research Materials | Research Protocol defined the roles and responsibilities of myself and community and established policy for ownership, control, access and possession of research materials |

Qualitative research is an interpretive method in which researchers attempt to make sense of phenomena in terms of the meanings people, or a group of people, bring to them (Denzin & Lincoln, 1994). Qualitative research begins with the acknowledgement that there is a gap in our understanding and that clarification or illumination will be of benefit. Qualitative research does not necessarily provide definitive explanations but it does raise awareness and increase insight. The methods provide rich and personalized information leading to a more complex understanding of the factors involved, in this case, indigenous education, specifically within the Lake Babine Nation (Mellor & Corrigan, 2004). My experience working with Lake Babine members has taught me they welcome the opportunity to tell their story and qualitative methods provide the opportunity for participants to express their voice.

In addition to meeting all research ethics requirements at Simon Fraser University, this research adopted guiding principles and processes from the Carrier Sekani Family Services (CSFS) Research Ethics Policy, which outlines community perspectives regarding the conduct of research. The guiding principle of the CSFS Research Ethics Policy is respect, specifically respect for participants and their informed consent to participate in research, protection of their confidentiality and protection of intellectual property and cultural integrity of information provided. As noted, "Carrier Sekani Family Services shall protect the (cultural, mental, spiritual, physical, emotional) interests of participants throughout the research process" (CSFS, 2004, p.2). Researchers carrying out research in Carrier communities are responsible to the communities for engaging

in research in a respectful and ethical manner that adheres to the specific protocol of each Carrier First Nation. The basic tenet of the research process involving Carrier First Nations is that the research must benefit the community directly. I incorporated this principle in a number of ways outlined in the research design below and noted in Table 1.

4.3 Sample Overview- Building Lasting Relationships

I chose the Lake Babine Nation as the community for this study for substantive and methodological reasons. Communities have often criticized researchers for their methods of conducting research because they enter the community, collect their data and are never heard from again (Brizinski 1993; St. Denis, 1992). I have built lasting relationships with the Lake Babine Nation professionally and personally over many years, having engaged in research with the Lake Babine Nation since 2003 when I conducted research with the community on a community needs assessment used to structure child welfare services. The community is also a participant in ongoing youth suicide research of which I am involved. Working closely with the community on an ongoing basis provided opportunity to consult with leadership of the Lake Babine Nation and develop a research question that was mutually important.

Personally, I have developed close relationships with community members and I, as well as my wife and children, actively participate in *bah'lats* and cultural activities, supporting the trust and responsibility necessary to conduct research in First Nations communities. Participation in cultural activities on an ongoing

basis, rather than only when associated with research, supports the idea that I am involved with the community and will not disappear after receiving the research information. It is a commitment to participating with the community in all aspects of culture to gain an understanding of Lake Babine ways, thus improving the work completed and building an understanding of the sensitivities and values needed to do this research humbly and respectfully.

The relationship provides unique opportunity and responsibility that add value to my scholarship. Living and working in the community creates obligations to ensure the accuracy of the information and produce a document that has local benefit. The trust also provides an environment where individuals feel open to speak because they know I will respect the information and I have a community responsibility to use the information in a manner that will do no harm to the participants. This method of research creates an important dynamic of being both an insider in some regards as well as an outsider.

With 2282 members, the Lake Babine Nation is the largest Carrier First Nation, one of the largest Bands in Northern British Columbia, and has enough members to provide a sample for this research. My goal is to create a community specific study that may have implications for other nations, following the principle that Indigenous knowledge is community specific. I am critical of studies and programs that suggest a Pan-Indian approach that devalues the richness in diversity of the First Nations of British Columbia. Even within Carrier territory, while Carrier nations share a language, they are culturally diverse with varying dialects, clans and governance systems as well as a vast geography with a

traditional territory spanning 70,000 square miles.⁴⁸ I did not want to delineate between possible community differences, as this was not the research goal and decided to choose one community to work with.

4.4 Participatory Action Research

Participatory action research most closely reflects the values and ethics policy of the Lake Babine Nation and forms the over-arching method of inquiry used in this research (Creswell, 2003). In the case of First Nations peoples, The Tri Council accepts participatory approaches as the preferred research strategy for research involving First Nations, noting many of the practices incorporated in participatory approaches as "good practice" (Tri Council Policy Statement, 2005, p.6.3). Action Research is appropriate because the research values are empowering and reflect an explicit commitment that the outcomes of the study benefit the participants and their communities. Dickson and Green (2001) advocate that participatory approaches are the best way to ensure research accurately reflects the experiences of marginalized peoples, including First Nations.

Participatory action research is a social process that explores the relationship between individuals and social systems. This research focuses on improving assessment practice by informing community leadership and the

..

⁴⁸ For example, some communities do not participate in the potlatch or *bah'lats* system.

⁴⁹ The *Tri-Council Policy Statement: Ethical Conduct for Research Involving Humans* (TCPS) describes the policies of the Canadian Institutes of Health Research (CIHR), the Natural Sciences and Engineering Research Council of Canada (NSERC), and the Social Sciences and Humanities Research Council of Canada (SSHRC), expressing the three Agencies' continuing commitment to the people of Canada to promote the ethical conduct of research involving humans.

school district regarding parent knowledge and expectations of education and education assessment. The process supports the concept of participatory research, employing an alternative philosophy of research associated with social transformation (Mitchell & Baker, 2005). According to Kemmis and McTaggart (2005) "through participatory action research, people can come to understand that- and how- their social and educational practices are located in, and are a product of, particular material, social and historical circumstances that produced them" (p.565).

In practice, participatory action research facilitates the ability of community knowledge holders and leaders to inform the research design, decision-making, research questions and analysis. The principles of inclusion shift the balance of power, allowing the community to control the research process affecting their lives (Minore, Boone, Katt, Kinch & Birch, 2004). Actively involving the community in the research process results in an improved likelihood that the research will benefit the community and have direct impact (St. Denis, 1992). Participatory action research engages individuals in examining their worldview and values to reflect critically on how their knowledge frames any action taken (Kemmis & McTaggart, 2005). The method reinforces the mutual production of knowledge between researcher and community (Mitchell & Baker, 2005). This research has assisted community members by providing information regarding assessment as part of the dialogue, and engages parent's in consideration of concepts such as assessment, which can powerfully impact their children.

4.5 Community Engagement

Specifically, the Lake Babine Nation was involved in all aspects of research project design and implementation through the methods outlined in Table 2:

Table 2 Community Consultation Steps

| Community Consultation Steps |
|---|
| Discussion with leadership to determine a relevant topic |
| Development of a Research Protocol outlying expectations |
| Letter of Support Signed by Chief of First Nation |
| Creation of a formal community research steering committee |
| Hiring and training of a community research assistant |
| LBN Steering Committee Meetings at each stage of research (Study and Questionnaire Design, Results interpretation, Final Report |
| Presentation of Final Report- LBN Steering Committee and Chief Copies of Report provided to Steering Committee and Chief |

I initiated the research process by engaging community leaders in discussion regarding an education research topic of interest to the community. Community leaders in political, education, health and social services were informally asked what areas within the realm of education required research. Informal discussions between myself, service providers, hereditary, and elected

leadership determined assessment to be an important area of enquiry in education.

Building on the initial consultation, I established formal relationships with the Lake Babine Nation including the development of a Research Protocol between the community and myself. Chief and Council provided a letter of support endorsing the project. The Research Protocol defined the roles and responsibilities for the community and me and established policy for ownership, control, access and possession of research materials (OCAP) (see Appendix A). The Lake Babine Nation and I agreed to create a formal research steering committee comprised of Lake Babine parents, elected and hereditary leadership, youth and education professionals to guide the research process and inform the process at strategic points in the research⁵⁰.

The Nation also agreed I would undertake the research with the assistance of Rosa Johnny, who is a member of the Lake Babine Nation, in order for members of the nation to gain valuable training in research skills and for me to gain further acceptance in the community while completing the research. Rosa engaged in a number of activities including participation in LBN Steering Committee meetings, calling potential participants and setting up interview times, attending the majority of interviews, peer debriefing, transcribing interviews and taking LBN Steering Committee meeting minutes. Calling potential participants and attending interviews was advantageous in establishing participant trust from first contact. In some cases, due to the mobility of potential participants it was

.

⁵⁰ Referred to throughout the document as LBN Steering Committee in order to distinguish between the community established committee and the Dissertation Supervisory Committee.

difficult to locate parents without the use of family networks that only a community member would possess. When interviewing community members who did not know me, having Rosa provide introductions established instant credibility that I was someone who is trusted by the Lake Babine Nation. I also informed the participants that the research was designed in partnership with a Steering Committee comprised of respected community members, and approved of by both the Committee and Chief and Council. In some cases, due to the mobility of potential participants it was difficult to locate parents without the use of family networks that only a community member would possess. Rosa knew individuals from each family and was able to contact relatives in order to determine where the potential participant was currently living.

The purpose of the LBN Steering Committee was to ensure the research was culturally appropriate and adhered to the highest research standards and ethics. Chief Betty Patrick and I determined the complexion of the LBN Steering Committee and the specific individuals who participated based on consultation⁵¹. For example, Chief Patrick noted that having a youth on the committee was important to capacity building through allowing a youth to provide perspective into the research but also learn about the research process and design. The LBN Steering Committee came together at strategic points including an initial meeting October 17, 2008 to discuss the project rationale, approve the overall research question, assist with research design and develop interview questions. I held a

_

⁵¹ If I had worked with someone previously I requested he/she participate and if not I sought advice from Chief Patrick. The Hereditary Chiefs and youth for the project I had worked with previously on youth suicide research. The counsel member was selected because she held the education portfolio.

second meeting November 6, 2009 to discuss the completion of the interviews and results and a third meeting February 19, 2010 to review a preliminary draft. At the first meeting, I asked the LBN Steering Committee what they would like to know about assessment, their view on success in education and how to create questions that would evoke a process of change in order to encourage parents to think more deeply regarding assessment. The LBN Steering Committee also provided their perspective on the possible sample including the geographic area of the study, sample size and representation within the Nation.

The LBN Steering Committee agreed with my suggestion to the focus on Grade 4 as an important transition period, from primary to intermediate grades. Children at this age are settling into the school system and beginning to demonstrate leadership as well as knowledge patterns. It is also early enough in the process to be useful in providing context for First Nations parents to consider important concepts such as assessment, and in the process potentially improve how they relate to their children's education and the system as a whole. Research by Henderson and Mapp (2002) regarding family school partnerships demonstrates it is easier to involve parents in their children's education early rather than when in secondary school (as cited in Martinez, Martinez and Perez, 2004). I also chose this sample because Grade 4 is the first time children participate in the Foundational Skills Assessment and the cohort provided information regarding parents' knowledge of large-scale testing and their perceptions of its usefulness. The LBN Steering Committee indicated these subjects as important.

At the October 17, 2008 meeting, the LBN Steering Committee expressed their pleasure that the research was taking place, noting, "We have been battling with the School District #91 for years and years to do assessments on our students, but they will not listen... This would be a really good document to take back to the school district and tell them "See this is what we have been telling you" (LBN Steering Committee Meeting Transcripts, October 17, 2008).

The LBN Steering Committee met November 6, 2009 to review preliminary information from the data analysis. I presented the information in the aggregate and names were kept confidential with only numbers attached to quotes in the same manner as provided in the dissertation. I attempted to keep the LBN Steering Committee Members informed of my progress when I visited the community. The majority of LBN Steering Committee members and Chief Betty Patrick were also present the first time I presented preliminary research results at an education conference in December 2009.

4.6 Questionnaire Design

In order to establish research questions, I designed a draft questionnaire based on dialogue with community leadership and the result of a literature review regarding parent understanding of assessment and provided a list of questions to the LBN Steering Committee for discussion and approval (Anderson & Bachor, 1993; Battle Bailey, 2006; Benner & Mistry, 2007; deGroot- Maggetti, 2002; Denney, English & Gerber, 2001; Dorman & Knightley, 2006; Guppy & Davies, 1999; Hallett et al. 2008; Kavanagh, 1998; Martinez, Martinez, & Perez, 2004; Miretzky, 2004; Prince & Howard, 2002; Prince, Pepper & Brocato, 2006;

Rosenzweig, 2001; Van de Grift & Houtveen, 2006). The questions remained in draft form until the LBN Steering Committee, my Dissertation Supervisory

Committee, and I all agreed the questions were acceptable (Coggins, Williams & Radin, 1997; McLean, 1997). The LBN Steering Committee played an important role in the process by ensuring questions encouraged conversation and used non-technical and non-threatening language. The LBN Steering Committee and I engaged in considerable dialogue until everyone was satisfied with the questions. Table 3, outlying changes made by the LBN Steering Committee, indicates the level of involvement in shaping the questions. I then provided those questions to the Dissertation Supervisory Committee who reviewed the questions based on the appropriateness of the questions for finding the needed data.

Table 3. LBN Steering Committee Assistance with Questionnaire

| Draft Questions | Change made LBN Steering Committee |
|---|--|
| What does it mean to be educated | What does education mean to you? |
| What do you feel is an important indicator of your child succeeding in school? | Deleted |
| How should we measure success? | How should we determine success in school? |
| How do you know how your child is doing in school? | What kinds of information do you receive about your child? |
| On an ongoing basis, what strategies do you use to check your son's/daughter's academic progress? | On an ongoing basis, what strategies do you use to check how your son/daughter is progressing in school? |

| When you think about student assessment or evaluation- what do you think teachers are doing in the classroom to assess your child? | What do you think teachers are doing in the classroom to understand how your child is doing? |
|--|---|
| Do assessments connect with learning activities you would like to see for your child? | Deleted |
| What do you feel are/should be the most appropriate methods of assessment used by teachers? | What do you feel would be the best way for teachers to find out how your child is doing in school? |
| What assessment strategies would help students achieve in both Carrier and Western culture? | What form of testing would help students achieve in both Carrier and Western culture? |
| What do you feel needs to be assessed in order for your child to be successful? | Deleted |
| What do you think of the provincially mandated assessments in Grade 4? | How do you feel about provincial testing in Grade 4? |
| What do you think the provincially mandated assessment in Grade 4 tells you about your child's progress? | Deleted |
| Added | Do you want to know how your son/daughter is progressing compared to other children his/her age? |
| Added | What is it you would like your child to learn at school? |
| Added | What is the role of a school in your child's life? |
| Added | How could schools make it easier for you to be involved in testing your child's progress in school? |

| Added | When you think about your child's understanding in school, do you agree with the results of school tests? |
|-------|---|
| Added | What should passing to the next grade be based on? |

4.7 Research Sample

The population sample for the research consisted of Lake Babine Nation parents of Grade 4 students in the 2007-2008 school year and living in areas between Prince George and Smithers⁵². With the assistance of the Lake Babine Nation Indian Registry Administrator, I determined there were 36 children who are born in 1999 and thus potentially in Grade 4. With LBN Steering Committee assistance, I concluded that at least one parent from half of the possible sample, or at least 18 separate families would be invited to participate in the study. According to Onwuegbuzie and Leech (2007) in qualitative research, "only one type of statistical generalization is pertinent, namely generalizing findings from the sample to the underlying population" (p.240). With a population of 36 children, I estimated half of the parents would be a large enough sample to assume I had heard most of the perceptions of assessment that might be important in order to achieve data saturation, while at the same time being a workable number of interviews that would enable deep, case oriented analysis

⁵² In consultation with the Steering Committee I excluded those who reside outside of the Lake Babine Nation Geographic area from participation for reasons of access.

(Guest, Bunce, & Johnson, 2006; Onwuegbuzie & Collins, 2007; Sandelowski, 1995).

Due to parental interest in joining the research project, all but one of the parents who were available at the time of a scheduled interview agreed to participate, raising the final sample size to 18 families. When contacting participants I typically asked for the parent who was responsible for primary caregiving role and involvement with their child's school to participate in the interview. In two interviews, both parents were present and I counted these interviews as one interview as one parent provided the majority of information and the other added a few comments.

Based on discussion with the Indian Registry Administrator I established that there were 29 possible people to interview in the Prince George to Smithers geographical area. However, through the actual tracking of parents for the research I concluded that parents have a high level of mobility as nine of the possible participants had moved outside of the geographic area within the year when the Indian Registry Administrator had updated the addresses. ⁵³

The movement of parents outside of the geographic area of the study resulted in a further reduction of the participant pool to 20 individuals to occupy 18 interviews. During the process, one person declined an interview and another

_

Mobility of parents suggests a concern regarding education performance. Residential moves and transfers to different schools have been associated with children dropping out of school, and is associated with negative impacts particularly for younger children (Alexander, Entwistle, & Horsey, 1997). Rumberger and Larson (1998) found that students who drop out are more likely to have changed schools before or during high school. Such research reflects even more negatively on children from reserve such as Fort Babine who have to leave home and their community in order to enter the high school systemThe research completed in this study also indicates that children from Fort Babine are ill prepared to enter higher grades putting them at an even greater disadvantage and increasing their likelihood of dropping out.

person who Lake Babine members believed to live in Smithers could not be located resulting in 18 possible families to be interviewed.

Prior to knowing the number of participants, which I did not confirm until late in the interview process, my intention was to select participants through a combination of purposive and random sampling (Boulton, 2005). When I discussed the project with the Chief of the Lake Babine Nation as well as the LBN Steering Committee one of their first questions and main concerns was that the research represents all Lake Babine communities. As posed in the LBN Steering Committee meeting "Are you going to do Fort Babine and Tachet? You will get a totally different perception from them" (LBN Committee Meeting Transcripts, October 17, 2008). In order to ensure this happened, the parents from Fort Babine and Tachet were purposefully included.

My intention was to determine the remaining participants through random sampling. Initially, in order to ensure all participants had a chance to participate, I assigned each child within the Prince George to Smithers area on the band list a number and used the computer program Research Randomizer to select numbers. I placed 18 names on the list and numbered the remaining people 19 to 29 as auxiliary if someone declined to participate. Rosa Johnny contacted parents associated with each number and requested the parent who assumed the majority of child rearing responsibilities or responsibility in education to participate in an interview. This process continued until 18 parents had agreed to participate. If parents had moved outside of the geographic area, I excluded them from participation and moved on to the next person on the list. Thus, it was not

until late in the sampling process, through directly trying to contact participants, that I knew my potential pool of participants was 18.⁵⁴

In addition to the mobility outside of the region, mobility of participants occurred within the geographic area and locating parents would not have been possible without community participation. The assistance of a community researcher in the process helped with tracking down participants as frequently this required knowing family lineages and contacting a parent, sibling or cousin when a person's address was not current or no phone number was available.

4.8 Data Collection

I used qualitative data gathering techniques as the most appropriate for this research because they are the best method for obtaining the detailed data required regarding parent's understanding, in this case of academic assessment. The purpose of the interviews was to explore topics and elicit responses that are embedded in the participants own perspectives. Qualitative open-ended interviewing begins with the assumption that the perspective of others is meaningful, which is an underlying theme of this research regarding the inclusion of First Nations parents in school processes (Patton 1990). Interviews involved collecting information from individuals who have special knowledge or perceptions that I would not otherwise have access to (Gall, Gall, & Borg, 2007). The use of open-ended (or semi-structured) interviews is appropriate because

It also appears that during the time of the study the movement was outside of the geographic service area. A review of those who had been excluded from the list initially provided indicated that none of those individuals had returned to the area.

there is no strong sense of First Nation parent perspective on assessment, so a format allowing unconstrained responses was used to provide the widest range of data collection. Because the research questions required in-depth parental engagement and an unconstrained exploration of issues, I used interviews rather than surveys or focus groups.

I collected data through face-to-face interviews, a method that is conducive to Lake Babine approaches to knowledge transfer allowing participants to reflect on their social reality *in situ*. Face-to-face interviewing is a culturally appropriate method of research with the Lake Babine Nation because of the personal nature of the technique and its maintenance of notions regarding the accountability of researchers, including myself, to the community (Boulton, 2005). I completed interviews between November 2008 and May 2009, during the 2008/2009 school year. Interviews ranged from 45 minutes to 150 minutes.

I recorded interviews and the interview format was semi structured, incorporating elements of quantifiable, fixed choice responses regarding demographic information as well as open-ended questions within the context of the target research area (Brewerton & Millward, 2006). The fixed-choice questions provided background information regarding the participants in order to provide participant context as well as differences and similarities within the participant group To mitigate possible discomfort with answering questions regarding socio-economic status I provided income categories to the participants.

-

⁵⁵ See Appendix C for a full copy of the interview questions and Protocol.

⁵⁶ See page 27 for guestions and page 30 for rationale stemming from literature review.

Frequently during the process, I also provided this sheet to the participant and had her/him circle the appropriate category.

I used open-ended questions to provide opportunities for the participants to discuss topics related to assessment in detail. If the participants had difficulty answering a question or provided only a brief response, I used cues or prompts to encourage the participant to consider the question further. I also enabled the participants to provide their own information within the interview by following lines of discussion introduced by the participant. While used to guide the process, the questions allowed parents to discuss assessment, as well as education in general, from their own perspective.

I wrote field notes after each interview regarding descriptions of events and notes regarding what people said, particularly if parents made comments when I was no longer recording the interview. Information documented included observations that would not be accessible through the transcripts such as: perceptions of the interview, thoughts about the parent's responses and general observations of parent's homes and actions of parents, and their comfortability with answering a question. I also recorded connections, or themes that began to emerge in interviews and areas to explore further in subsequent interviews to determine if there was a pattern to information that parents provided.

At times, parents asked questions regarding assessment prior to or after an interview and I recorded this information in field notes. For example, one parent who did not know about provincial assessment in Grade 4 wrote about the information provided regarding the Foundation Skills Assessment on a white

board on the fridge as a reminder to ask the school about the provincial assessment the next time she was at the school. Parents were often curious regarding provincial assessment and had a number of questions. Due to the reciprocal nature of the research noted above, I also saw the interviews as an opportunity to inform parents and have them think about assessment and its impact on their children. I answered questions regarding the intention of the provincial assessment strategy but did not want to persuade parents in either a positive or a negative manner regarding the exam. For those who indicated they did not know about provincial testing I provided a brief explanation regarding the Foundation Skills Assessment. The explanation consisted of a script informing parents the assessment is completed in Grade 4 and 7, it is compulsory for all students, it focuses on numeracy, reading and writing as foundational skills, it is used to evaluate how well students are achieving basic skills, and is reported at a provincial, district, school and individual level.

The tape-recorded interviews⁵⁷ were transcribed as soon after the interview as possible. Rosa Johnny, the research assistant transcribed the interviews and I reviewed the transcripts for accuracy. I identified transcripts by code only based on the interview date and the interview number, with identifying information kept in a locked cabinet separate from tapes and transcripts. If a parent used their child's name during an interview, I kept this information in the transcript but removed identifying information from any discussion within any public documents. In addition, I removed names of specific district schools from

_

⁵⁷ In each interview the participant consented to having the interview tape-recorded.

public documents as parent comments regarding a school seemed to be isolated to specific issues rather than systemic problems with a school within the school districts involved. If a number of parents identified a specific school within a district and I deemed this information important, I would have sought approval from the district to release the information.⁵⁸

In a meeting with the LBN Steering Committee after analyzing the information, I brought up comments made regarding the school at Fort Babine and asked if the information should be included because parents in Fort Babine and those who had moved to other communities, were critical of the school. There was consensus from the LBN Steering Committee that the information be included in order to draw attention to the issue. The LBN Steering Committee also stated the information was important because it provides them the documentation to take to School District No. 54 in order to develop a plan for targeting children who leave the community of Fort Babine to attend school. The LBN Steering Committee was clear that they wanted children entering the public system from Fort Babine assessed to determine their level of education. The LBN Steering Committee stated the information was known at a community level but the research reaffirmed the information and provided the evidence they required to meet with various levels of government to advocate for funding. The LBN Steering Committee discussed the issues faced by the communities of Old Fort, Fort Babine and Tachet in accessing funding because Indian and Northern Affairs Canada focuses on the larger administrative centre Woyenne. The LBN

_

⁵⁸ When the Band school was mentioned by various parents, I met with the Lake Babine Chief to discuss possible solutions and develop strategies for the school.

Steering Committee Stated that education is one of the major issues at the elementary level as well as those attempting to access funds for university. I also met with Chief Betty Patrick to discuss the information supplied by parents and sought her approval for the inclusion of information regarding Fort Babine.

I provided parents full transcripts of their interviews and gave participants three weeks from the time the transcripts were mailed to reply with comments. I made the participants aware of the reply time and process through a letter provided with the transcript. I also provided this information during the discussion at the beginning of the interview when I asked participants to consent to the interview being tape-recorded. ⁵⁹ If the participants did not reply by the date, as noted in the letter, I included the transcripts as complete and approved. None of the participants contacted me with comments or additional information.

4.9 Ownership of the Collected Data

In keeping with the OCAP Principles (ownership, control, access, possession) of data and information obtained from research involving Aboriginal peoples, I collected and housed the data for the period of time necessary to complete my dissertation (Canadian Institutes of Health Research, 2007). Upon completion of the dissertation, I archived copies of the research material in facilities designated by Carrier Sekani Family Services for those purposes in trust for the Lake Babine Nation. A protocol agreement with the Lake Babine Nation

112

⁵⁹ All of the participants consented to having their interview tape recorded.

and consent forms approved by Simon Fraser University outlined acceptable uses of the research material.

4.10 Data Analysis

Theme analysis was the primary analytical process used with the interview transcripts. The analysis focused on identifying a small number of qualitatively distinct categories regarding the manner in which participants understood (or conceptualized) assessment. The method fits very well with earlier discussion of Taylor and the understanding of how people imagine their social existence.

According to Opler (1945)

In every culture are found a limited number of dynamic affirmations, called themes, which control behaviour or stimulate activity. The activities, prohibitions of activities, or references which result from the acceptance of a theme are its expressions... the expressions of a theme aid us in discovering it (cited in Ryan & Bernard, 2003, p.86)

Within the process, I sought to discern patterns of understanding as well as describe differences in the manner parents ascribed meaning to the world around them. The aim was to reach a summary expression of the meaning of data as close to the participant meaning as possible. Content analysis involves developing a category- coding procedure in order to organize data around key themes and determine how well information fits into the categories (Gall, Gall & Borg, 2007). A typical content analysis consists of determining the frequency of the occurrence of each coding category within the sample. Opler (1945) suggests the importance of a theme is associated with how often it appears and how persistent it is across various cultural principles. Coding is the process of

combining data into themes, and categories so that I could compare and analyze information provided (Ryan & Bernard, 2003). The process of coding began with the proofreading of transcripts, which allowed me to understand key phrases and commonly occurring topics. The information typically fit well into the themes naturally created by the interview schedule, termed a priori coding and I used this technique as a starting point for determining themes (Reflected in Table 4). In some instances, the information provided by parents when asked different questions elicited similar responses and I combined this information into one theme. In this manner, the LBN Steering Committee assisted in developing a priori themes by determining what topics to cover in the interviews and how to query participants regarding the topic areas (Ryan & Bernard, 2003). Many of the topic areas determined by the LBN Steering Committee became the theme headings in the results chapter. The process also related to the validity of the inquiry through the process of collaboration and peer debriefing (Creswell & Miller, 2000). The LBN Steering Committee and Rosa Johnny, as collaborators, were involved throughout this research study as partners and the LBN Committee reviewed all chapters including the results chapter to determine if it accurately reflected the community. Rosa Johnny, who typed up the transcripts and sat in on the majority of interviews and thus was familiar with the expressions of participants, participated in validating and agreeing to how the themes were applied to the data (Ryan & Bernard, 2003). Patton (1990) terms this validating technique of agreement among investigators "triangulation through multiple analysts" (p.468).

Table 4. A Priori Themes

| Questions | A Priori Theme |
|---|---------------------------------------|
| What does education mean to you? | Education |
| How do we know someone is educated (Western)? How do we know someone is educated (Cultural)? | Western Educated Carrier Educated |
| What do you want your child to achieve from school? How should we determine success in school? | Achieve in School |
| What is the role of a school in your child's life? | Role of School |
| What kinds of information do you receive about your child? How often? When? | Information Received |
| How do you use this information? | Behaviour after Receiving information |
| When you think about your child's understanding in school, do you agree with the results of school tests? | Agree with Results |
| What do you think teachers are doing in the classroom to understand how your child is doing? | Teacher's Methods of Assessment |
| On an ongoing basis, what strategies do you use to check how your son/daughter is progressing in school? | Strategies used to check |
| What should passing to the next grade be based on? | Passing to Next Grade |
| How do you feel about provincial testing in Grade 4? | Provincial Testing |
| What do you feel would be the best way for teachers to find out how your child is doing in school? What form of testing would help students achieve in both Carrier and Western culture? | Additional Forms of Assessment |

I completed the organization of information from each interview into themes using various stages of analysis. First, I added all of the *a priori* themes identified to each transcript as headings and began a process of cutting and pasting information into each theme.

Table 5. Sample of Transcript Excerpt Edited for a priori Theme

PARTICIPANT NO. 04/08/09-015:

What does education mean to you?

Education means – I want them to live life to the fullest. I don't want them to struggle like the way we are right now. I try to talk to them about that too. We are living on low income and you guys won't like that. Just try talk towards them, that'll help them want to keep going to school. It's just hard right now not having an education and having 7 kids.

I would rather them finish off school and go to college, have a better education than I did.

The process included a word search using the edit- find tool in Microsoft word to search out a list of words that I identified as repeatedly used by parents in the interviews when reading and re-reading the transcripts. The word search assisted in identifying information that fit within themes but also when a parent discussed a theme in response to a different question. At times parents would provide information that fit within an *a priori* theme prior to me asking a question or would refer back to an idea they had discussed earlier so word search assisted in finding this information. Listed in Table 6 are words commonly

occurring in the text and the number of interviews in which they occurred. The number of times the words occurred is not intended to indicate that a certain theme is more important to individuals even though they were referred to more often (Sandelowski, 2001).

Table 6. Words Commonly Occurring

| Words Commonly Occurring in the Interview Text | Themes | Number of times Occurring | Number of Respondent s N=18 |
|--|--|---------------------------------|--------------------------------------|
| Assess (test) ⁶⁰ | Agree with Results Teacher's Methods of Assessment Provincial Testing Additional Forms of Assessment | Assess 31 Test 122 | Assess 11 Test 16 |
| Carrier | Carrier Educated Subjects | 96 | 16 |
| Communication (talk, speak) | Communication | 127 | 18 |
| Compare | Passing to Next Grade Challenging Carrier Students Provincial Testing | 21 | 14 |
| Educate/ Education/ Educated | Education Western Educated Carrier Educated | 149 | 18 |

⁶⁰ As noted in chapter 2, the LBN Steering Committee suggested parents would understand the word test rather than assessment. Seven of eighteen (40%) of respondents only used the term test in interviews. Sixteen of eighteen parents used the term test at least once in their interview. This is also evident in the number of times parents discussed testing (122) versus assessment (31).

| English | Achieve in School | 36 | 14 |
|------------------|---|-----|----|
| Grade | Achieve in School Passing to Next Grade Challenging Carrier Students Provincial Testing | 153 | 18 |
| Job | Role of School | 46 | 15 |
| Learn | Achieve in School Role of a School Subjects | 149 | 17 |
| Math | Achieve in School Subjects Passing to the next Grade | 59 | 18 |
| Read/ Reading | Achieve in School Subjects Behaviour after Receiving information Challenging Carrier Students Reading Group | 123 | 16 |
| School | Education Achieve in School Role of School | 427 | 18 |
| Teacher | Teacher's Methods of Assessment Behaviour after Receiving information Communication | 230 | 18 |
| Write/ Writing | Achieve in School Subjects | 50 | 8 |

I placed text that did not fit with determined *a priori* themes at the end of the transcript and re-analyzed the information against the other transcripts to determine if there were additional themes or commonalities between interviews. I used repetition as the technique to determine additional themes looking for topics that occurred and reoccurred in the material (Ryan & Bernard, 2003). This process resulted in the identification of four additional themes (*Communication, Challenging Carrier Students, Subjects, and Reading Group*) that I did not identify in the fixed set of open-ended questions. Once I determined the themes, I cut and pasted the information from each participant into a single document reflecting the theme.

Table 7. Sample Excerpt of Transcript into a Single Document

PARTICIPANT NO. 04/08/09-015:

What does education mean to you?

My son was home all week. He's not happy with the teacher that he has. He goes to [name removed] School. They have been sending my kids home every day at two o'clock and everybody else gets off at 2:30. He's tries to get there right at 2:30 to pick them up; but it's the traffic and he's always driving around. They said something about them going into a vehicle, but that only happened one time. I talked to my kids about it and they said they wouldn't do it again; but still they let them go at two. That last time they told them to go home at two, they made them walk home. When he gets angry, he has a real bad temper tantrum. Now this week they've just decided that they'll have him do his work at home. I've met with his teacher once, and I've met with the Principal. I want them to live life to the fullest. I don't want them to struggle like the way we are right now. I try to talk to them about that too. We are living on low income and you guys won't like that. Just try talk towards them, that'll help them want to keep going to school. It's just hard right now not having an education and having 7 kids.

Within the theme documents, I used additional techniques of coding including word repetition cutting and sorting, and compare and contrast (Ryan & Bernard, 2003). I searched for commonly used words and analyzed word use within participants to determine if there was significance to the Lake Babine

worldview and setting. Cutting and sorting involved using a word processor to cut and paste sections of transcripts into separate themes documents so I could further analyze and reread them for coding, as reflected in the sample within Table 8.

Table 8. Sample of Theme document – Code Identification

PARTICIPANT NO. 261108-001

Education

The future and education what I understand is *work*. I talk to my boys and if they ask me about it, like why they have to go to school? Tell them that schooling is very important. Nowadays things are in computer-wise and I believe now schools have computer classes in grade school anyways. I know high schools have always had them. I myself don't have any experience in computers.

PARTICIPANT NO. 11/26/08-002:

Education

To learn stuff. To learn the stuff that you need to know as adults! Like knowing how to live on your own and knowing how to cook for yourself. Knowing how to look for a *job* and learning how to be independent.

PARTICIPANT NO. 11/26/08-003:

Education

Education means a lot to me for my kids, because without education they will

not be able to get a *job* anywhere in this world.

Emphasis added to demonstrate the prevalence of association between education and employment

I identified quotes that appeared important and arranged them into segments that went together (Ryan & Bernard, 2003). Each time I selected and coded a passage or theme I compared it with all those passages already coded in order to ensure coding was consistent. This also enabled me to determine if passages corresponded or that the conceptions in the passages would be better coded another way (Corbin & Strauss, 2008). I completed coding and analysis of the data manually using thematic content analysis.

I interpreted results for theories and patterns and examined information for frequencies of responses as well as differences between respondents. I wrote short definitions for each code and assessed new information of a similar nature against the definition to determine if the information fit as reflected in Table 9.

Table 9. Example of Education Theme Coding

| Codes | Number of Respondents |
|-------------------------------|--------------------------|
| Employment | 15 |
| Technology | 3 |
| Knowledge/Training/ Skills | 8 |
| Improve Life | 10 |
| Graduate | 15 |

If the information did not fit, I created a new code. I analyzed Interviews against field notes. The themes arising in each interview were reviewed in light of new interpretations in order to see if anything was omitted (Boulton, 2005). As I reviewed information, I explored patterns in successive cycles of data interpretation and analysis. I repeated the process numerous times until I deemed categories exhaustive and I had a mass of results reflecting the major findings (Booth, 1997). In keeping with the methods of qualitative research and the need to include the voice of parents, I used a number of direct quotations in the results section.

The final process of the data analysis stage was providing the information back to the LBN Steering Committee for comment. The feedback stage provided an opportunity for members of the Lake Babine Nation to analyze the information through their lens and provide additional feedback on the themes. This process fulfilled the commitment to include members of the Lake Babine Nation in all

aspects of the study and as a data analysis strategy provided verification by the LBN Steering Committee of the information presented reflecting the views of the community.

One of the questions asked of the LBN Steering Committee related to the number of quotes used. Although the literature typically suggests explaining a concept while limiting the number of quotations in favour of simplicity, the LBN Steering Committee agreed that they wanted extensive use of quotes in the results section in order to provide voice to the participants (Cutcliffe & Harder, 2009). As one LBN Steering Committee member stated, "it would be cool if we can take this [result section] and say you need to read this document. It is going to be our evidence so I want every quote, every word of this kept" (November 6, 2009). I attempted to maintain balance between including voice of the participants and not overloading the reader with quotations. As agreed to by the LBN Steering Committee, I used at least one quote from each interview participant. The LBN Steering Committee reviewed the results section and provided advice regarding some of the information supplied by parents to explain Carrier approaches to communication, residential school experience and Fort Babine School. I added some of this background information regarding Carrier history to the results and introductory chapters.

5: CHAPTER FIVE: RESULTS

As noted in the introduction, the purpose of this dissertation is to provide voice to parents regarding their perception of education and assessment. The majority of parents involved in this research noted taking an active role in the education of their children, including working with children on homework and meeting with teachers to discuss their child's progress. The information provided by parents provides greater understanding of the attempts by parents to ensure their children have improved lives and in doing so challenges stereotypes of uninvolved First Nations parents. The results of interviews with parents demonstrate a need to maintain culture and knowledge of Lake Babine roots as well as recognition of the importance of Lake Babine children acquiring knowledge in subjects such as numeracy, writing and reading in order to succeed as members of the Lake Babine Nation within the larger society. The definition of success as well as education and educational achievement, as outlined below also reflects parent's divergent worldview that incorporates the need for varying

knowledge systems. Parents placed emphasis on subjects including math and English as well as culture and had a very practical view of education, indicating it is key to improved socio-economic status.

Parents noted communication is vital to understanding how their children were progressing and provided guidance to parents regarding areas for improvement and strengths so that they are able to assist their children. This chapter provides examples of positive interaction between teachers and parents including direct communication on a regular basis. Communication will also assist parents in understanding how teachers use assessment in the classroom as well as the use of other assessment methods such as large- scale assessment. More than half of parents did not know that provincial Foundation Skills Assessment would be completed in Grade 4 and many of the parents did not know how teachers assessed their children in the classroom despite feeling confident teachers knew how to assess children appropriately.

In structuring this chapter, I have attempted to utilize the natural flow of conversation within the *a priori* themes and have added the additional themes into sections of discussion that I believe fit best with the logical flow of discussion. Throughout the chapter, I have included examples of the coding for themes (Tables 13 – 13) in order to provide additional information regarding the coding process and reflect the number of participants who discussed a specific code.

5.1 Demographic Information

The study involved 18 parents of the 36 Lake Babine Grade 4 students in the 2008-2009 school year. While this study is not intended to be quantitative in nature, the demographic information does provide a reflection of important indicators that typically affect achievement of First Nations children within the education system such as geography, parental education levels and socioeconomic status. In all categories surveyed, the information for the entire Lake Babine Nation and the sample of this study are similar. The demographic information in this study corresponds to previous studies and information regarding the Lake Babine Nation education, socio-economic status and employment levels (Fiske & Patrick, 2000; Holyk & George, 2003; Indian and Northern Affairs Canada, 2009).

I completed nine of the interviews on Lake Babine First Nation reserves of Woyenne, Fort Babine and Tachet. The remaining nine interviews I completed in the communities of Prince George, Burns Lake and Smithers. Table 10 compares residency of the overall population to the sample population.⁶¹

_

⁶¹ In order to maintain anonymity I have not indicated how many interviews were completed specifically in each community as the populations in some of the communities is very small and may provide detail as to who was involved.

Table 10. Residency

| Residency | 2001 Census | | San | nple |
|----------------|-------------|-----|--------|------|
| | Number | % | Number | % |
| On Own Reserve | 1387 | 61% | 9 | 50% |
| Off Reserve | 903 | 39% | 9 | 50% |
| Total | 2290 | | 18 | |

(Indian and Northern Affairs Canada, 2010)

In terms of education levels of parents, 33%, 6 out of 18, of the parents involved in this study had completed Grade 12 or higher with a graduation certificate⁶². None of the parents had completed a university degree and two parents had completed college trade training. Table 11 reflects the similarity between the overall population and Sample population.

Table 11. Education Characteristics

| Education, | 2001 Census | | San | nple |
|---|-------------|-----|--------|------|
| | Number | % | Number | % |
| With High School Graduation Certificate | 140 | 26% | 6 | 33% |
| Persons with some post secondary | 75 | 14% | 2 | 11% |
| Total (over 15 years of age) | 535 | | 18 | |

(Indian and Northern Affairs Canada, 2010)

⁶² Two other parents completed grade twelve (Dogwood (6), Leaving Certificate (1) and GED (1). If partners of those who participated in the study are also included 31%, 9 out of 29 parents, completed Grade 12 or higher with a graduation certificate.

Although the majority of participants, 14 of 18⁶³, indicated they had enough money to support their basic needs, the level of income reflects many participants living in poverty (de Groot- Maggetti, 2002). Ten of 16 people (63%) who responded to the question of income noted that they made less than 24,000 dollars per year. The numbers are supported by the fact that seven of 18 (39%) of the participants indicated being currently employed⁶⁴. Six participants (33%), all of which are female, were single parents. The three men who participated in the study had partners to support them in child rearing. Employment rates and income level are similar between the sample and general population over 15 (Table 12)

Table 12. Income and Employment

| Income and Employment | 2001 Census | | San | nple |
|--|-------------|----------|----------|------------|
| | Number | Average | Number | |
| Average Income (persons over 15 with income) | 485 | \$12,294 | 14 of 18 | < \$24,000 |
| Employment rate | n/a | 28.7% | 7 of 18 | 39% |

(Indian and Northern Affairs Canada, 2010)

5.2 Education

In order to understand what parents believe are the important aspects of assessment, it is important to identify their views regarding education. When

⁶³ Three individuals stated sort of and one indicated no.

⁶⁴ A question was not asked whether a spouse was employed.

breaking down the a priori theme of education, codes included parents' conception of education, including being educated in a Western system and a Carrier system.

The parents interviewed for this study hold a very practical conception of education. Parents voiced the knowledge gained through education in terms of what their children would gain in the future rather than specific aspects of what is learned. Parents recognize the importance of education as fundamental to receiving employment and view education as the tools for a better life. When discussing what is learned, parents spoke of math and reading as the tools enabling their children to go out into the world.

Education is very important to me for the kids, because, myself I am going back to school right now. I told them that it was a struggle, even as an adult going back to school. Today in this age you need education, there's education in everything you do. Even if they don't do well, I told my son you can always learn to read and write, because my son has learning difficulties and I try to encourage him. He knows how important education is (03/13/09, 009).

A number of Lake Babine parents living in poverty view education as enabling their children to have a better life and as such placed emphasis on their child completing Grade 12. Many parents recognized that not completing Grade 12 in their own lives has resulted in barriers to employment and financial stability and identified that education was key to improved lives for their children.

I really never thought much about education, but nowadays you need education to get a good job. That is why I try to tell my grandchildren that they have to go to school and they have to get their education to have a better job, better than what your grandma has now. I am having a real hard time supporting my grandchildren right now (03/11/09, 008).

Education means, I want them to live life to the fullest. I don't want them to struggle like the way we are right now. I try to talk to them about that too. We are living on low income and you guys won't like that. I just try talk to them, that'll help them want to keep going to school. It's just hard right now not having an education. I would rather them finish off school and go to college, have a better education than I did. (04/08/09, 015)

It's more important for me to teach my younger one to do it because I never actually graduated. My family is encouraging her to stay in school and I am trying to help her with it. Most of my brothers and sisters haven't graduated as well. We are trying to encourage all of them to do it because we haven't really graduated. Education means...I am trying to teach her to, staying in school is leading to jobs and everything. We are trying to encourage her to stay in school, but everything is not going good right now (05/06/09, 017)

In all interviews, parents placed emphasis on children attending school and receiving education. In addition, 61% of participants (11 of 18), when speaking of what education meant, spoke directly of the importance of their children graduating. This finding opposes stereotypes generally held at a community level that suggest parents do not care about education. Responding to the words of parents below, the LBN Steering Committee noted, "We are all stereotyped that we are all the same. We have our own language and dialects. This [research] is what helps us advocate for who we are. Sometimes people will not believe you unless it's on paper. We tell our children that they have to keep up with their education" (Steering Committee November 6, 2009).

Education is the most important thing in your life, without it you aren't going anywhere in life, period... All of the kids need their education. Every child out there should have the opportunity to express themselves through their education and go on to further their lives and better themselves in life. If they don't go to school, like right now, if they don't go to university, who's going to hire them. Nobody wants you if you don't have Grade 12. If they don't have a Grade 12 and get a job, that is pure luck (03/19/09, 011).

That's the way it is these day! Education is a number one priority for kids, especially in the village here. I want my son to definitely be better than me, and my daughter for that matter. I want them to get their Grade 12 educations and move on to college and then move on to university and do something with their life. Even here there are good jobs, but you need Grade 12 and no one has Grade 12 around here, maybe two or three (03/30/09, 012).

Like every parent, I want them to do good and have a good future. I talk to my daughter about that all of the time. I really struggled in school and sometimes she does too, a lot of the times she struggles. We just work very hard at it. I talk to her about it all of the time. I was a high school drop out because I felt like I was dumb compared to the other students. I felt like I wasn't learning the way everybody else was learning at the same pace. I would be failing tests and it would be like a big issue. It was like a big stab at your self-esteem basically. You see people around you doing awesome and you can't seem to figure it out. I talked to her about that all of the time. You've really got to work hard, listen, and pay attention because some day...jobs aren't easy to come by these days and it's only going to get harder and harder to find them. Make sure that you have an education to fall back on and you have the opportunity to go to college or university; that way you are secure in your future (05/08/09, 018).

As one example of the emphasis placed on the importance of reading and education, one parent's living room walls consisted of 8x10 flashcards with each letter of the alphabet, words associated with the letter and a picture associated with the word, similar to those seen in a school classroom, in alphabetical order stretching around two walls. School pictures of her two sons from each school year attended were placed under the alphabet on the wall, highlighting the importance of education in that family. Another observation of the researcher was that the majority of parents had computers.

Parents interviewed in this study noted that people who are educated in Western terms have higher paying jobs and money to support their needs.

Parents indicated being able to tell someone's level of education by their

accomplishments including completion of Grade 12, attendance at university and material possessions they have acquired. Parents asserted being educated also means having the ability to communicate well and translates to personal conduct. Parents associated those who were educated with having higher self-esteem, caring about themselves and their appearance, and treating others with respect. Parents suggested people who are educated have increased life experiences because they are more readily able to go out into the world.

While parents stated that some of the characteristics of someone who is educated in the Western system are shared with the Lake Babine or Carrier system, there are additional characteristics of a Carrier educated person. The Western Education system is the formal education system that all children are expected to attend whereas the Lake Babine system is the learning from Elders and leaders to prepare children for their role in Lake Babine systems. As noted in Table 13, the most common responses centred on Carrier culture and practice.

Table 13. Carrier Educated Theme Coding

| Codes | Number of Respondents |
|---|--------------------------|
| Speak Carrier | 10 |
| Practice Culture (hunting Fishing) | 8 |
| How to Conduct Self | 3 |
| Community focused (social responsibility) | 3 |

In particular, those in the smaller reserve communities of Fort Babine and Tachet spoke not only of the need to maintain culture and language but also the Carrier value of respecting one another and assisting others in the community in order to maintain social harmony. 65 As noted by one member:

An educated Carrier person is one who does things for other people around the community. Like my grandmother said, if you do right to the people, do right to the community and do right on the job, if you don't bad things happen to you (03/31/09, 013).

Another aspect of being educated in Carrier systems noted by parents as important was the ability to speak the Carrier language and participate in Carrier practices.

They have got to speak Carrier fluently and if they know their culture and practice it, then you know they are educated (01/26/09,006).

To be educated in Carrier is to speak the language; it teaches you your culture and it's very important to know your heritage and your culture; where you come from and where your grandparents come from, and down the line. It's important for our kids to know. It should be the elders teaching them; even some of the things that we had to do to get our kids where they are today; and slowly they are phasing that out. The kids don't know who they are dealing with anymore. They forgot their parents, their grandparents, and their roots; their grass roots, they forgot all of that. That's hard on those kids, I am sure it is (03/19/09, 011).

It is how you conduct yourself. Carrier educated would be, that's a pretty tough question. I don't want to overstep the hereditary chiefs. You know it's, because a lot of our thing is potlatch and I respect that. Language plays a big part (03/31/09, 014).

While a number of the parents noted the importance of language and cultural practices, they also noted that they did not speak the language. Others indicated that they were unable to speak to this question because they did not

⁶⁵ All three of the responses regarding focus on community and social responsibility were from on reserve participants. Two of the responses were from the smaller reserve communities.

grow up in their culture or participate in such activities as hunting or fishing because they grew up off reserve. This research also supports findings in other research indicating that grandparents who were responsible for their grandchildren were much more likely to speak the Carrier language and speak of Carrier practice (Holyk & George, 2003).

As for Carrier education, I don't even speak my own language... Right now my boys are learning Carrier in school. But I don't speak it and it would be kind of hard to see if they are educated (26/11/08, 001).

I don't really know, because I've hardly lived around my nation. I've basically just lived in Prince George all of my life. Yes, it would be great for them to learn things that I didn't learn as a Carrier person, like hunting and all of that stuff. I didn't get a chance to learn all of that stuff. I want them to have a better life, a better everything (04/08/09, 015).

5.3 Achievement in School

The goals for achievement in school flow closely from parent's perceptions of education. The discussion of parents regarding achievement reflects the understanding of parents that their children must know who they are as part of the Lake Babine Nation but also have the skills to succeed in the larger society. As noted in Table 14, graduation, completion of subjects such as reading, writing and math as well as culture and language were all noted by a number of participants.

Table 14. Achievement Theme Coding

| Codes | Number of Respondents |
|--|--------------------------|
| Graduation | 15 |
| Completion of subjects (reading, writing, math, English, phys ed.) | 13 |
| Culture/Language | 12 |
| Attendance | 5 |

There was also concern that Lake Babine members are losing their Carrier beliefs and ways. Parents noted that in order to learn the language and culture the inclusion of classes in school must be throughout the education system and more than one hour per day. As one parent noted,

I mean in all of the schools systems around here, I know a lot of them offer small classes for teaching Carrier language. I know the kindergarten do quite a bit of it, but then once they get into the public schools then it's really half an hour a week, which doesn't really give them much to go by. I know my daughter was really good at Carrier language and then she got into the public school and was kind of was forgetting everything, like the colours and the seasons and all of that stuff that she had learned in kindergarten, which is unfortunate because I don't speak Carrier and her grand parents do. I think it's important that she learns to speak Carrier, living here yeah for sure (05/08/09, 018).

When parents who are able to speak the Carrier language discussed how they learned, they indicated that their parents spoke to them in Carrier until they replied in the same manner. It was hands-on learning of language and culture that parents suggested can be reinforced in the school but cannot be an add-on or special day at school but rather part of regular classes. Having Carrier classes

at school was also noted as a benefit because children who do not speak can learn together in a controlled setting. Many parents suggested they were intimidated in the community when people spoke to them in Carrier and they did not understand. The fear of being ostracized by Carrier speakers resulted in not asking questions or trying to learn the language.

At times, as noted in the comments below, several of the parents seem conflicted between the importance of Carrier and Western systems in their children's lives (also seen in the responses in Table 15).

Table 15. Theme Subjects

| Codes | Number of Respondents |
|---------|--------------------------|
| Math | 18 |
| Reading | 16 |
| Carrier | 16 |
| English | 14 |
| Writing | 8 |
| Culture | 8 |

Many parents viewed Carrier culture as a foundation for promoting selfawareness for citizens. In many cases, parents noted the depth of Carrier of knowledge required by children as being dependent on whether children would remain in the community and actively participate in community events. Parents stated completion of all grades in order to graduate was important in assessing achievement in school. Parents placed an emphasis on reading and math in order to graduate and in many cases move on to university with the requisite skills for enrolment. Many parents also noted the importance of learning the Carrier language and children understanding where they come from. Another area of importance indicated by many parents was the role of technology in society today and their children having computer and technological skills. Various parents suggested achievement in school included teaching their children social skills to communicate and treat others with respect.

Mainly that he understands everything. I want him to be up to where he could be and whatever grade he is in and a better understanding of what he is doing, math and English. As for Carrier, he wanted to take Carrier; he would be more than happy and that would be good. I guess they could introduce it more often in school, learning more about heritage and culture and maybe a language course of some kind for them, so that he knows where he comes from. If he decides to go that way in life then he will understand more about it, like the *Bah'lats* system and if he attends them or takes on a name. I would want him to learn life skills and social skills; like learning how to get along with other people. (01/26/09, 007).

The big thing is learning how to write properly, like sentences and reading is a big thing for me, and a bit of their own culture I would like them to learn. But that is something that is really hard for them to do; because I don't know too much about my own culture like most people. I think it should be taught in the schools for my daughter to learn. The reading and writing is most important for me because everywhere you go, you see that and it's better for them to do their work. I realized when I got into my Grade 11 courses it was like oh my god! I got stuck for a good month on one subject. That's more important for them to know how to read and write (03/13/09, 009).

As long as she knows where she came from and knows her culture and her history. We are at a point where we are not really considering our traditional ways within the *Bah'lats* system. When I look at it in 50 years, it either will be gone or it will all be done in English. I am not pushing Carrier on my daughter, because I know her plans are out there. But, if she is planning to stay within our community, then I would push the Carrier stuff

on her. Today and in the future, it is going to be nothing but the technology. That's how I see it. English, math, and now they have a course called "Career planning" and it wouldn't hurt for them to learn another language too. It is good for them to know Carrier. That will be good if they live around here. But, if my daughter is planning to go back to Europe, what good is Carrier to her. She will need some other language, like French is common. But, if she is planning to come back to Burns Lake to work, then she would need Carrier. My vision for my children is worldly (01/26/09, 006).

We need everything in life, calculating goes with math and English. Everywhere we go, we really need that and she really needs her Carrier. That's the main thing I want her to do is her Carrier because quite a few people know about it. I don't know I just want her to achieve in everything. That's the only thing I want her to know is math because I know down the road she will learn more about Carrier with my side of the family. I know she will succeed in that. The only thing that she always gives up is the math; that's the only thing I want her to pass in (05/06/09, 017).

Similarly, parents determined success at school to be completion of all grades in order to graduate. Parents stated graduation of children as paramount to determining success at school. Some parents built on this notion suggesting this meant graduating with a diploma in order to move on to university. Parents also reflected the Carrier view that each person has gifts to provide and affirmed that determining success was difficult because it is dependent on the skills of the child. Parents of children with special needs reflected this concept suggesting success meant enabling children to do as well as their abilities provided. Parents judged success by comprehension in a number of subjects such as English, math, and culture already noted in this analysis, but attendance also featured prominently in discussion of success in school. For some parents, they acknowledged assessment as a means for determining success as it provides information regarding how children are progressing.

I would like to see my kids graduate, to see both of my kids graduate. My niece will be graduating next year. It will be a great thing to see all of the native kids graduate from school. I thought about university for both of my daughters (11/26/08, 003).

To determine success in school should be by their marks and their grades, and grading their exams (03/19/09, 011).

For health wise, we need more physical education because a lot of our newest members are basically all obese. That means heart problems and diabetes and stuff like that. Our principal started a health program (03/31/09, 014).

Determining success in school would be hard to do because every child is different. My son can be good in English but won't be so good in math; just like my other son. It's hard to say that for other kids too, they exceed in certain things than other kids (01/26/09, 006).

5.4 Role of the School

Stemming from achievement and success at school, parents specifically described their view of the role of a school in their child's life. While parents affirmed academics as important within the role of a school noting "the role of the school in my son's life is to help him to become what he wants to be and to achieve his goals" there was recognition of the important social function that schools play in the lives of children (04/08/09, 015). As one parent suggested, schools are the caregiver for children, teaching them social values and protecting them for seven hours of the day for almost ten months of the year. In some cases, parents acknowledged that programs such as breakfast or lunch programs were the major source of nutrition for children whose parents are of limited income. For children at risk in particular, the school plays a large role in the wellbeing of a child.

Parents noted support for the wellbeing of children by teachers required having a strong understanding of the children and their specific circumstances. The cohort of this study had experiences outside of school that challenged their academic performance and behaviour in class including being a victim of sexual abuse, being transferred to care of the Ministry of Children and Family Development, and having close family members pass away. Without teachers understanding the children and their circumstances parents indicated they may be labelled and re-victimized.

In addition to the children's lives outside of school that affected learning, parents acknowledged wanting schools where every student felt safe and welcome to participate in all activities within the school. Parents stated bullying as a major factor in children not doing well in school or eventually dropping out. According to parents, bullying within the group of students had already impacted their attendance and willingness to go to school. The welcoming nature of a school was a theme among numerous parents.

They are supposed to teach the kids to all get along with one another. Teach them to get along when they grow old and when they are on their own, that they know how to communicate with others. They should learn respect for each other. It starts at a very early age (03/19/09, 011).

A successful school will try to make everyone feel welcome (05/06/09, 017)

I've heard a lot of bullying that's happening in a lot of schools. Children should feel safe; because some of the ones that are falling behind more is due to being pushed around and put down, not by the school teachers but by other children (03/31/09, 014).

5.5 Information Received

Assisting parents in understanding how their children were progressing, parents observed that the school provided various sources of information including direct contact report cards, letters to parents, newsletters and a daily journal.

Table 16. Theme Information Received

| Codes | Number of Respondents |
|--------------------------------|--------------------------|
| Report Card | 13 |
| Journal | 9 |
| Direct Contact | 7 |
| (meeting, phone) | |
| Work from Student | 7 |
| (home work) | |
| Letter/paper notification | 3 |
| Child specific | |
| Newsletter | 3 |
| Information from an assessment | 2 |

Many parents noted the journal signed by the parent and teacher that children were required to bring home daily, or in some cases weekly, as the most important tool for information exchange. A number of parents viewed the journal as a manner of daily communication and integral to knowing how their child was progressing. Parents preferred sources of information such as journals for the

balance they provided as opposed to information such as a letter from the teacher that only came home when a child had done something wrong. Other tools such as report cards parents mentioned as coming home every three months and many of the parents who discussed report cards stated that they could not remember when teachers sent the last one home. The usefulness of report cards was dependent on the specific student and the level of information provided. Some parents suggested that report cards were less useful because they only contained numbers, which did not assist in providing parents with ways to assist their children in the same manner as the journal or direct communication from the teacher. Others suggested the report card was useful because it was very detailed and specific to their child and provided sufficient information regarding areas for improvement.

Report cards are much better now than at [her old school] because now if she can't get the help in the classroom she can get it at home. I think we get three reports cards a year and the teacher is very detailed and it's specific to her obviously. It says exactly where she's struggling with and what she is having a hard time with (05/08/09, 018).

Report cards, all it has is numbers, the success, it's very good and very bad, that's the only thing, no notes or anything (04/08/09, 015)

Journals were effective to the extent that parents and teachers kept them up to date. In a number of instances, children no longer brought home their journal or when they did, teachers were no longer writing in it. In such situations, it appeared parents and teacher had a communication breakdown as parents recognized the journal had not been home or they had not received information and yet they had not contacted the teacher. It also appeared that teachers in

these examples no longer wrote in journals or conversely did not investigate why parents had not written in the books for an extended period. Again, as noted above regarding communication failure, the children who were in most need of assistance and stated by parents as often being in trouble at school, were the students indicated as not bringing their journal home, resulting in another example of children who are in most need of assistance falling behind. In these cases, parents still indicated wanting to know how their children were progressing and the usefulness of the journals but did not contact the school.

I'm pretty sure that they have been sending stuff home with her, but I'm pretty sure that she's been throwing them out. I'd rather for me to know who I am talking to, it's better to have it face to face. But I am pretty sure that she is doing pretty good, because the teachers haven't phoned me and told me anything. The only time that I had it was a couple of months ago. They do journals, it's a little agenda book; saying this is what they did today and I had to sign it She sends it back saying that it was okay with me. But I haven't seen that book for the past couple of weeks now, I asked her a couple of days ago where the book was and she said that she didn't know. She probably left it in her backpack at school. (05/06/09, 017).

They actually don't tell me, like when they brought him home the other day, all he said was that he was throwing his books around and talking back to the teacher. The only thing that they leave out is why. They don't tell me why he's doing that. The Principal is always phoning me, the teacher, I haven't heard from the teacher at all. It should be the teacher phoning me and letting me know what is going on. The communication with the teacher needs to improve. That way we can put both heads together and see what's causing this trigger with [my son]. They have a student planner, but lately they said they don't use them anymore. I asked [my son] about it and he said that too. They started and then just in the middle of the school year I don't see them back. I don't know what to do. He's still in that same class, same teacher, same students. The information, it started off with what to do at home, what work needs to be done, now it's just not that way any more. I did what was written on there, reading and his math and all of the work that needed to be done. To make it easier for me to be involved in his progress would be one-on-one with all of us together (04/08/09, 015).

They have journals but she does not fill them out. Then they don't send any homework home with them, they just send them home. When they do send something home, it's one of the school's papers announcing the best student in the school and he isn't even on the list. Now why send me that if he's not even in it? The journal that he used to bring home was brought home daily. But now there is nothing in it. I send it back to school with him every day; and they don't even put the homework or what he needs to catch up on. We receive very little information about our grandson from the school, only when it is bad stuff. I'd really like to know how he is doing in school and if there is something there going wrong here that can be improved, I'd like to know. It would give us the heads up to help him out in school or help the teacher out if the teacher is having a problem (03/19/09, 011).

5.6 Behaviour of parents after receiving information

Parents responded to the information received by teachers in a various ways depending on whether the information was positive or children were in need of improvement.

Table 17. Theme Behaviour after Receiving Information

| Codes | Number of Respondents |
|-------------------------|--------------------------|
| Talk to Child | 8 |
| Talk to Teacher | 6 |
| Provide Assistance | 6 |
| Review work | 4 |
| Reward | 4 |
| Praise | 3 |
| Get Extra External Help | 2 |
| Punish | 1 |

When the results were positive parents praised their child, offered encouragement and provided rewards⁶⁶. Rewards included gifts such as a new video game, money or IPods as well as more time on the computer or video games. When the results indicated that children were in need of improvement parents used various strategies including analyzing assignments where the child had done poorly for possible causes, inquiring with their child or contacting the teacher to determine reasons, reviewing homework and providing additional assistance, and following recommendations of the teacher.

_

⁶⁶ In some cases parents used multiple methods such as talking to the teacher, talking to their child and providing rewards. Each response was coded in each response category.

Recommendations from the school included placing their child in additional reading classes or obtaining one-on-one assistance at the school. The actions of parents reflect Carrier values in communicating and allowing children to find their own way. Even when results were poor, parents indicated use of positive strategies rather than punishment. One parent indicated the use of punishment in the form of not allowing a child on her video games when a report indicated she was not listening.

If the information is bad, then I help him out at home as much as I can and if it's good, I praise him with an hour of a game or whatever. Yes, he gets rewards for doing well in school. The teacher and I talk weekly sometimes, so if he is doing real bad or if he's doing a real good job she will phone me. I ask how he's been doing and how his attitudes been doing at school and ask if he's listening 'cause sometimes he just sits there in his own world and he'll play with the kids. I always ask how he's doing, if he's listening and everything (11/26/08, 002).

By praising them and showing them that they are doing well. A happy child that is doing well in school and is being told by their parents they are doing well in school will go further than one who doesn't get that type of attention (03/19/09, 011).

If she's struggling in a certain area, we try to get education into her life without her really knowing it. She's doing all of this stuff in math. Normally we try to teach her how to do that stuff without her knowing. Like if she is struggling in a certain area, if she is struggling with reading....try to make it fun. Sometimes I don't even notice myself doing it. If she wants to bake I try to get her to read me the ingredients and how much is 3/4 cups. I am really just trying to help her in the area that she is struggling but try to do it in a way that she doesn't realize that she is learning something. I think the schools should make it fun. If something is fun then it is easier to learn. A learning process is something that you'll be able to have self-confidence in. If you don't have that confidence, you are not going to gain from it. I work really hard with her when it comes to doing her spelling test because I want her to go in feeling like, I know this; with that kind of attitude. If she gets one wrong its okay. If she gets two wrong it's okay, no matter how many she gets wrong it's okay to her, just as long as she had the chance to get it wrong and know the difference and learn from her mistakes (05/08/09, 018).

5.7 Agree with Results

The majority of parents agreed with the results of information sent home and felt that it accurately reflected how their children were doing in school. Some parents agreed with the information but noted wanting additional information so that they could assist their children on a regular basis. As noted above, parents had some concern that teachers should better understand what some children were going through outside of class and its impacts on behaviour. However, when teachers sent home information indicating such things as a child being easily distracted or active in class parents agreed it was correct.

I agree with the results of school tests and feel they are pretty accurate. The report cards show what level she is reading at. I am pretty sure that the teachers are doing their job that they are supposed to be doing (11/26/08, 003).

Yes, I agree with the results, because I know she's been struggling these last few years, but I agree with the marks she gets now. I'm happy with the marks that she's got now; there is no complaint there, because she's at a C+ average and a B average but that's where I'd like to see her and that just goes to show how much she's trying as well and how much she does want to learn. I'm happy that she does and I'm happy where she's at, the grade is good. They are not just B.S.ing saying that she's got a C+; and I agree where she's at with that. She got a B in art class; because I know she wouldn't get a B in math, maybe math but not in social studies, science or English. Especially English, I know she won't get a B (03/13/09, 009).

Yeah I agree with how he is doing. [My son] was a good student until he came to Grade 4 and all of a sudden all of these problems came up and I got really mad and I complained. So maybe that's why I get more information from them (01/26/09, 006).

Most parents viewed teachers as professionals who have been trained in assessing children and as a result should know how to assess the work of their

children. In many cases, it appeared parents agreed to suggestions by teachers without specific explanation or understanding of rationale. According to the LBN Steering Committee, this practice is common within community members who have often been conditioned not to challenge those in authority (November 6, 2009).

5.8 Communication

Stemming from a priori themes of information received was discussion of communication, the most often discussed theme by parents in this research. Communication is an important component of parents obtaining, and using information received about their children, and parents viewed communication as central to them being involved in assessment. Parents wanted as much information as possible in order to know how their child was doing but also to assist them in the proper manner. Parents suggested that marks from exams or numbers on report cards were not as useful in communicating because they did not provide areas where a child could practice. Parents wanted the school to be more accessible including more scheduled visits to the school and parent teacher nights as well as teachers sending home information specific to their child that provided what some parents termed inside information. Parents indicated that increased face-to-face interaction is beneficial because it enables parents to keep informed of progress. There was varying amounts of information provided to parents depending on the teacher. In some cases, parents did receive

information such as an overview of what students completed during the week while other parents indicated such information would be useful.

The schools could make it easier for us to be involved in testing his progress by sending us inside information from throughout the year. If they are not working with the parents, there will always be a loophole and lots of kids fall through that hole. You don't want the kids to fall through the hole; and those that do fall through the hole wind up druggies or alcoholics, or suicidal (03/19/09, 010).

I think it would be good, I know it will be a lot on the teacher, but if the teacher could write down how he's doing in certain subjects. The more often, like a letter coming home more often. More parent/teacher interviews maybe, I've only met with her once so far this year (01/26/09, 007)

Parents indicated that most often they went to the school when they were invited including events such as parent tea and parent teacher night. For some parents there is a perception they should only speak to the teacher, or will be contacted by the teacher if there is an issue or concern with their child.

I have met their teacher when they have Christmas concert and when some things are going on. I wish I could be there every day to check on him. They usually have days where we can go in and meet with them, but I haven't been doing that lately (03/31/09, 013).

To make it easier in school they should have more parent/teacher nights. They only have those once a month. My husband always wants to talk to the teachers (03/30/09, 012).

Although all caregivers of children indicated they were comfortable entering the school and had spoken to their child's teacher, grandparents in particular also shared stories of their harsh treatment within the school system, which provide rational for levels of mistrust. There were also events such as parent teacher interviews that were rushed or discussion a parent tried to have with a teacher regarding bullying that were perceived by parents as not being

deemed as important by the teacher that contributed to parents level of trust.

Most of the events were specific to interactions with particular teachers with
some parents indicating that the relationship was much better the year before.

The amount of communication and parent's perceptions of ability to communicate with teachers and others at the school ranged significantly between parents. The range of communication varied from daily communication through phone calls and journal entries to meetings only once during the school year. Parents who had worked for the school in various capacities indicated that they were comfortable speaking to their child's teacher, and often related that teachers treated them well because of the previous relationship suggesting other parents may not receive the same treatment. Parents who indicated their children were doing well in school were often the parents who also indicated that communication with the school was good. The weakest communication appeared to be with children who are in the most need of assistance and not doing well in school. Parents with children who were struggling or who had been in trouble at school indicated a need for improved communication so they could assist their children. They viewed themselves as advocating for their child when things had not gone well. Similarly, when parents noted their children were not doing well or continually in trouble, parents perceived racism as one of the reasons. While this may not be the intention of teachers, history and current encounters are very important in their impacts to the perceptions of parents and some of the information provided by parents is very important in framing that perception and thus noted here.

For a while there, I thought that all this time that I was doing everything wrong; with all of the punishment that I have been getting. Now that I am older, I realize that I never have done anything wrong. But I have a problem, I use my fingers to subtract or add. They come around behind you and with the tip of the ruler, they hit me on the head and told me "Use your head." So I got tired of him hitting my fingers, so I put it under the desk. Life was harsh back then, in school. I remember being thrown out of the bus in over .40° below. The bus was packed and these white kids were shoving me around and everything; so I turned around and shoved one back, and a few of them fell. The bus stopped and the bus driver kicked me off of the bus. I wasn't even dressed for winter, nothing. I walked home and by the time I got home, my fingertips were all hurting. The next day they were all hurting and they all turned black; my ears all turned black because I got frostbite. My ears were all peeling and I lost all of my nails, so my grandmother kept me away from school for six months. She kept treating me at home; she wouldn't even take me to the doctor or nothing; she didn't trust anybody. I always talk to my children and grandchildren about this. I always talk to my grandchildren and to make sure that they tell me if they have problems at school, like if somebody ever bullies them or anything. When it comes to schooling my ears are shut, because the school that I went to; I'm not really leaning towards school because they damaged me (03/11/09, 008).

We're dealing with the residential [school] thing. All they are doing is turning it over to the other schools and its still going on. It's not being dealt with at all. It has never stopped. It has gone on. It's no different than the Catholic schools. It's the same treatment I got when I was in the Catholic school. How are we going to stop this cycle? It has to stop. We have to be able to sit down and talk with the people involved. You have to treat people like you wish to be treated (03/19/09, 011).

I have met her teacher, but the meeting did not go well because when I dropped her off she didn't have much time to talk. No, I don't feel that I can go and talk to my daughter's teacher any time because when I tried that with the bullying thing I had to be real quick. I don't feel that the school is very welcoming, because they say to check in at the office. The teacher also said that we did not need a parent/teacher interview. They should let us know what is happening at school regarding testing. I would like more information such as teacher interviews (01/12/08, 005).

I've noticed with the Grade 4 teacher, there really is no communication and at the parent/teacher meeting she just kind of wanted to sit there and push us out the door. I heard her while I was outside and she said that they were falling behind, you better look through this and she called us in and didn't really introduce herself. Even at the open house it was the same, she didn't come to approach me and introduce herself. She just

stood in the corner and let me browse around with [my son] and his class. She wasn't really someone, I don't know to me it seemed like she was prejudiced. My cousin had a problem with her stating that she always had negative things to say about her son. I thought it was just because her son is active, but now that I know to witness what she is doing. We were in there less than five minutes. She quickly went through the papers and that was it. She asked if I had any questions and I said I want to see how far behind my son is in reading and if he's really accomplishing; and if things are being done about it (26/11/08, 001).

The suggestions of mistreatment contrasted with a number of parents who expressed a strong relationship with their child's teacher and school and should serve as examples of best practice. Examples of communication below demonstrate how parent and teacher can work as a team to ensure students complete assignments. Strong communication also aides parents in knowing how their child is progressing and areas where they can assist their children to improve through homework. In one example below, if not for communication, the child the teacher viewed as not grasping a concept in math and the child may have been believed to be struggling by his parent and teacher. Parent experience at school as well as education level did not appear to be a determining factor in current relationships with a child's teacher. Parents who had completed Grade 12 or college also reported negative experiences with their child's teacher.

The teacher contacts me if, if she asks him to do his science. Like last week, he was supposed to do his science work. All [my son] said was that it was too boring and did not want to do it. She phoned me up that same day saying he should stay to complete it and just stayed in school for 20 minutes more because he had to finish his science. So, they tell me everything, whatever is going on with him in school. They always have direct contact with me weekly. I just talk to his teacher to see how he's doing. If he's not doing well then I'd help him more at home. The last time that I met with [my son's] teacher she said that he was two years behind in

his reading and it's been awhile and now he does at least an hour of reading before bed time (11/26/08, 002).

He did have some problems in math but I always participate in parent/teacher interviews. There was a case where he was doing columns adding and he had gotten a whole page wrong and I looked at it because I knew like in reading he goes left from right so what he was doing was he was adding from this side and carrying it over this way. So I explained to the teacher, this is what is he is doing and she was just like oh you're right and stuff like that. He did a whole paper but she let him redo it after I brought that up but if I didn't say anything he would have got a zero. She would not have noticed because he did the whole page, starting from the left instead of the right (11/27/08, 004).

For me, the teachers and me, we communicate regularly. We try our best to make it work for her and if we realize that it isn't working then we always get back to each other and try to figure out if we should do this for her or another program or if we should stop the program that she is in. I met my children's teacher and during the first month I met her every day. After that, probably about 40 times and I phone there regularly especially when they miss something or forget something at school. She phones me once in a while too. So the communication is very good (03/13/09, 009).

For many of the parents, particularly those whose children were having difficulty, the suggestion of having more First Nations role models in the school is important to making it easier for parents to be involved in education. This has been a long-standing concern and one that many school districts have worked to improve. The parents wanted not only Aboriginal liaisons but also teachers within the school. In some instances parents suggested that the Aboriginal liaison had the difficult task of being the communicator of bad news when their child was in trouble or suspended from school.

I think for my grandchildren they should have a Carrier person to assist with the children in school. I try to help them here. I think this has something to do with their low self-esteem. I try to let them know that they should try not to let that get in the way and to be proud of who you are; no matter what they say (03/11/09.008).

What I think that would be better for the children is if they had an Aboriginal worker or Aboriginal teacher to have a program in the school for some of the students that have learning difficulties in every area. Someone who could help them improve in those subjects and have afterschool care. Something that will help them improve their interest in that subject or the course (03/13/09, 009).

I strongly believe that we should have our own teachers in school. Maybe if they were seen, present right in school where they got used to it with other youth it would be more frequent. We don't live alone in this world. That's why I was hoping maybe we can have a few of our members get their teaching certificates, whatever is necessary. I'd like to see more of them in school, not just as a T/A (03/31/09, 014).

5.9 Teacher's Methods of Assessment

Parents appeared unsure of the methods used by teachers to assess children in all aspects of education despite agreeing with information sent home. When asked what parents perceive teachers are doing in the classroom to understand how their children are doing, 11 of 18 parents had a response while the other parents indicated they were unsure or did not know. However, parents were confident that teachers were properly assessing their children. Raising the question sparked interest in the assessment process and all parents suggested they would be interested in knowing the various tools used to assess children. The most frequently suggested methods parents did indicate were being used included talking to the student, watching and listening, tests (spelling), and examining the child's work. The responses demonstrated knowledge of some of the methods used including performance based assessment and summative assessment but also reflected uncertainty.

The only way I could think of is that they talk to him and look at his homework to see if he's doing it right. Overall, I guess how his assignments are (01/26/09, 007).

I don't know about the testing. I can't say because they do test everything, so they are being checked on how they are progressing in their subjects on a weekly basis. I know that they go through a weekly testing in the school and to see where they are at completely is to do a full assessment on them; like every term I guess (03/13/09, 009).

Teachers are probably communicating with the students. There is 26 of them in one class and there's two teachers and extra helpers so they get one-on-one when they do a subject together. He always asks for help and they help him. Probably look back at his work. Ask him if he needs any help or if he's having a hard time learning in a subject (05/06/09, 016).

5.10 Strategies used to check

In addition to information provided by the school, parents used a number of strategies to check the ongoing progress of their children including talking to the child, talking to the teacher and reviewing assignments, which assisted parents in understanding and agreeing with the results of school assessment provided by their child's teacher. The most often used strategy at home, as noted in Table 18, was assisting children with homework and having them practice math or reading skills in order to see how the child was progressing.

Table 18. Strategies to check Student Progress

| Codes | Number of Respondents |
|--|--------------------------|
| Assist with Homework (work with child) | 18 |
| Review assignments | 11 |
| Talk to Child | 8 |
| Talk to Teacher | 5 |
| Review Assessment (Report Cards | 2 |

All but one of the parents participating indicated that they spend time working on homework with their children.⁶⁷ In the additional case, the parent indicated that her mom (the student's grandmother) assisted with homework and her goal was to take a more active role once her current situation had settled. I asked follow up questions to determine reliability of responses including, type of homework worked on, areas in need of improvement, how parents assessed if children had improved in areas such as reading and math that parents noted as working on. Based on the follow up questions, all but one parent was able to provide specific information in order to respond to the questions, indicating involvement in homework. Parents used a number of methods to determine student progress including seeing how many math questions their child could complete in a specific amount of time and the number of correct answers, determining how many errors a child made while reading and determining

⁶⁷ The term "their children" is used to indicate guardianship as grandparents with responsibility for a child were also involved in the study.

comprehension of a story by asking questions after a passage was read. Parents also noted the growing vocabulary of their child, noted in one case as the number of five-dollar words understood.

Every day when I get home from work, I usually ask if they had a good day and they usually tell me. I get them to read something to me, even if it's just a sentence; because one of them, his English, he's having trouble with his English and then I ask him questions (01/26/09, 006).

I usually talk to him a lot and ask him some questions about different things, things that he's having a hard time with. I also read over his homework as well to see if he's actually doing it (01/26/09, 007).

But you have to always read a lot, so he agrees with us and he grabs a book every now and then and reads it. I told him a book, not an NHL book. They want him to get into more novels and all that. We read part ways, and we get him to read two pages, and we get him to read silently on his own. If he doesn't know a word he'll either asks us or he looks it up on the computer (05/06/09, 016).

The level of input into homework by parents was also commensurate with the level of education of parents. While some parents stressed the importance of education and homework, and sat with their child and listened to them read or practice other skills, they were not as comfortable assisting directly with homework. While this was not as much of a concern for their children in Grade 4, parents with children in later grades noted the progressing difficulty in providing assistance. The high school completion rates provide a key area for need in supporting parents. As one grandparent noted,

I didn't get my education. I quit school in Grade 5 and I learned as I went on. I tell her that I will try because I am not a good person to ask what word is what and I told her that I dropped out and didn't finish Grade 5. She writes down all of her letter things. Now she's learned how to do hand writing. I make her practice at home with her writing. She writes letters and poems. One time she told me that when she grows up she wants to be an artist (03/11/09, 008).

5.11 Challenging Carrier Students

As noted throughout this discussion, Lake Babine parents recognized the importance of education to the future wellbeing of their children and attempted to assist them in any way possible. As part of the strategy for success, parents want high standards for their children that include equal participation in education. Parents also related equality to ensuring standards are consistent throughout the province so that children are able continue their education at the institution of their choice.

I'd like to see the kids all be treated the same, all have the same learning standards and participate in the same subjects. Work together in class, a social network, and have the teacher chart their achievements for the day, even if it was one area. The teacher in Vancouver was very good at doing that, when he didn't bring his homework home she would call and wanted to know why. Strong communication is the key (03/19/09, 010).

I want him to achieve top marks, I want him to be treated the same as everybody else and I want him to be prepared for later in life; like to reach post secondary and whatever training he wants to get into (11/27/08, 004)

We would like him to achieve the highest standard of education. By that, I mean high grades, good, good grades so that he can go on and he would be accepted in any school anywhere without hesitation because you have that high standard of education and good ability to learn and that would bring him, probably open up to the world. Now everything is going computer and stuff like that, where your education is really important (03/19/09, 011).

I hit the roof when they told me that [my son] was a special student, because it never showed and all this time there was nothing wrong with him. I believe they are doing this to other parents as well (01/26/09, 006)

The goal of parents was to ensure that their children could meet typical standards for children in Grade 4. One concern indicated by many parents was the discrepancy between the education received in the public school system and

those on reserve. Parents discussed the right of children to be educated in a manner that allowed them to continue to university and suggested that children who attended the on reserve school were denied this right based on the quality of the education provided. There was concern regarding funding levels that resulted in out of date textbooks and teaching materials.

Carrier First Nations children are more advanced nowadays, then they were back in my day because we were all in [Reserve] Schools. They didn't try to push you or try to make scientists out of you. My children should be tested the same as anybody else. I'd like them to be tested the same way as the other students in school, because I want my kids to try to push themselves. To try to learn and if it's hard for them then they'd let me know. I would like my children to be tested the same way as anybody else and the only reason why I say that because I went to school in Burns Lake and I was going to the school down town. The schoolwork that they were giving me was the work that I did in Grade 1 and I was in Grade 4 (03/13/09, 009).

He should have that choice, to be able to graduate Grade 12 and he should be able to go to any university. He should have that choice as well as anybody else. There is no reason he can't. But if he's going to be treated like this, through his smaller classes for Aboriginal students he's going to rebel and he's going to walk out of school and he won't make it to Grade 9 or 10, already he's frustrated (03/19/09, 010).

Lots of kids go to school in Fort Babine; and when they come to Smithers to start into high school; they fall them all the way back to Grade 5 because they are not qualified to go to grade 7. Those kids from Babine, they put them to grade 7 and they have to fall all the way back to Grade 5. If they are in Grade 7 and get tested, if they fail, they have to go back to Grade 5 again. There is not enough education for them down there. They just need more education down there. Fort Babine goes from kindergarten to Grade 6. I have two staying with me and it's really hard for them to learn. They are really interested in the school, but they always complain the work is too hard. Most of them they put in the learning centre because they couldn't learn in a regular classroom. They didn't learn enough at Fort Babine. They just need better education and they need to learn more about it. It seems like they are not teaching them the right skills down there. That's why I have my kids up here from the beginning (05/06/09, 016).

When I was in school in Immaculata all I learned was religion, which was it. I didn't learn anything more than that, which is probably why I didn't finish school. Subjects that are most important are math, history, and maybe science; and physical education (04/08/09, 015).

The education is not, on a scale of 1 to 10 I would give them a 4. I'd give them a 10 if they were doing all of the above. There are people here that are my age, who can't read or write and they went to this school. They are not educated like they should be in a regular school. Everyone gets the same homework. Out here I guess you're lucky to go school, but the people here don't know how to read and write (03/30/09, 012).

5.12 Extra Reading Classes

One of the most commonly mentioned issues by parents (12/18 interviews) that I did not elicit through a direct question was discussion of extra reading classes provided by the schools to Aboriginal students. The discussion of extra reading classes by parents tied to assessment strategies, challenging Carrier students and communication. While many parents noted the extra reading classes as beneficial, there was also concern that they were solely for Aboriginal students and parents were concerned that students may fall behind in other areas when teachers took them from class for the reading class. Reflecting assessment and a lack of communication, many parents were unsure as to how they determined their child needed extra reading classes.

In most cases the only communication to parents by the school was a letter that indicated their children would receive the extra class if parents did not respond indicating they did not want the class. Two parents did note the use of an assessment as the basis of inclusion in the class but were unsure of the indicators and had not seen further assessment that specified whether their child

had progressed or how teachers would determine if their child had achieved standards and no longer needed the classes. Other parents contested the results when teachers indicated their children needed additional classes because their child's marks did not reflect this need and these parents viewed the classes as solely based on their children being Aboriginal. One parent indicated that when she complained about the school placing her son in the extra class he was not required to attend.

British Columbia began to allocate funds in the 1980s under the English as a Second Language Policy to support Aboriginal students (Battisti, Friesen & Krauth, 2009). Between 1999 and 2004, there was a rapid expansion in schools that began accessing funding for Aboriginal students. School districts receive an operating grant English as a Second Language/Standard English as a Second Dialect supplement of \$1100 per student (Battisti, Friesen & Krauth, 2009). While districts have discretion in the services provided, districts are required to conduct an annual assessment of proficiency in Standard English for each student, and design an annual instruction plan that lists services the student will receive to improve proficiency (Battisti, Friesen & Krauth, 2009). Battisti, Friesen & Krauth (2009) argue funding has been highly effective at improving the reading test score gains of Aboriginal students between Grade 4 and 7. Parents were unaware of this more global information regarding the procedures and impacts of extra reading classes (I investigated after the interviews) and were unaware of specific assessment completed on their children. Parents, who often read with their children, did not attribute reading improvement to the extra class.

He was assessed for his reading and his math but I don't know the process that they used to do it. But they did tell me that he is one of the Aboriginals that needed help in English. They had funding for it in school, so they assessed him. They said that he was eligible for extra English (11/26/08, 002).

Some letters were sent home from school and I took a bit of insult to it because she said that [my son] was recognized as needing this extra assistance. So, when I met with the teacher and asked her about it I asked her how he was recognized and it was basically based on his ancestry, the colour of his skin. They had recommended [my son] to go into these extra classes and so I went in and I asked them to show me where he needed it and they couldn't show me. So I said well he doesn't need reading and an Aboriginal support worker. I said English is his first language, because he doesn't speak Carrier. So, I never signed any consent. They said on the letter that if they didn't receive any comment back they would automatically take it as your approval. So, this year, he got the same letter, and the content was changed a little bit, and he receives this letter every year. There is not official assessment his marks are the same as everybody else, he's an A/B average and they couldn't explain to me how he was recognized as needing this extra help. No, he doesn't need that extra help. I think it's based on funding, right? I mean if he needed I would, but he doesn't need it. That is why I went in and asked how you guys recognize [my son] as needing it. It was basically because he was Aboriginal (11/27/08, 004).

She is in a special class for Aboriginals because that is one of the things that they got me to sign. This is for reading and writing. I haven't gotten anything about how she is doing in the classes (01/12/08, 005).

I know that she's taking some extra classes, like a reading group to help her with her reading because that's been her struggle ever since the beginning. I know that in those classes she also brings that kind of stuff home where she's been in this reading group and they send home flash cards with words that they think she should know or is struggling with. I don't know exactly what they've done to assess her to put her in that class, but it is helping her and is benefiting her. I never thought to ask if that class is only for Aboriginal students. They got consent for me to do a test, Peabody test I think, to see where she was and that's when she started in these classes to help her with her struggling in reading but I never thought to ask if it was only for Aboriginals. She's been in that extra class since the beginning of Grade 2. I'm not too sure when they meet the criteria. I never, every time I get her report cards it says that she is not meeting her expectations yet so I never really questioned it. Actually, I never really thought of that but now that it's being brought up. It would be

important for them to let her know how they have come to assess her for this class (05/08/09, 018).

5.13 Passing to Next Grade

Parents viewed passing to the next grade as an extension of challenging Carrier students. The majority of parents (15/18) indicated that passing to the next grade should be based on achieving determined skills and standards for the grade. They often stated that they did not want their children to be pushed through the system without learning the requisite skills. Parents suggested assessment is an important factor in ensuring the children had completed the standards for each. Parents also recognized that skills built on each other and if children had not learned the information from early grades they would fall further behind.

Passing to the next grade should be based on everything; their attendance, their participation, their marks, their homework assignments, like their whole contribution to the class, I think. If he wasn't ready, I wouldn't want him to move on until he was ready because that is kind of setting up for failure. If he wasn't ready to move on to Grade 5 then he shouldn't (11/27/08, 004).

Passing to the next grade should be based on her education, that she understands and learns everything. If she does not understand it, why pass her up? I am not in favour if they just push a child through to the next grade even if they don't understand. They have always done that to our children (01/12/08, 005).

Passing to the next grade should be based on if they can do the work. They have this new thing where they don't keep students back and they just keep moving them. But they got to do it with discretion and they have to keep an eye on those, on the kids like my son. I don't want him to end up in Grade 7 and find out he can't read. We better find out right now if there is something wrong with him. That way we can work on it before

they move him; because I know they are going to keep moving him. You see now the thing is that they keep moving them no matter what. But there should be a little string attached, saying that he is ready to move on. We might have a Grade 12 student who can only read at a Grade 7 level and that's because they move them along. I would see some sort of progress report, instead of just pushing them right into Grade 7 (01/26/09, 006).

While parents believed passing to the next grade should be based on achievement many also understood the damage being held back could cause as it was seen as punishment and in that manner did not fit with Carrier values. The education system also demonstrates a difference between Carrier and formal education learning systems. In Carrier teaching there is not the same emphasis on age as a determinant of preparedness. Parents indicated that within the Carrier system, children are watched and when they have demonstrated a skill, they are allowed to move to another skill. The Carrier system respects the ability of an individual to learn at one's own pace.

That would be hard. It would be nice if he's done better. Passing to the next grade provides something to look forward to, like he was so happy to pass to Grade 4. I'd rather have him at grade level before he moves because if he falls behind he won't know what to expect in the next grade. If he fails, it feels like I failed (26/11/08, 001).

Passing to the next grade, should be based on the children, but some of these kids are in school do not meet the actual grade level. Like for my daughter she is not a Grade 4 level and the teacher told me that she does not meet the Grade 4 level. But she is doing a whole lot better this year than the last couple of years (11/26/08, 003).

There again, a lot of people learn differently, we're not all identical. Some probably feel more comfortable staying longer than others and some are, like I said earlier, are really quick in the books and speed passed everyone else and are sitting there totally bored out of their mind waiting to graduate. That's the scale, probably be slower than others and a lot quicker than others. Passing to the next grade should be based on, you know that's like a humane question to answer. You don't want to be

pushing somebody back when they are not quite at the level that the government expects them to be at. That's a hard question to answer. Maybe a little of both, I don't want my children not knowing nothing at all and then passing into Grade 12 and graduating. I remember when I had to go to special ed., I was ashamed to go in there (03/31/09, 014)

I don't think, like thinking back to my old days they didn't say ok now you can cut fish, it just came to us. It was just presented to us and we had to do it. There is no such thing as going from stage to stage, we were just exposed to it. We just learned and we were just immersed in it (01/26/09, 006).

They taught us how to prepare our own foods, and how to sleep and eat out in the wilderness, and how to survive by ourselves, and how to live well on our own. They know you are ready because they groom you at an early age. By doing so, you proved yourself to them because they watch you pretty closely. Otherwise, they would never let you have a gun or snowshoes. There were certain things that you couldn't do until they said that you could do it. Observing a child learning tells you whether he can carry on into something higher (03/19/09, 011).

5.14 Provincial Testing

Parents viewed provincial testing as one mechanism for ensuring their children were prepared within the education system. It should be mentioned, that only seven of 18 parents knew about provincial testing or that the process would take place in Grade 4. Parents were often curious regarding provincial assessment and had a number of questions. I answered questions regarding the intention of the provincial assessment strategy but did not want to persuade parents in either a positive or a negative manner regarding the exam.⁶⁸

_

⁶⁸ As noted in Chapter 4, for those who indicated they did not know about provincial testing I provided a brief explanation regarding the Foundation Skills Assessment. The explanation consisted of a script informing parents the assessment is completed in Grade 4 and 7, it is compulsory for all students, it focuses on numeracy, reading and writing as foundational skills, it is used to evaluate how well students are achieving basic skills, and is reported at a provincial, district, school and individual level.

Specific to the topic of provincial assessment raised during the interview, one parent was very interested to hear of the provincial assessment and indicated that she was going to talk to her friend, who also has a child in Grade 4, about the provincial assessment. The interview also heightened her interest in assessment and education. Her friend's child attended another school in Burns Lake and she became curious about the type of programs her friend's child was receiving, and what type of reporting that school was sending home. Another parent wrote provincial assessment on her white board at home as a reminder to ask the school about the provincial assessment the next time she dropped her child off and saw her teacher.

There was consensus among parents (18/18) that provincial testing was a useful tool in order to ensure that no children fall through the cracks and the level of education a child receives is similar anywhere in the province. In one instance, caregivers were concerned that their grandson was reading at a Grade 1 level in preschool when living in Vancouver and after moving to Burns Lake for four years is now below averages in Grade 4. In their example, they also stated that the school suspended their child during the time of the Foundation Skills Assessment and he did not take part.

They have got to catch them now. That's where all of the learning begins; the important learning begins in Grade 4. Primary is more for preparing them for Grade 4, but once they hit Grade 4 they have homework and the whole works; and that's where the testing begins. It would help me, at least concentrate on what they need. It would give me a better understanding, what I used to do at home. Get them to where they should be (01/26/09, 006).

I think if they're in school, everybody should get the same education and if they're falling back there's a problem I think it's important to know how the kids are doing at a young age, with my daughter. I find that to get at her now would help her out in the long run. I think that it's a good thing to get at them at an early age, if we can get her now, we can help her with the headaches in high school. If you can't read something, it's going to be difficult for her in high school and I don't want that for her (03/13/09, 009).

I want to know if he's on their level or what? He doesn't have to be smarter. I didn't finish school and when I tried to go back to complete my first aide level three I never realized how hard it was to go back. Staying with it would have been easier (03/19/09, 011).

Many parents viewed provincial testing as another tool in their understanding of how their child was progressing and felt it was important to know how they are doing compared to other children in Grade 4. Seventeen parents indicated they want to know their children are progressing compared to other children and provincial standards. The conception of parents was that the information would allow them to work with their child in areas that needed improvement. Parents, at times added the caveat that knowing how their children were doing compared to other children was only important so that they could assist them in a particular subject because children learn at their own pace and ability.

I think the provincial testing is a good thing. I agree with the provincial testing. All of the above is necessary for the education of a child. I want to know how he is doing compared to other kids, to see if they are learning what Grade Fours should be learning (03/30/09, 012)

The report cards are okay, but I am in favour of knowing where he ranks. That would help out even more, to see where he is at (04/08/09, 015).

I have heard about the provincial testing. I don't mind it, they want to see how the kids are progressing (05/06/09, 016).

5.15 Additional Forms of Assessment

Parents commented that tests as well as other forms of assessment are required to help children achieve in school. Similar to questioning regarding how teachers assess, parents were unsure as to assessment methods that schools should use, with ten parents having a response. Parents who did respond wanted various methods applied recognizing the importance of determining that children understand concepts. One parent indicated that hands on and oral assessment should be used to determine understanding, recognizing that some students' reading comprehension is low. Parents recognized that methods for assessing children's understanding of math, English or Carrier language may require different methods depending on the knowledge teachers assessed.

Regular tests are also useful because like I said we don't live alone around here, because we do have to eventually leave the reserve to get employment, even special needs students. All forms of testing are important (03/31/09, 014).

As for forms of testing, everybody is different and I feel that, I noticed in her, her personally it's seems like she learns a lot better when she has the opportunity to actually do it instead of just writing answers. She needs a good example where she can go out and produce a result then that's how she learns. Maybe they should put more of that into the classroom, where they are not just writing and thinking of everything in their head (05/08/09, 018).

Parents did agree communication was vital to assessment strategies.

According to one parent, assessment must not only be summative, it must provide explanation to children and parents regarding areas they can practice and improve. Parents again noted that communication to both students and parents was vital to the process so everyone understood and could work together

in achieving goals. The majority of parents suggested the best way to communicate with children to understand how they are doing and relay instruction is one-on-one. Parents expressed concern that in bigger classes it is more difficult to determine if children are grasping concepts in the material. They noted having time to work directly with the student and assess them on a regular basis, providing opportunity for children to demonstrate they understand what teachers are teaching.

I think the teachers have to keep contact with the parents like they do with me because we are partners in their education. That way we both know what happens and what should be done, what homework should be happening (01/26/09, 006).

The teacher seems to understand how he's doing, I mean she collects all of his work and she goes to him one-on-one. If she sees something or if I mention something and when we discussed his math problem, she said she doesn't want to ask [my son] in front of the whole class, so that he feels bad about it. Stuff like that she will take him aside and talk to him (11/27/08, 004)

Parents advocated for one-on-one methods as important because many of the students lack the self-esteem to ask questions or indicate that they do not understand. Parents suggested confidence issues were amplified in First Nations students who often have self-esteem concerns, particularly regarding how they are perceived by non-First Nations peers, due to existing stereotypes. Parents indicated teachers should be cognisant of self-esteem as an important consideration in assessment as well as teaching styles. Parents specify that assessment must be done in a balanced manner that does not contribute to impaired self-esteem. Similarly, parents were concerned that testing may result

in labelling of a child that would impact them for life, including the damage caused to self-esteem.

You can build a child's self-esteem up with the report cards. Information is mostly bad, something negative and never when he does well. He should be able to complete the work that he is given and if he can't they should find out why? Maybe there is a little problem there and they should work on that little problem. Sometimes a word or those three letters on the end, does not look right to a kid that age. That could make a whole difference for that whole year. Sometimes they know it doesn't look right but they can't tell the teacher that (03/19/09, 011).

She said she gets all stressed out and tired because there is no one-on-one for her at school. They need to be shown that they are special and they can learn just as much as any other children. My self-esteem is way low and I don't think too much about myself. There is nothing I can do about it. I couldn't even finish, I just couldn't (03/11/09, 008)

I think that one-on-one time would be the best thing. As a teenager I remember not wanting to ask for help, because of that reason, feeling embarrassed to ask in front of peers. There are constant barriers when it comes to school. You feel embarrassed to be asking in front of your peers, especially when you are around a bunch of people who look at you thinking how did you not know that? You know that kind of attitude. Like I said, there needs to be more one-on-one. I felt like I didn't want to ask, maybe if someone asked me how I understood or something then maybe it would have been easier. It was hard for me to put my hand up and ask for help (05/08/09-018).

Parents suggested testing should inform practice and enable children to meet expected outcomes rather than indicating to a child that they are unable to complete a measure. Assessment should not be based on a deficit model and should be completed with the assumption that children have skills and can complete tasks if taught in a manner that makes sense to them. ⁶⁹ In some cases parents stated that formalized testing had a negative impact on their children

_

⁶⁹ This is not to suggest that parents were unrealistic about some of the learning disabilities faced by some of the children. They indicated the difficult job of teachers in meeting the needs of all students and making sure all met their potential based on such factors.

because of the manner it was completed. Parents wanted opportunities for their children to practice prior to testing so that they were prepared. They also noted the high-pressure environment that testing created and the impairment it causes in many children when taking exams.

If you're testing a child and they know that they are being tested, they are going to be nervous. If there was some way that they could at least practice for it so that they have more confidence going into it. They are not just sitting there completely dumb-founded at what's going on; with a book slapped in their face. As a student, all students are entitled to study for any test. Why are you putting these little, tiny children in a position where they are under so much pressure for something that they haven't had the opportunity to learn. I think it really helps with my daughter's confidence when I'm able to help her with quizzes and stuff like that. Get her knowing the word and in the morning and the test is coming up she says that she can't wait for her test, I know how to spell this and I know how to spell that, because she had to practice for the week (05/08/09-018).

Parents with knowledge of Carrier practice suggested one-on-one instruction and assessment methods are common to Carrier practice. According to parents, Carrier teaching and assessment is based on a model of exposing, instruction, and watching and listening, all of which fit well with parents' definition of methods that should be used.

My mom would speak our language and she would say that she does not understand us until we speak the language. We can't get anything until we speak our language. Most of it was watching while I learned. She would look at what I just did to see if I did it right (01/12/08, 005).

She just told me that I was ready, because I went out with her every day and learned everything. She wanted me to watch how she did it and I would do it. She sat there and told me. She showed me and watched me do it and then let me do it on my own. (03/11/09, 008)

Parents indicated that teachers or teaching assistants should monitor those who are struggling more often to ensure they are progressing. Parents

continued, stating that teachers should continually assess methods of teaching used in these cases to determine if they are the right methods.

I want them to know that they can do it because with me with education, there is always a way to learn. If there is one way that they can't learn, then there is always another way. I'm sure that the teachers who teach know that (03/13/09, 009).

5.16 Final Remarks

This chapter provides a snapshot of the perceptions of parents of Grade 4 students within the Lake Babine Nation and what they deem as important indicators of their children's achievement in school. Specifically, parents link achievement in school to the preparation of children for university and employment. Parents recognize that an indicator of preparation for university and employment is completion of Grade 12 with the requisite skills in each grade. Parents want their children to be able to coexist within the Lake Babine Nation and the larger society and the practical skills required for coexistence include reading, writing, math and Carrier language and practices. Parents also view social skills as an important indicator of achievement.

Parents recognized the important care giving, socializing and preparatory role schools play in the lives of their children and wished to be partners in the process. Key to achievement of their children was partnership. Parents described the importance of communication between school and parents so that parents understood how children were progressing and the subjects taught so they could assist at home. Parents also viewed communication as important so the school

understood what is happening in the lives of children outside of school and the impacts external issues have on their children's behaviour and ability to learn.

In terms of understanding of academic assessment, parents often did not know how teachers were completing assessment in the classroom and few parents knew about provincial testing taking place in Grade 4. Parents agreed to the importance of various methods of assessment, including one-on-one time and large-scale assessment, to determine how children are progressing. Parents viewed assessment as a tool for determining subject knowledge. They wanted high standards for their children and did not want their children pushed through the system without meeting standards. One concern parents relayed regarding assessment was doing so in a manner that does not damage children's self-esteem. According to parents, testing should inform practice and assist children in meeting desired outcomes rather than only providing information regarding areas where children are not doing well.

6: CHAPTER 6 DISCUSSION, RECOMMENDATIONS AND IMPLICATIONS

6.1 Caveat

The sample size of this study is very specific and is meant to reflect the culture, geography and local knowledge of the Lake Babine Nation and may or may not be generalizeable to other First Nations. The schools, the First Nations parents and the students who form a portion of the school system are heterogeneous and need to be treated as such. This research recognized the lack of parent dialogue regarding assessment and intentionally sought the perspective of parents in order to bring their voice forward. The research is intended to illustrate, from the perspective of parents, the possible disconnect between policy, services and the perceptions of parents. For example, while strategies may be in place at a school and district level to improve such things as communication, the strategies were not always known by parents in this study. I provide recommendations to enhance work already being done by schools that are reflective of the views of the group of parents interviewed. Areas for further

research include looking at the specific components of effective communication for educators and Aboriginal parents; examining educator perspective on communication; and examining educator perspective on the assessment of Aboriginal students.

6.2 Discussion

In this dissertation, Lake Babine Nation parents of Grade 4 children deem communication as a primary tool that links home and school, facilitating feedback of their children's achievement in school. The two main outcomes or indicators parents noted as wanting their children to achieve from school were graduation and completion of subjects including math, English (reading, writing), and Carrier language and culture. The goal of these parents was to ensure that their children could meet typical standards for children in Grade 4. Completion of subjects and graduation should be connected because Lake Babine students may ultimately complete Grade 12, but parents believe that many have done so with a leaving certificate rather than a Dogwood diploma, limiting their post graduation options and reflecting a lack of basic skills. Parents were concerned about children being pushed through the system without the required skills in each grade and felt that passing to the next grade should be based on achieving determined skills and standards for the grade. Parents emphasized education as necessary for obtaining adequate employment and viewed education as a tool for establishing a better life. They recognize that in order for their children to improve their lives they must graduate. Many of the parents in the study who did not graduate

recognized the barrier this created to employment and financial stability in their own lives.

Parents supported measures such as provincial testing to ensure the level of education a child receives is similar anywhere in the province. Parents viewed provincial testing as another tool, along with other classroom-based measures, in their understanding of how their child was progressing. They suggested aligning of the local context with provincial or national assessments could help the district and parents understand where their children fit within the larger society.

Parents who did respond to questions about the best assessment methods wanted various methods used, recognizing the importance of multiple approaches to determine understanding. The LBN Steering Committee noted methods including hands on and oral assessment should also be used to determine understanding, recognizing that while some students' reading comprehension is low, they may understand the overall concept. Parents distinguished that assessment will vary depending on the subject (e.g. math, English or Carrier language), and that teachers must recognize the appropriate tools of measurement, engaging in various assessment methods.

Historically, as noted in Chapter 2, the role of assessment has changed. The democratization of education and employment demands have shaped assessment methods in the past and must continue to adapt to meet the needs of First Nations students. The demographic composition of the Lake Babine Nation, as with many other First Nations, is a young population representing a large group entering the school system and workforce. Parents noted they

wanted their children prepared for employment and felt that assessment methods that do not ensure Lake Babine youth are performing as well as their peers throughout the province can have far-reaching and damaging effects on Lake Babine young people, their community, and Canadian society.

This research has implications for the research design of others interested in working with marginalized groups. The research methods serve as a template for communication, trust and understanding required for an outsider to build relationships necessary for working with First Nations, whether in research or in service provision. Communication included discussion with leadership regarding the utility of the topic and inclusion of community members in research design, question development and data analysis. Relationship building required participation in events outside of the research project and a demonstrated commitment to learning.

6.3 Key Findings

Communication

Parents identified communication between the school and home as an important component of student success in school. Communication is an important component of parents obtaining and using information received about their children. Parents also viewed communication as central to them being involved in the education system and the assessment of their children. Parents wanted as much information as possible in order to know how their child was doing but also to enable them to assist their child in an appropriate manner. One

observation based on the level of education of some of the parents and their identified difficulty in assisting children in Grade 4, is that communication to parents should be easily understandable and provide specific directions, recognizing literacy issues among some parents.

Communication Formats

Schools provided various sources of information including report cards, letters to parents, newsletters and a daily journal. Parents overwhelmingly noted the journal signed by the parent and teacher that children were required to bring home daily, or in some cases weekly, as the most important tool for consistent information exchange. Direct conversation is also useful, and in some cases was very frequent, however this may be difficult to maintain with each parent, and the journal provided up to date information on what children were doing in class. A number of parents viewed the journal as a manner of daily communication and integral to knowing how their child was progressing and what they were being taught, and thus what parents should focus on at home. For parents who discussed report cards, they noted considering report cards useful in obtaining information about their child but also noted information could be clearer and more specific. Child specific information, learning needs and tutoring strategies offer parents direction on areas where they can provide assistance at home. As noted by the LBN Steering Committee and addressed by two parents in the study, communication with the parent is vital if teachers are to understand the

circumstances of the home-lives of each child and act accordingly to meet the needs of the child in the classroom.

Communication Frequency

The range of communication varied between participants—from daily communication through phone calls and journal entries to meetings only once during the school year. As a result, communication activities may be strengthened through monitoring at the school level, or development and enforcement of a district wide policy on parental communication. The children who were in most need of assistance and stated by parents as often being in trouble at school, or having poor attendance or marks, were most often the ones who lacked a strong communication network between parents and teacher. When problems occurred and children did not bring their journal home or the teacher stopped providing information in the journal there was no mechanism for ensuring communication took place. Parents knew the communication stream was broken, but did not know how to effectively re-establish contact or did not feel empowered to act. 70 Assisting in rebuilding relationships and opening lines of communication may be an area of action for the Aboriginal support worker.

Additional sources of communication that allow direct contact with the parent, including electronic communication or regular phone calls could be examined as they might help mitigate the student responsibility for bringing home important paperwork. An observation of the researcher was that the majority of

⁷⁰ This is the parent perspective. The teacher's perspective may be different regarding attempts to contact parents or reasons for not contacting parents.

parents had computers. Such a system may have benefited various parents in this study whose children told them that nothing was being sent home. As many mediums as possible that will help parents feel involved with what their children are doing are critical for maintaining parent-school connection.

Schools also have to ensure that communication is built into plans because despite wanting to be involved, and indicating that they were comfortable going to the school (two frequently mentioned barriers in previous studies), many parents only went to the school when invited to events. In American research, the amount of parent involvement in school-sponsored events was also linked to poverty, with parents living in poverty less likely to attend events including open houses and parent teacher meetings (Carey, Lewis & Farris, 1998). With a large portion of parents from First Nations communities living in poverty, diversity of strategies for communication must be considered.

Historical Factors impacting Communication

Due to past experience at residential school and priests and nuns being positioned as ultimate authorities, many Lake Babine parents have been trained to sit back and observe rather than being forward (Fiske, 1981; LBN Steering Committee, November 6, 2009). As a result, they may not go to the school unless invited. Additionally, their behaviour while at the school for meetings may be reserved, as they may feel it is disrespectful to challenge the teachers or be ill

equipped to do so.⁷¹ Passive conduct is much more common in a situation where parents are out of their own community in a place such as a school, which parents still see as dominated by people outside of the Carrier culture (LBN Steering Committee, November 6, 2009). Passive nature can also be perceived as apathy—an assumption shown to be incorrect based on this research.

That standing in the corner comes from residential schools. I am a survivor and you have been told to keep quiet and sit in your place and if you speak up watch out... At the meetings at the school, I wonder how many people think they are in trouble because they are called to a meeting?.. I'd rather sit in the back, it's a big challenge just to be there (LBN Steering Committee Meeting, November 6, 2009).

The classroom itself is a convergence of Carrier and non-Carrier systems in which past experience has shown the teacher as professional and someone to be listened to but also follows cultural norms where there is an expectation that the teacher is the expert. Teachers should engage parents and initiate conversation, much as a knowledge holder at a community level would be the one to engage listeners.

This does not suggest that agendas and communication be unidirectional but rather requires reciprocal decision-making. Pushor and Murphy (2004) argue typically the agenda is set by the school, and parents serve that agenda receiving information from the schools about their children and their programs but rarely asked to give information. Pushor and Murphy (2004) continue

Given the privileging of expert professional knowledge, combined with infrequently scheduled opportunities to meet with parents, teachers often take control of the agenda at meet the teacher nights, at parent-teacher conferences, or in meetings with parents... In such moments teachers

⁷¹ It should be mentioned that Lake Babine parents have a variety of styles that reflect the diversity of the Nation and this information is discussed to provide suggestions reflected in the study.

review curriculum, report on student progress and share concerns about student learning or behaviour with little time provided for parent input (p.234).

As identified by participants in this research, communication with parents and First Nations communities is needed to build trust, and is important when addressing any specific student issues. Parents and First Nations community resources are a key contact for understanding children as well as allies in preventing misbehaviour and conflicts. Teachers must learn about community in order to understand how parents impart expectations of their children. Communication strategies will help parents to understand what is going on at school and to clarify doubts when there are problems. If the relationship is solidified and communication is in place early, educators and parents can be united in addressing concerns when they arise as opposed to parents feeling that they are only contacted when something goes wrong. Harris (2009) suggests that educators must ensure that children experience a learning context that is continuous between home and school. Collaborating with parents in the education of their children will ensure continuity between education emphasized at school and at home. Parental involvement in the lower grades will help children to create positive academic habits, do homework, prevent absence, promote positive attitudes toward school and help to maintain an orderly school climate.

Communication to Help Parents Assist Their Children

The results chapter reflected that Lake Babine parents want the best for their children but may not be able to provide the guidance to reach goals such as university or other post-secondary institutions. All children had at least one caregiver who worked on homework with their child or grandchild. However, the education levels of parents suggest they would benefit from partnerships with the schools⁷² to help them better understand school assessments and provide specific feedback regarding areas of focus and tutoring strategies they might use to help their children. The level of input into homework by parents in this study was also commensurate with the level of education of parents. While some parents stressed the importance of education and homework, and sat with their child and listened to them read or practice other skills, they were not as comfortable assisting directly with homework. ⁷³ While this was not as much of a concern for their children in Grade 4, parents with children in higher grades noted the progressing difficulty in providing assistance.

It is imperative that schools provide parents with the tools necessary for assisting their children in maximizing their potential, including adequate information in a format parents can understand. Inadequate preparation of parents is an obstacle to the effective use of assessment. Teachers must work with parents to ensure parents can accurately use information for encouraging

_

⁷² By schools I mean specific individuals in the schools (whether teachers, Aboriginal Support Workers etc.) because relationships were noted as key in parents seeking out support.

⁷³ Reading at home is noted by Rumberger (1995) as a positive indication of parental involvement in schooling, which has been shown to be associated with improved school performance.

their children to achieve and assisting them in areas where they require assistance.

Trust

The colonial history, as reflected in Chapter 3, has aided in the disruption of Lake Babine cultural, political and economic systems resulting in mistrust or apprehension that is magnified when parents are not involved in their children's education. The worldviews of the Western system are embedded in the education system and at times conflict with parent's views. Parents of the Lake Babine Nation live within two cultures and worldviews, and work to reconcile differences in ways of understanding. Despite historical issues, parents have a degree of faith that the current education system is a tool for an improved life.

Trust of Schools

Parents noted wanting to be involved in assessment and gain a better understanding of the methods used. Despite historical issues, parents believe educators were properly assessing their children and wanted various methods used to ensure a holistic approach to measurement. Parents noted involvement in assessment as an important support for their confidence that the schools were maintaining high standards for their children. They wish to know the specific indicators educators are using and how their children are progressing measured against those indicators. They also wanted to know how their children are progressing compared to other students in order to promote equal participation.

The negative parent reactions to the imposition of extra reading classes found in this research provide one example of the mistrust that can occur when parents are not informed of assessment methods and progression of their child. When programs seem to be tracked to First Nations specifically, parents may see them as unequal. While the literature suggests the importance of added reading classes, parents in this study did not understand or believe in the benefits of such classes, nor did they think they would lead to student improvement.

Trust of the Education System

Parents suggested passing to the next grade should be based on acquiring skills and were concerned that many Lake Babine children were being pushed through the system without an adequate skill and knowledge base. While the research on retention and promotion is complicated (and beyond the scope of this study), parents clearly expressed the need for access to data that can help them understand the progress of their children, and approved of standardized tests, such as the Foundation Skills Assessment, as one measure of assurance that their children were meeting appropriate grade-level standards. Similarly, parents see the Dogwood as an important indicator of achievement.

Can you imagine if you thought your child was doing fine and then find out that you are actually not? It's no wonder they don't finish Grade 12. It's time that we demand that the school tests the children. We have a right to demand that. That's where we can voice stuff like that. We should not sign Learning Enhancement Agreements until we demand the proper attention for our children (LBN Steering Committee November 6, 2009).

Parents in this study expressed the importance of cultural knowledge as well as subject knowledge. As such, districts should look at ways to equalize the

treatment of cultural knowledge, and this includes consideration of assessment strategies. Parents noted wanting an education system that meets their needs in terms of curriculum that is reflective of Western and Carrier knowledge and parental involvement ensures these goals are met. Based on the comments of parents in this study, educators must work to ensure students are academically successful while maintaining and developing their First Nations identity. This perspective is strongly supported by previous research (Battiste, 2000; Helin, 2006; Watt-Cloutier, 2000).

Harmony and Social Responsibility Connected to Trust

Harmony and social responsibility were core values of learning noted as important to parents of Lake Babine Nation, particularly on reserve. Parents indicated the connection of harmony and social responsibility to issues such as bullying that impact the culture of the school and their children's academic performance. Parents felt bullying resulted in their children rejecting the school system. They suggest bullying impacted marks and, if not dealt with, eventually may lead to their child dropping out. Past treatment is still a concern for parents and parents viewed bullying as one aspect of a continuation of historical racism that reduced trust in the school and teachers who did not address the situation. Parents also deemed that if teachers did not deal with their concerns teachers were perpetuating racism and devaluing their children resulting in a loss of trust.

6.4 Recommendations

Communications Systems

To address many of the issues identified in this research, schools may wish to involve First Nations communities in developing a formal parent communication system that: 1. Defines communication; 2. Defines key components of communication; 3. Defines tools to support communication; 4. Defines appropriate venues for communication; and 5. Defines responsibilities.

Communication, as described by the parents in this study, is the manner in which they are involved in the school and relates specifically to the information received about their child and the exchange of information by parents regarding the culture their children bring to class. When dealing with parents from different social, cultural and historical backgrounds, as with First Nations, communication between parents and teachers is even more important because of the increased barriers to understanding. Teachers require information regarding the values, beliefs and practices of children and their parents in order to create learning environments that will contribute to learning. Parents require specific information regarding what teachers are doing in the classroom to assess children and provide feedback to parents so that parents can assist their children to improve.

Key components of communication may include expectations of both parent and teacher including frequency of communication, information required, and communication formats. It was clear from this research that parents viewed certain forms of communication such as journals as working and others, based on technology as areas to be enhanced to support children and parents. My

observation of the communities was that social media is playing an increasing role in parent's lives. Report cards and other tools are also important but parents noted needing specific information that they can apply.

Venues are also important so that power imbalances between schools and First Nations parents can be remedied. Venue and method are particularly important to improving communication. Parents noted that events such as parent teacher interviews felt very structured and rushed. In one case, the teacher told the parent they must move quickly because the teacher was falling behind. Parents suggested that having to sign in at the office or make an appointment prior to entering the school was not welcoming and sent both conscious and unconscious messages about the importance of communication. Parents expressed negative past experiences with the school system and perceived insensitivity or hostility and providing neutral sites for communication to occur or even meeting parents in their own community, provides a demonstration of meaningful and responsive communication.⁷⁴

It may also be useful to hold meetings in settings that are appropriate to First Nations parents and in non-school environments. As noted by an LBN Steering Committee member:

We invited the Principal and he came up. What an uncomfortable feeling when you are not in your own setting. They have to leave their building once in a while for meetings. Now he knows how the parents feel...We've also said that the teachers need to come to our comfort zone. We invite all of the schools here. That is where you will find out who will make an effort to get to know their child's life. If this person has enough respect to come, I will respect him/her. You don't know all of the teachers. You get

_

⁷⁴ The Lake Babine Nation now holds an education night were they invite teachers to the community.

to know who the teachers are at the schools. You go to find out who was approachable and who is not (November 6, 2009).

Teachers building trust requires meeting parents in settings outside of the school and beyond specific school sanctioned events, where power imbalances are reduced and the voice of parents strengthened. The LBN Steering

Committee viewed teachers who were willing to come to the community as demonstrating effort and respect for First Nations parents. The trust built by the act of a home visit improved communication for the school year. A Steering committee member continued,

When I went to school, we had genuine teachers that liked you as an Aboriginal child. They even came to visit us on reserve. My parents could not believe the teacher wanted to come to my house. They really liked her that she was respectful enough and kind enough to visit. After that any time my dad had a concern he contacted the teacher and they sorted it out together (November 6, 2009).

Based on recommendations from the LBN Steering Committee and supported by this research, home visits and participation in other community events should be a regular part of education delivery. The reciprocal nature of these events requires that the Lake Babine Nation also take an active role in organizing and inviting school personnel to community events. This can include events specific to education and other cultural events such as *bah'lats* or food harvesting that the Lake Babine Nation feel will create greater understanding of their community, and as a result, parents. Cross-cultural understanding requires that the Lake Babine reach out as often outsiders to the Lake Babine Nation may find it difficult to extend outside of the school system without formal invitation, just as parents noted entering the school without a specific occasion.

The school districts might wish to implement monthly multiple-family group meetings, adopting an integrated approach by involving teachers and other supports such as the Aboriginal Support workers, would allow educators to consult with and inform families regarding what is taking place in school and would provide the opportunity to set up additional individual meetings regarding concerns specific to parents. Group meetings reflect traditional approaches to community planning and problem solving and may provide a less formal occasion for parents to meet with the teacher and as a result feel comfortable setting up individual meetings to discuss specific concerns related to their child. Groups could also be set up according to student need such as reading, or discipline and plans developed and monitored as a group.

Defining responsibilities is important to ensuring all parties understand their role in the communication strategy and that responsibility for monitoring of communication is assigned, and parties supported, if communication breaks down. This research indicates that while communication was excellent in some cases, communication had broken down in other examples and there was no mechanism for overseeing parent teacher communication as a core function of education. At a district level, districts may want to implement a formal communication structure, based on the definition of communication and formats, with measurable outcomes that parties could report to the Aboriginal Education Board as well as school administrators to monitor and support home and school communication. At a classroom level, it may also be useful for parents and teachers to meet and agree to expectations and define roles and responsibilities,

and formalize these agreements so that disputes in expectations can be resolved through various channels. As noted in this dissertation, parents and teachers share responsibility for the education of children and as a result, children will benefit from parents and teachers sharing similar goals and communicating with one another.

6.5 Implications

Parents were clear that they wanted access to progress reports and ability to monitor the performance of their children, and working together, parents and educators could establish common frameworks for reporting and the communication of student outcomes. Educators and parents could meet to decide upon common goals that could be reinforced and reflective of both the home and school environment. Contracts between schools and parents could include parents agreeing to support the school and their children and the school agreeing to support children in their achievement of established goals. Parents, educators and administrators could then be provided necessary training to ensure they are able to meet their commitments.

While strides have been made to increase the involvement of parents and this research provides examples of improved communication, improvement can still be made. This research showed a relationship between parent satisfaction with communication and their child's performance. Without change, we risk seeing continued low achievement rates and a gap between Lake Babine children and other citizens, particularly for the at-risk students who would most

benefit from improved communication. If achievement is to improve, parents must be involved in and understand the system. Change is a joint responsibility. While educators and administrators must take responsibility for working with parents, parents must also take responsible for reaching out to the schools—ultimately supporting improved outcomes for BC's Aboriginal students.

A wet Zah

7: REFERENCE LIST

- Aasen, W.K. (1992). Should the Clans Decide? The Problems of Modeling Self-Government Among the Lake Babine Sekani Indians of British Columbia. Edmonton: University of Alberta. Unpublished MA Thesis.
- Anderson, J.O. & Bachor, D.G. (1993). Assessment Practices in the Elementary Classroom: Perspectives of Stakeholders. Paper Presented at the Annual Meeting of the National Council for Measurement in Education, Atlanta Georgia, April 1993, 1-21.
- Ball, J. (2004). As if Indigenous Knowledge and Communities Mattered: Transformative Education in First Nations Communities in Canada. *American Indian Quarterly*, 28 (3 & 4), 454-479.
- Battiste, M. (2000). Maintaining Aboriginal Identity, Language, and Culture in Modern Society. In M. Battiste (Ed.), *Reclaiming Indigenous Voice and Vision* (pp. 192-208). Vancouver: UBC Press.
- Battisti, M., Friesen, J. & Krauth, B. (2009, September). Non-Standard English at School: Can Targeted Funding Improve Student Achievement. Retrieved from http://www.sfu.ca/cerp/research/esd_working_paper_cerp.pdf
- Battle Bailey, L. (2006). Interactive Homework: A Tool for Fostering Parent- Child Interactions and Improving Learning Outcomes for at-risk Young Children. *Early Childhood Education Journal 34*(2), 155-167.
- BC Teachers Federation. (2008). Foundation Skills Assessment. Electronic Document Retrieved April 23, 2008 from http://bctf.ca/fsa.aspx

- Becker, K.A. (2003). *History of Stanford-Binet Intelligence Scales: Content and Psychometrics*. (Stanford-Binet Intelligence Scales, Fifth Edition Assessment Services Bulletin No. 1). Itasca: Riverside Publishing.
- Ben Jaafar, S.B. & Anderson, S. (2007). Policy Trends and Tensions in Accountability for Educational Management and Services in Canada. *The Alberta Journal of Educational Research 53*(2), 207-227.
- Benner, A.D. & Mistry R.S. (2007). Congruence of Mother and Teacher Educational Expectations and Low Income Youth's Academic Competence. *Journal of Education Psychology 99*(1), 140-153.
- Berkes, F. (1999). Sacred Ecology: Traditional Ecological Knowledge and Resource Management. Philadelphia: Taylor and Francis.
- Berliner, D.C. & Biddle, B.J. (1995). *The Manufactured Crisis: Myths, Fraud, and the Attack on America's Public Schools*. New York: Basic Books.
- Booth, S. (1997). On Phenomenography, Learning and Teaching. *Higher Education Research and Development*, *16*(2), 135-158.
- Bordeaux, R. (1995). Assessment for American Indian and Alaska Native Learners. ERIC Digest. Charleston: ERIC Clearinghouse on Rural Education and Small Schools.
- Bougie, E. (2009). Aboriginal Peoples Survey, 2006: School Experience of Off-Reserve First Nations Children Aged 6 to 14. Ottawa: Statistics Canada, Social and Aboriginal Statistics Division.
- Boulton, A. (2005). *Provision at the Interface: The Maori Mental Health Contracting Experience*. PhD Thesis. Palmerston North: Massey University.
- Bourdieu, P. & Passeron, J.C. (1977). *Reproduction in Education, Society and Culture*. London: Sage.
- Brady, P. (1996). Native Dropouts and Non-Native dropouts in Canada: Two Solitudes or a Solitude Shared? *Journal of American Indian Education 35*(2), 10-20.
- Brewerton, P. & Millward, L. (2006). *Organizational Research Methods: A Guide for Students and Researchers*. Thousand Oaks: Sage.
- Brizinski, P.M. (1993). The Summer Meddler: The Image of the Anthropologist as Tool for Indigenous Formulations of Culture. In N. Dyck & J.B. Waldram (Eds.). *Anthropology, Public policy and Native Peoples in Canada* (pp. 146-165). Montreal: McGill Queens Press.

- Caldwell, J.Y., Davis, J.D, Du Bois, B., Echo-Hawk, H., Shepard Erickson, J. Goins, T., Hill, C., Hillabrant, W., Johnson, S., Kendall, E., Keemer, K., Manson, S.M., Marshall, C. Running Wolf, P., Santiago, R.L., Schacht, R. & Stone, J.B. (2005). Culturally Competent Research with American Indians and Alaska Natives: Findings and Recommendations of the First Symposium of the Working Group on American Indian Research and Program Evaluation Methodology. *American Indian and Alaska Native Mental Health Research: The Journal of the National Center*, 12(1), 1-21.
- Canadian Institutes of Health Research (2007). CIHR Guidelines for Health Research Involving Aboriginal People. Ottawa: Canadian Institutes of Health Research.
- Carey, N., Lewis L. & Farris, E. (1998). *Parent Involvement in Children's Education: Efforts by Public Elementary Schools.* Washington DC: National Center for Education Statistics. Office of Educational Research and Improvement. US Department of Education.
- Carrier Sekani Family Services (2004). Research Ethics Policy. Unpublished internal document.
- Coggins, K., Williams, E. & Radin, N. (1997). The Traditional Tribal Values of Ojibwa Parents and the School Performance of their Children: An Exploratory Study. *Journal of American Indian Education*, *36*(3), 1-15.
- Corbin, J.M & Strauss, A. (2008). Basics of Qualitative Research: Techniques and Procedures for Developing Grounded Theory (3rd Edition). Los Angeles: Sage Publications.
- Cowley, P. & Easton, S. (2007). Report Card on British Columbia's Schools 2007 Edition. Vancouver: The Fraser Institute.
- Cowley, P. & Easton, S. (2006). Report Card on Aboriginal Education in BC, 2006 Edition. Vancouver: The Fraser Institute.
- Creswell, J. (2003). *Research design: Qualitative, Quantitative and Mixed Method Approaches* (2nd Ed.). Thousand Oaks: Sage Publications.
- Creswell, J. & Miller, D.L. (2000). Determining Validity in Qualitative Inquiry. *Theory Into Practice*, *39*(3), 124-130.
- Cronshaw, S.F., Hamilton, L.K., Onyura, B.R & Winston, A.S. (2006). Case for Non Biased Intelligence Testing Against Blacks has not Been made: A Comment on Rushton, Skuy, and Bons. *International Journal of Selection and Assessment 14*(3), 278-287.

- Croizet J-C. & Dutrevis, M. (2004). Socioeconomic Status and Intelligence: Why Test Scores do not Equal Merit. *Journal of Poverty 8*(3), 91-107.
- Cutcliffe, J.R. & Harder, H.G. (2009). The Perpetual Search for Parsimony: Enhancing the Epistemological and Practical Utility of Qualitative Research Findings. *International Journal of Nursing Studies 46*, 1401-1410.
- Darling-Hammond, L. (2004). Standards, accountability, and school reform. *Teachers College Record*, *106*(6), 1047-85.
- deGroot- Maggetti, G. (2002). A Measure of Poverty in Canada: A Guide to the Debate about Poverty Lines. Ottawa: Public Justice Resource Centre.
- Depew, D.J. (2010). Darwinian Controversies: An Historiographical Recounting. *Science & Education*, *19*(4-5), 323-366.
- Denis, C. (1997). *We are not you: First Nations and Canadian Modernity.* Peterborough: Broadview Press.
- Denney, M.K., English, J.P., & Gerber, M. M. (2001). Family Home Literacy Practices: Mediating Factors for Preliterate English Learners at Risk. Paper Presented at the annual AERA Conference in Seattle Washington, April 2001. 2-15
- Denzin, N.K. & Lincoln, Y.S. (1994) Introduction: Entering the Field of Qualitative Research. In N.K Denzin & Y.S. Lincoln (Eds.), *Handbook of Qualitative Research* (pp.1-17). Thousand Oaks: Sage.
- Dickson, G. & Green, K. (2001). Paricipatory Action Research: Lessons Learning with Aboriginal Grand Mothers. *Health Care for Women International*, *22*(5), 471-482.
- Dochy, F. & McDowell, L. (1997). Introduction: Assessment as a Tool for Learning. *Studies in Educational Evaluation 23*(4), 279-298.
- Dorman, J.P. & Knightley, W.M. (2006). Initial Use of the Perceptions of Assessment Tasks Inventory (PATI) in English Secondary Schools. *The Alberta Journal of Educational Research* 52(3), 196-199.
- Dussel, E. (2002). The Invention of the Americas: Eclipse of the 'other' and the Myth of Modernity. New York: Continuum.
- Dussel, E. (1996). *The Underside of Modernity: Apel, Ricoeur, Rorty, Taylor, and the Philosophy of Liberation*. Atlantic Highlands: Humanities Press.

- Edwards, W. (2007). Exploring Evidence in Aboriginal Health. Vancouver: 2007 Indigenous Knowledges Dialogue Circle: Ways of knowing. February 28. Prince George: National Collaborating Centre for Aboriginal Peoples Health.
- Epstein, J.L. (2001). School, Family and Community Partnerships: Preparing Educators and Improving Schools. Boulder: Westview Press.
- Erwin, L.A. (1985). The Contribution made to Elementary School Self Assessment by Parent and Teacher Surveys of Climate Perceptions. Burnaby: Simon Fraser University. Unpublished MA Thesis.
- Fiske, J.A. and Patrick, B. (2000). *Cis Dideen Kat (When the Plumes Rise): The Way of the Lake Babine Nation*. Vancouver: UBC Press.
- Fiske, J.A. (1993). From Customary Law to Oral Traditions: Discursive Formation of Plural Legalisms in Northern British Columbia: 1857-1993. Invited Paper presented at Oral Traditions Conference: Humanities Research Centre. October 28. Saskatoon: University of Saskatchewan.
- Fiske, J.A. (1981). And Then We prayed Again: Carrier Women, Colonialism and Mission Schools. Vancouver: University of British Columbia. Unpublished MA Thesis.
- Freire, P. (2009). *Pedagogy of the Oppressed*. 30th Anniversary Edition. New York: Continuum International Publishing Group.
- Gall, M., Gall, J. & Borg, W. (2007). *Educational Research: An Introduction* (8th ed.). New York: Pearson Education Inc.
- Guest, G., Bunce, A., & Johnson, L. (2006). How Many Interviews Are Enough? An Experiment with Data Saturation and Variability. *Field Methods, 18*(1), 59-82.
- Gould, J. (2008). Non-Standard Assessment Practices in the Evaluation of Communication in Australian Aboriginal Children. *Clinical Linguistics & Phonetics*, *22*(8), 643-657.
- Gould, S.J. (1996). *The Mismeasure of Man: Revised and Expanded Edition*. New York: Norton.
- Graveline, F.J. (1998). *Circle Works: Transforming Eurocentric Consciousness*. Halifax: Fernwood Publishing.
- Grolnick W.S., Benjet C., Kurowski, C.O. & Apostoleris N.H. (1997). Predictors of Parent Involvement in Children's Schooling. *Journal of Education Psychology*, 89(3), 538-548.

- Guppy, N. & Davies S. (1999). Understanding Declining Confidence in Public Education. *Canadian Journal of Education 24*(3), 265-280.
- Gutmann, A. (1995). Relativism, Deconstruction and the Curriculum. In J. Arthur & A Shapiro (Eds.). *Campus Wars: Multiculturalism and the Politics of Difference* (pp.57-68). Boulder: Westview Press.
- Haig-Brown, C. (1995). *Taking control : Power and Contradiction in First Nations Adult Education*. Vancouver: UBC Press.
- Hallett, D., Want, S.C., Chandler, M.J., Koopman, L.L., Flores, J.P. & Gehrke, E.C. (2008). Identity in Flux: Ehtnic Self- Identification and School Attrition in Canadian Aboriginal Youth. *Journal of Applied Developmental Psychology 29*, 62-75.
- Hara S.R. & Burke, D.J. (1998). Parent Involvement: The Key to Improved Student Achievement. *School Community Journal*, 8(2), 219-228.
- Harris, F. (2009). Can *Maori* Children Really Be Positioned as "Deficient" Learners for Reading English? *English Teaching: Practice and Critique, 8*(3), 123-245.
- Helin, C. (2006). *Dances with Dependency: Indigenous Success through Self Reliance*. Vancouver: Orca Spirit Publishing and Communications Inc.
- Henderson, A.T. & Mapp, K.L. (2002). A New Wave of Evidence: The Impact of School, Family and Community Connections on Student Achievement. Annual Synthesis, 2002. Austin: National Center for Family and Community Connections with Schools. Southwest Educational Development Laboratory.
- Hertzman, C. (1998). The Case for Child Development as a Determinant of Health. *Canadian Journal of Public Health 89*(1), 14-19.
- Hickling-Hudson, A. & Ahlquist, R. (2003). Contesting the Curriculum in the Schooling of Indigenous Children in Australia and the United States: From Eurocentrism to Culturally Powerful Pedagogies. *Comparative Education Review*, *47*(1), 64-89.
- Hilliard, A.G. (1991/92). Why we must Pluralize the Curriculum. *Education Leadership*, 49(4), 12-16.
- Hillocks, G. (2002). *The Testing Trap: How State Writing Assessments Control Learning*. New York: London Teacher's College Press.

- Holyk, T. (2008). Research in Carrier Sekani Communities: Carrier First Nations Perceptions of Quality Research Design and Ethics. Panel Presentation. National Collaborating Centres for Public Health. August 5-7. Kelowna BC.
- Holyk, T. & George, E. (2003). Lake Babine Nation Community Needs Assessment 2003. Prince George: Carrier Sekani Family Services.
- Holyk, T. Shawana, P., Adam, W. (2005). *Whu' Neeh Nee'* (Guiders of Our People): Strengthening Lake Babine First Nation Law Through Research and Training. Invited Paper, presented at the World Indigenous Peoples Conference on Education. November 28- December 1. Hamilton New Zealand.
- Hudson, D.R. (1983). *Traplines and Timber: Social and Economic Change Among the Lake Babine Indians of Northern British Columbia.* Edmonton: University of Alberta. Unpublished PhD Thesis.
- Hulpia, H. & Valcke, M. (2004). The Use of Performance Indicators in a School Improvement Policy: The Theoretical and Empirical Context. *Evaluation and Research in Education*, *18*(1&2), 102-119.
- Indian and Northern Affairs Canada. 2010. First Nations Profiles. Lake Babine Nation. Retrieved May 12, 2010 from, http://pse5-esd5.ainc-inac.gc.ca/fnp/Main/Search/FNMain.aspx?BAND_NUMBER=607&lang=eng
- Indian and Northern Affairs Canada. 2009. First Nations Profiles. Lake Babine Nation. Retrieved October 7, 2009 from, http://pse5-esd5.ainc-inac.gc.ca/fnp/Main/Search/FNMain.aspx?BAND_NUMBER=607&lang=eng
- Indian and Northern Affairs Canada. (2008). Registered Population. Lake Babine Nation. Electronic Document retrieved March 23, from http://sdiprod2.inac.gc.ca/fnprofiles/FNProfiles_GeneralInformation.asp?BAN D_NUMBER=607&BAND_NAME=Lake+Babine+Nation
- (2007). Joint Special Report. Health and Well-being of Children in Care in British Columbia: Educational Experiences and Outcomes. Victoria: Office of the Provincial Health Officer.
- Joseph, S. (1991). Assimilation Tools: Then and Now. BC Studies (89), 65-79.
- Kavanagh, B. (1998). Reaching For Success: A Discussion Paper. Vancouver: First Nations Education Steering Committee.
- Kelm, M.E. (1998). *Colonizing Bodies: Aboriginal Healing in British Columbia* 1900-1950. Vancouver: UBC Press.

- Kemmis, S. & McTaggart, R. (2005). Participatory Action Research: Communicative Action and the Public Sphere. In, N. Denzin & Y. Lincoln (Eds.), *The Sage Handbook of Qualitative Research* (3rd Ed.) (pp.559-603). Thousand Oaks: Sage Publications.
- Kendall, P. (2009). Pathways to Health and Healing: 2nd Report on the Health and Well-being of Aboriginal People in British Columbia. Provincial Health Officer's Annual Report 2007. Victoria: Office of the Provincial Health Officer.
- Klenowski, V. (2009). Australian Indigenous Students: Addressing Equity Issues in Assessment. *Teaching Education*, *20*(1), 77-93
- Klug, B.J. & Whitfield, P.T. (2003). *Widening the Circle: Culturally Relevant Pedagogy for American Indian Children*. New York: Routledge Falmer.
- Knudtson, P. and Suzuki, D. (1992). *Wisdom of the Elders.* Toronto: Stoddart Publishing Co. Limited.
- LaFrance, B. (2000). Culturally Negotiated Education in First Nations Communities: Empowering Ourselves for Future Generations. In M.B. Castellano, L. Davis & L. Lahache (Eds.). *Aboriginal Education: Fulfilling the Promise* (pp.101-113). Vancouver: UBC Press.
- Laitsch, D. (2005). A Policy Primer on Testing and Assessment. ASCD, 41 online. Retrieved from http://www.ascd.org/publications/newsletters/infobrief/jul05/num42/toc.aspx
- Laitsch, D. (2003). Classroom Assessment and Student Motivation. *ASCD. Research Brief Translating Education Research into Action (1)* 17. Retrieved from http://www.ascd.org/publications/researchbrief/v1n17/toc.aspx
- Lemert, C. (1994). Dark Thoughts about the Self. In, C. Calhoun (Ed.), *Social Theory and the Politics of Identity* (pp.100-130). Cambridge: Blackwell.
- Leyva, R. (2009). No Child Left Behind: A Neoliberal Repackaging of Social Darwinism. *Journal for Critical Education Policy Studies*, 7(1), 364-381.
- Macionis, J.J. & Gerber, L.M. (2005). *Sociology.* 5th Canadian Edition. Toronto: Prentice Hall.
- Marks, G.N. (2006). Are Between- and Within- School Differences in Student Performance Largely Due to Socio-Economic Background? Evidence from 30 Countries. *Educational Research*, *48*(1), 21-40.

- Martinez, R-A., Martinez, R. & Perez, M. (2004). Children's School Assessment: Implications for Family- School Partnerships. *International Journal of Educational Research*, *41*, 24-39.
- MClaren, A. (1990). *Our Own Master Race. The Eugenic Crusade in Canada*. Toronto: McClelland and Stewart.
- McLean, D.L. (1997). Rural Alaska Native Perceptions of Cultural Transmission: Implications for Education. *Journal of American Indian Education*, *36*(3), 16-26.
- Meaghan, D. E., & Casas, F. R. (1995). On the Testing of Standards and Standardized Achievement Testing: Panacea, Placebo, or Pandora's Box? *Interchange*, *26*(1), 33-58.
- Mellor, S. & Corrigan, M. (2004). *The Case for Change: A review of Contemporary Research on Indigenous Education Outcomes*. Camberwell: ACER Press.
- Ministry of Aboriginal Relations and Reconciliation. (2008). New Relationships with Aboriginal People and Communities in B.C. Victoria: Ministry of Aboriginal Relations and Reconciliation.
- Ministry of Education. (2010a). Aboriginal Education Enhancement Agreements. Electronic Document. Retrieved April 13,2010 from www.bced.gov.bc.ca/abed/agreements
- Ministry of Education. (2010b). Provincial Examinations. Electronic Document. Retrieved April 12,2010 from http://www.bced.gov.bc.ca/exams/specs/
- Ministry of Education. (2008a). Student Assessment. Electronic Document. Retrieved April 23,2008 from http://www.bced.gov.bc.ca/assessment/98highlights/
- Ministry of Education. (2008b). Foundation Skills Assessment (FSA). Retrieved April 23, 2008 from http://www.bced.gov.bc.ca/assessment/fsa/
- Ministry of Education. (2008c). Aboriginal Education Enhancement Agreements. Retrieved April 30, 2008, from http://www.bced.gov.bc.ca/abed/agreements/welcome.htm
- Ministry of Education. (2000). *BC Performance Standards: Social Responsibility*. Victoria: Ministry of Education, Student Assessment and Program Evaluation Branch.

- Ministry of Education (1994). Performance Assessment: Assessment Handbook Series. Victoria: Ministry of Education. Curriculum Development Branch.
- Minore, B., Boone, M., Katt, M., Kinch, P. & Birch, S. (2004). Addressing the Realities of Health Care in Northern Aboriginal Communities Through Participatory Action Research. *Journal of Interprofessional Care, 18*(4), 360-368.
- Miretzky, D. (2004). The Communication Requirements of Democratic Schools: Parent- Teacher Perspectives on their Relationships. *Teachers College Record* 106(4), 814-851.
- Mitchell, T.L & Baker, E. (2005) Community- Building Versus Career-Building Research: The Challenges, Risks, and Responsibilities of Conducting Research with Aboriginal and Native American Communities. *Journal of Cancer Education*, 20 (1), 41-46.
- Moore, R. (1978) *The Historical Development of the Indian Act.* Ottawa: Treaties and Historical Research Centre, P.R.E. Group, Indian and Northern Affairs.
- Morley, L. & Rassool, N. (1999). *School Effectiveness: Fracturing the Discourse*. New York: Falmer Press.
- Munro, J.B. (1945). Language, Legends and Lore of the Lake Babine Indians. Ottawa: University of Ottawa. Unpublished PhD. Thesis.
- Neisser, U., Boodoo, G., Bouchard, T. J. Jr., Boykin, A. W., Brody, N., Ceci, S. J., Halpern, D. F., Loehlin, J. C., Perloff, R., Sternberg, R. J., Urbina, S. (1996). Intelligence: Knowns and Unknowns. *American Psychologist*, 51(2), 77-101.
- Nelson-Barber, S. & Trumbull, E. (2007). Making Assessment Practices Valid for Indigenous American Students. Journal of American Indian Education 46(3) 132-147.
- Onwuegbuzie, A.J. & Collins, K.M.T. (2007). A Typology of Mixed Methods Sampling Designs in Social Science Research. *The Qualitative Report 12*(2), 281-316.
- Onwuegbuzie, A.J. & Leech, N.L. (2007) Sampling Designs in Qualitative Research: Making the Sampling Process More Public. *Qualitative Report* 12(2), 238-254.
- Patton, M. Q. (1990). *Qualitative Evaluation and Research Methods*. Newbury Park: Sage Publications.

- Pewewardy, C. (2002). Learning Styles of American Indian/Alaska Native Students: A Review of the Literature and Implications for Practice. *Journal of American Indian Education*, *41*(3), 22-56.
- Prince, D.L. & Howard, E.M. (2002). Children and Their Basic Needs. *Early Childhood Education Journal* 30(1), 27-31.
- Prince, D.L., Pepper, K. & Brocato, K. (2006). The Importance of Making the Well-Being of Children in Poverty a Priority. *Early Childhood Education Journal* 34(1), 21-28.
- Pushor, D. & Murphy, B. (2004). Parent Marginalization, Marginalized Parents: Creating a Place for Parents on the School Landscape. *The Alberta Journal of Educational Research*, *50*(3), 221-235.
- Raptis, H. & Fleming, T. (2006). Large-Scale Assessment Outcomes in British Columbia, 1876-1999. *Canadian Journal of Education* 29(4), 1191-1222.
- Reyhner, J. (1992). American Indians out of School: A Review of School-Based Causes and Solutions. *Journal of American Indian Education*, *31*(3), 37-56.
- Richards, J. & Vining, A. (2004). Aboriginal Off-Reserve Education: Time for Action. *CD Howe Institute Commentary No. 198 April*. Ottawa: Renouf Publishing.
- Robinson-Zanartu C., (1996). Serving Native American Children and Families: Considering Cultural Variables. *Language, Speech, Hearing Services in Schools*, *27*(4), 373-84.
- Robinson-Zanartu, C. & Majel-Dixon, J. (1996). Parent Voices: American Indian Relationships with Schools. *Journal of American Indian Education*, *36*(1), 33-54.
- Roman, L.G., Brown, S., Noble, S., Wainer, R. & Young, A.E. (2009). No Time for Nostalgia!: Asylum-Making, Medical Colonialism in British Columbia (1859-97) an artistic Praxis for Social Transformation. *International Journal of Qualitative Studies in Education*, *22*(1), 17-63.
- Rosenzweig, C. (2001). A Meta Analysis of Parenting and School Success: The Role of Parents in Promoting Students' Academic Performance. Paper Presented at the annual AERA Conference in Seattle Washington, April 2001. 1-44.
- Royal Commission on Aboriginal Peoples. (1996). For Seven Generations: An Information Legacy of the Royal Commission on Aboriginal Peoples. Ottawa: Liraxus Inc. CD-ROM.

- Rumberger, R.W. (1995). Dropping out of Middle School: A Multilevel Analysis of Students and Schools. *American Educational Research Journal 32*(3), 583-625.
- Ryan, G.W. and Bernard, H.R. (2003). Techniques to Identify Themes. *Field Methods*, *15*(1), 85-109.
- Sandelowski, M. (2001). Real Qualitative Researchers Do Not Count: The use of Numbers in Qualitative Research. *Research in Nursing & Health*, *24*(3), 230-240.
- Sandelowski, M. (1995). Sample Size in Qualitative Research. *Research in Nursing Health*, *18*(2), 179-183.
- School District No. 91. (2006). 2005-2006 Report on Aboriginal Education in School District No. 91 (Nechako Lakes). Vanderhoof: School District No. 91 and The First Nations Education Council.
- Shawana, P. (2007). Legal Processes, Pluralism in Canadian Jurisprudence, and the Governance of Carrier Medicine knowledge. In Law Commission of Canada (Ed.) *Indigenous Legal Traditions*. Vancouver: UBC Press.
- Shepard, L.A. (2000). The Role of Assessment in a Learning Culture. *Educational Researcher*, 29(7), 4-14.
- Shepard, L.A. (1989). Why we need Better Assessments. *Educational Leadership*, 46(7), 4-9.
- Shields, C.M. (1997). Learning about Assessment from Native American Schools: Advocacy and Empowerment. *Theory into Practice 36*(2), 102-109.
- Smith, Linda Tuhiwai (1999). *Decolonizing Methodologies: Research and Indigenous Peoples*. New York: Zed Books.
- Smylie, J. (2007). Exploring Evidence in Aboriginal Health. Vancouver: 2007 Indigenous Knowledges Dialogue Circle: Ways of knowing. February 28. Prince George: National Collaborating Centre for Aboriginal Peoples Health
- Smylie, J., Martin, C.M., Kaplaan-Myrth, N., Steele, L., Tait, C. & Hogg, W. (2003). Knowledge Translation and Indigenous Knowledge. *Circumpolar Health*, 139-143.
- St. Denis, V. (1992). Community Based Participatory Research: Aspects of the Concept Relevant for Practice. *Native Studies Review*, 8(2), 51-74.

- Steinhauer, E. (2002). Thoughts on an Indigenous Research Methodology. *Canadian Journal of Native Education*, 26(2), 69-81.
- Stiggins, R. (2002). Assessment crisis: The absence of assessment for learning. *Phi Delta Kappan, 83*(10), 758-65.
- Stiggins, R. (1997). *Student Centered Classroom Assessment* (2nd Ed.). Upper Saddle River: Merrill.
- Svihla, V., Vye, N., Brown, M., Phillips, R., Gawel, D., & Bransford, J. (2009). Interactive Learning Assessments for the 21st Century. *Education Canada*, 49(3), 44-47.
- Taylor, C. (2004). Modern Social Imaginaries. Durham: Duke University Press.
- Thrupp, M. (2001). Sociological and Political Concerns about School Effectiveness Research: Time for a new *Research Agenda*. School Effectiveness and School Improvement, 12(1), 7-40.
- Tobias, J. (1991). Protection, Civilization, Assimilation: An Outline History of Canada's Indian Policy. In, Miller J.R. ed. *Sweet Promises: A Reader on Indian-White Relations in Canada*. Toronto: University of Toronto Press. 127-144.
- *Tri Council Policy Statement. Ethical Conduct for Research Involving Humans* (2005). Ottawa: Interagency Secretariat on Research Ethics.
- Turkheimer, E., Haley, A., Waldron, M., D'Onofrio, B. & Gottesman, I.I. (2003). Socioeconomic Status Modifies Heritability of IQ in Young Children. *Psychological Science* 14(6), 623-628.
- Ungerleider, C. (2003). *Failing Our Kids: How we are ruining our Public Schools*. Toronto: McClelland and Stewart Ltd.
- UNESCO. (1995). *Families and Education*. Occasional Papers Series, No. 18. Vienna: UNESCO.
- United Nations Declaration on the Rights of Indigenous Peoples. (2007). Adopted by General Assembly Resolution 61/295 on 13 September 2007.
- Valencia, R., & Villarreal, B. (2003). Improving students' reading performance via standards based reform: A critique. *The Reading Teacher, 56*(7), 612-21.
- Van de Grift, W.J. & Houtveen, A.A. (2006). Underperformance in Primary Schools. *School Effectiveness and School Improvement* 17(3), 255-273.

- Volante, L. & Ben Jaafar, S. (2008). Profiles of Education Assessment Systems Worldwide: Educational Assessment in Canada. Assessment in Education: Principles, Policy & Practice, 15(2), 201–210
- Volante, L. (2006). An Alternative Vision for Large-scale Assessment in Canada. Journal of Teaching and Learning, 4(1), 1-14.
- Watt-Cloutier, S. (2000). Honouring Our Past, Creating Our Future: Education in Northern and Remote Communities. In M.B. Castellano, L. Davis & L. Lahache (Eds.). *Aboriginal Education: Fulfilling the Promise* (pp.114-128). Vancouver: UBC Press.
- Weiss, H., Little, P., Bouffard, S.M., Deschenes, S.N. & Malone, H.J. (2009). Strengthen What Happens outside School to Improve What Happens inside. *Phi Delta Kappan*, *90*(8), 592-596.
- Wiggins, G. (1993). Assessing student performance: Exploring the purpose and limits of testing. San Francisco: Jossey-Bass.
- Wilbrink, B. (1997). Assessment in Historical Perspective. *Studies in Educational Evaluation 23*(1), 31-48.

8: APPENDIX A: LBN RESEARCH PROTOCOL

LAKE BABINE NATION/RESEARCHER ASSESSMENT RESEARCH RESEARCH PROTOCOL AGREEMENT

| BE | | |
|----|--|--|
| | | |
| | | |

Lake Babine Nation

AND:

Travis Holyk

(hereinafter "The Researcher")

WHEREAS:

- A. Travis Holyk in the fulfillment of his Doctorate of Education Degree at Simon Fraser University must complete research.
- B. Travis Holyk with the assistance of the Lake Babine Nation intends to undertake research for the purposes of understanding parent's perceptions of academic assessment.
- C. The Lake Babine Nation wishes to collaborate in the research project- U'h Nook Noh Ga'nah (in our words): Understanding of Student Assessment by Lake Babine Nation Parents- with the Researcher for the benefit of its membership.

1. PURPOSE

- 1.1. The parties enter into this agreement for the following purposes:
 - To define the roles and responsibilities of each party in the research process.
 - b. To ensure that the Lake Babine Nation assessment research is conducted in a respectful, professional and ethical manner.
 - c. To define policy for archiving and management of research materials.

1

2. ROLES AND RESPONSIBILITIES

- 2.1. The Researcher is responsible for completing the research and employing individuals such as transcribers who may be needed in the fulfillment of the project. Wherever possible, these individuals will be members of the Lake Babine Nation.
- 2.2. The Researcher, with assistance from the Lake Babine Nation will establish a Steering Committee, reflecting Elected Leadership, Hereditary Leadership, and individuals involved in education at the Lake Babine Nation.
- 2.3. The Researcher is responsible for any costs associated with honorariums paid to interview participants. Honorariums will be paid at the Researcher's discretion in consultation with the Lake Babine Nation in order to adhere to generally accepted policies for payment of honorariums within the Nation.
- 2.4. The Researcher, in collaboration with the Lake Babine Nation, will ensure that at all times the researcher conducts interviews and pursues research activities in a manner that is consistent with the highest standards of ethical and scholarly social scientific practices as noted by the Research Ethics Board of Simon Fraser University.
- 2.5. The Lake Babine Nation will provide the Researcher with names and contact information of parents who have school-aged children as required. The Researcher will make the request to the education department at the Lake Babine Nation.
- 2.6. The Lake Babine Nation will provide the Researcher with space to conduct interviews if required.

3. ETHICAL RESPONSIBILITIES

- 3.1. The Researcher, prior to conducting research with the Lake Babine Nation, will obtain research ethics approval from the Simon Fraser University Research Ethics Board.
- 3.2. The Researcher, in collaboration with the Lake Babine Nation and in accordance with SFU Research Ethics guidelines, will ensure that each potential interview candidate voluntarily gives her or his informed consent to participate in the interview process. Informed consent includes a potential interview candidate freely agreeing to participate in the interview process based on information provided including, but not limited to, who is conducting the research, the purposes of the research, information use and storage and statements of confidentiality.

- 3.3. The Researcher, in collaboration with the Lake Babine Nation and in accordance with SFU Research Ethics guidelines, will ensure that potential interview participants are fully informed of potential risks and possible benefits of their participation.
- 3.4. The Researcher, in collaboration with the Lake Babine Nation and in accordance with SFU Research Ethics guidelines, will ensure that potential interview participants are given adequate opportunity to discuss and contemplate their participation in the interview process.
- 3.5. The Researcher, in collaboration with the Lake Babine Nation and in accordance with SFU Research Ethics guidelines, will ensure that potential interview participant are provided with the name and contact information for a person who may be contacted in case of concerns and/or complaints regarding the research.
- 3.6. The Researcher, in collaboration with the Lake Babine Nation and in accordance with SFU Research Ethics guidelines, will ensure that potential interview participants are aware that their participation is voluntary and that at any time they are entitled to withdraw their consent to participate in the interview process or decline to answer a question without sanction.

4. RESEARCH MATERIAL

- 4.1. The researcher will collect and house the data for the period of time necessary to complete his dissertation.
- 4.2. Upon completion of his dissertation, the Researcher will archive copies of the research material in facilities designated by Carrier Sekani Family Services for those purposes in trust for the Lake Babine Nation. The Researcher will have the right to access and use the Research Materials for reports and scholarly articles in perpetuity. Any new use of the Research Materials, not outlined above, will require consent of the Lake Babine Nation.
- 4.3. The researcher will destroy material, including lists of participant names provided by the Lake Babine Nation after the researcher completes a final report accepted by the Lake Babine Nation.
- 4.4. During the research process, the Researcher will restrict access to research material to only those persons whose access is necessary for the purposes of the research project. Access by any other party will require the express written consent of the participants.
- 4.5. The Researcher will provide copies of interview transcripts to participants for their review and comments.

5. FINAL REPORT

- 5.1. The Researcher will provide the results of the questionnaires, in the form of a report, at a Steering Committee meeting and will provide a copy of the final research report to the Lake Babine Nation.
- 5.2. The Researcher will be the owner of the final research report, Dissertation and any articles/publications resulting from the research and written by the researcher.
- 5.3. The information provided to the Researcher by the participants remains the property of the participants.

6. BREACH OF CONTRACT

6.1. Breach of the terms of this agreement by the Researcher may result in the Lake Babine Nation withdrawing their participation in the research.

FOR LAKE BABINE NATION

THE RESEARCHER

Travis Howk

9: APPENDIX B: LBN LETTER OF APPROVAL



Lake Babine Nation

225 SUS AVENUE, P.O. BOX 879 BURNS LAKE, B.C. VIJ 1EO TEL: (250) 692-4700 FAX: (250) 692-4790

June 2, 2008

Simon Fraser University Office of Research Ethics Simon Fraser University 8888 University Drive Multi-Tenant Facility Burnaby, B.C. V5A 1S6

Re: U'H NOOK NOH GA'NAH (IN OUR WORDS): LAKE BABINE NATION PARENTS' UNDERSTANDING OF SCHOOL ASSESSMENT

On behalf of the Chief and Council of the Lake Babine Nation, this letter is in support of Travis Holyk's proposal to undertake research regarding assessment within our nation. We recognize the connection education has to community well being and while we have made great strides in improving education within our nation, graduation rates continue to fall behind the general population. Assessment has always been a challenge for our nation as we struggle to ensure tools are fair and meet the needs of our students.

This project represents an excellent opportunity for partnership with Simon Fraser University in strengthening the role of Indigenous peoples in the academic system. Ensuring a voice for parents will make the education setting a more welcoming place for parents and students, hopefully improving student outcomes.

We are pleased that this research respects the important role parents play in the education of their children by engaging them in dialogue and documenting their views.

Sincerely,

Chief Betty Patrick

10: APPENDIX C: RESEARCH QUESTIONS AND INTERVIEW PROTOCOL

<u>D</u>6

| emo | <u>ograph</u> | ic Information: |
|-----|----------------------|--|
| 1. | Do you | live on reserve off reserve? |
| 2. | Are you | u currently employed? |
| | 1. 2. | Yes No |
| 3. | If empl | oyed, What type of work do you do? |
| 4. | | feel you have enough money to support your basic needs ag, food, shelter) |
| | 1. 2. | Yes No |
| 5. | What is | s your monthly income? |
| | 2. 3. 4. 5. | 0-500 501-1000 1001-2000 2001-3000 3001-4000 4001-5000 5001+ |
| 6. | What is Guardi | s the highest level of education you completed (Father, Mother an)? |
| | 2. 3. 4. 5. | Grade High school – Dogwood High School- Leaving School Certificate College Trade University Degree University Masters Degree |

| 7. V | Nhat is the highest Level of Education your spouse completed? 1. Grade 2. High school – Dogwood 3. High School- Leaving School Certificate 4. College Trade 5. University Degree 6. University Masters Degree |
|-------------|---|
| 8. H | How many Children do you have? |
| 9. V | What age are they? |
| 10.V | What grade are they in? |
| 11.[| Do you assist your child with home work? |
| 12. H | Have you met your child's teacher? |
| <u>Open</u> | <u>Questions</u> |
| 1. V | What does education mean to you? |
| 2. H | How do we know someone is educated (Cultural and/or Western)? |
| 3. V | What do you want your child to achieve from school? |
| 4. V | What is it you would like your child to learn at school? |
| 5. V | What is the role of a school in your child's life? |
| 6. H | How should we determine success in school? |
| 7. V | What kinds of information do you receive about your child? |
| | a. How often? When? |

9. On an ongoing basis, what strategies do you use to check how your son/daughter is progressing in school?

8. How do you use this information?

- 10. What do you think teachers are doing in the classroom to understand how your child is doing?
- 11. What do you feel would be the best way for teachers to find out how your child is doing in school?
- 12. What form of testing would help students achieve in both Carrier and Western culture?
- 13. How do you feel about provincial testing in Grade 4?
- 14. Do you want to know how your son/daughter is progressing compared to other children his/her age?
- 15. How could schools make it easier for you to be involved in testing your child's progress in school?
- 16. When you think about your child's understanding in school, do you agree with the results of school tests?
- 17. What should passing to the next grade be based on?

Interview Protocol

1. I began each interview by discussing and having participants sign the Informed Consent By Participants in a Research Study form (Attached as Appendix D). Parents were also informed that permission was obtained by Chief and Council to conduct the research and a Steering Committee made up of community members assisted in the project design and questions. Participants were informed that Chief and Council and the Steering Committee would not have access to who participated in the study and only aggregate information with no identifying information would be presented.

No information was recorded until consent was obtained from the participant.

2. I requested permission to tape record interviews.

If participants spoke while recorder was off, I recorded notes and sought approval for inclusion in the project. Occasions where this happened included when a break was taken because of a phone call or other interruption and information was provided prior to being able to turn on the tape recorder. Or, when an interview had been concluded and a person then added information as I was leaving the tape recorded was no longer available. Notes were included at the end of transcripts sent to each participant in order to ensure accuracy of information.

- 3. I used same questions at each site. Prompts were used if required. Prompts included repetition of parts of answers and asking if the participant had anything else add or clarifying the response by asking did you mean?... Examples of additional Prompts included:
 - 1. Would it be important to you that your child learned her traditions?
 - 2. How often does this happen?
 - 3. So that would be helpful... can you explain
 - 4. Any other things that you can think of?
 - 5. Can you think of a better way they could do it [assessment]?
- 4. At times participants did not understand a question and explanation was provided. Further explanation was only provided if parents did not understand and either did not respond, clearly indicating they did not understand or asked directly what was meant.
 - How do we know someone is educated (Cultural and/or Western)?
 By cultural it was explained as If you looked at a person from the
 community what would make you say they were educated in LBN
 or Carrier traditions. Western was defined as Nedo or White
 education if the term was not understood.
 - 2. What is the role of a school in your child's life? Thinking of the roles people play in the lives of a child, children are at school for quite a few hours per day, what do you see as the role of the school in your son's life?
 - 3. How do you feel about provincial testing in Grade 4? Many parents were unaware of provincial testing. I first asked if parents knew about provincial testing or had heard about the Foundation Skills Assessment. If they were unsure, they were informed of the Foundation Skills Assessment through the following script: Assessment is completed in grade 4 and 7. Every child in public school must participate. The assessment focuses on numbers, reading and writing which the Ministry of Education has determined are the foundational skills that each child needs in order to be successful in the school system. It is used to evaluate how well students are achieving basic skills, and is reported at a provincial,

- district, school and individual level (which means you will receive results). Parents were encouraged to contact their school if they wanted more information.
- 4. When you think about your child's understanding in school, do you agree with the results of school tests? When you read with or help your daughter with her homework, when you think about how she is doing, is it similar to the information sent home?
- 5. What is your monthly income?
 - 1. 0-500
 - 2. 501-1000
 - 3. 1001-2000
 - 4. 2001-3000
 - 5. 3001-4000
 - 6. 4001-5000
 - 7. 5001+

Some Participants were uncomfortable stating their income. When this happened I gave them the option of circling a category themselves on the question sheet.

| 11: APPENDIX D: INFORMED CONSENT BY PARTICIPANTS IN A |
|---|
| RESEARCH STUDY |

Form 2- Informed Consent By Participants In a Research Study

The University and those conducting this research study subscribe to the ethical conduct of research and to the protection at all times of the interests, comfort, and safety of participants. This research is being conducted under permission of the Simon Fraser Research Ethics Board. The chief concern of the Board is for the health, safety and psychological well-being of research participants.

Should you wish to obtain information about your rights as a participant in research, or about the responsibilities of researchers, or if you have any questions, concerns or complaints about the manner in which you were treated in this study, please contact the Director, Office of Research Ethics by email at hweinber@sfu.ca or phone at 778-782-6593.

Your signature on this form will signify that you have received a document which describes the procedures, whether there are possible risks, and benefits of this research study, that you have received an adequate opportunity to consider the information in the documents describing the study, and that you voluntarily agree to participate in the study.

Title: U'H NOOK NOH GA'NAH (IN OUR WORDS): LAKE BABINE NATION PARENTS' UNDERSTANDING OF SCHOOL ASSESSMENT

Investigator Name: Travis Holyk

Investigator Department: Education

Having been asked to participate in the research study named above, I certify that I have read the procedures specified in the Study Information Document describing the study. I understand the procedures to be used in this study and the personal risks to me in taking part in the study as described below:

Purpose and goals of this study:

The goal of this research is to analyze Lake Babine Nation parents' perceptions of assessment in primary school. The analysis includes definitions of student achievement and the concept of education from a Lake Babine Nation perspective.

What the participants will be required to do:

You will be asked to participate in an interview that will take roughly one to two hours to complete, during which time you will be asked to respond to questions asked by the researcher.

Risks to the participant, third parties or society:

There are no perceived risks to participating in this study.

Benefits of study to the development of new knowledge:

This project will allow parents direct involvement in the development and maintenance of their children's education through discussion and awareness of assessment. The knowledge created can build reference to First Nation perspectives regarding assessment and may inform policy and practice for all stakeholders working with First Nations. The overall goal is to improve First Nation children's outcomes in school by improving the school system. Assessment plays a large role in the school system and as a result should involve parents.

Statement of confidentiality: The data of this study will maintain confidentiality of your name and the contributions you have made to the extent allowed by the law.

All the information the researcher collects from you (Your Research Material) will be kept confidential and your anonymity will be protected. Interviews will be tape recorded (with consent), and transcribed as soon after the interview as possible. Transcripts will be identified by code only, with identifying information kept in a locked cabinet separate from tapes and transcripts. The researcher will provide you with full transcripts of your interview so that you have the opportunity to reply with comments or changes.

Interview of employees about their company or agency:

Permission has been obtained from Chief and Council of the Lake Babine Nation to conduct this study.

Inclusion of names of participants in reports of the study:

Names will not be used in the final report. Your perspective and knowledge will be included in the dissertation without identifying information.

Contact of participants at a future time or use of the data in other studies:

Upon completion of his dissertation, the Researcher will archive copies of the research material in facilities designated by Carrier Sekani Family Services for those purposes in trust for the Lake Babine Nation. The Researcher will have the right to access and use the Research Materials for reports and scholarly articles. Any new use of the Research Materials (beyond reports and scholarly articles of the researcher) will require you and the Lake Babine Nation to provide consent.

I understand that I may withdraw my participation at any time. I also understand that I may register any complaint with the Director of the Office of Research Ethics.

Dr. Hal Weinberg
Director, Office of Research Ethics
Office of Research Ethics
Simon Fraser University
8888 University Drive
Multi-Tenant Facility
Burnaby, B.C. V5A 1S6
hal weinberg@sfu.ca

I may obtain copies of the results of this study, upon its completion by contacting:

Travis Holyk Carrier Sekani Family Service 987 4th Ave Prince George BC V2L 3H7 Via written request

I understand the risks and contributions of my participation in this study and agree to participate:

| The participant and witness shall fill in this area. Please print legibly | | | | |
|---|---|--|--|--|
| Participant Last Name: | Participant First Name: | | | |
| Participant Contact Information: | | | | |
| | | | | |
| Participant Signature (for adults): | Witness (if required by the Office of Research Ethics): | | | |
| Date (use format MM/DD/YYYY) | Contact at a future time / use of data in other studies | | | |
| | | | | |